



Protection Coordination Study in Bhutan Transmission Network

Thesis for the Degree of Master of Science

Thinley Gyeltshen

Department of Energy and Environment *Division of Electrical Power Engineering* CHALMERS UNIVERSITY OF TECHNOLOGY Göteborg, Sweden, 2010

Protection Coordination Study in Bhutan Transmission Network

Thinley Gyeltshen

Examiner: Dr. Tuan A. Le, Chalmers University of Technology Supervisor: Dr. Tuan A. Le, Chalmers University of Technology



Department of Energy and Environment Division of Electric Power Engineering CHALMERS UNIVERSITY OF TECHNOLOGY Göteborg, Sweden Protection Coordination study in Bhutan Transmission Network

THINLEY GYELTSHEN

© THINLEY GYELTSHEN, 2010

Master's Thesis 2010

Department of Energy and Environment Division of Electric Power Engineering Chalmers University of Technology SE-412 96 Göteborg, Sweden

Telephone + 46 (0)31-772 1000

Acknowledgments

This thesis has been carried out at Division of Electric Power Engineering under Department of Energy and Environment of Chalmers University of Technology.

I would like to extend my profound gratitude to my supervisor Dr. Tuan A. Le, for his invaluable guidance, inspiration and assistance throughout this thesis work. Without his encouragement and constant guidance I could not have complete this thesis. I also express my indebted gratitude to Professor Dr. Lena Bertling, Head of Division of Electric Power Engineering for buying CAPE software for my thesis.

I would like to thank the Management of Bhutan Power Corporation Limited, Thimphu, Bhutan for the scholarship to pursue my studies, without which, this study wouldn't have been possible.

My special thanks to Gustavo Pinares, who helped me to work with the CAPE software. I also thank CAPE support center for the prompt support whenever I had problem with the software. I would like to thank all my friends in thesis room for making thesis work period fun and also for creating friendly atmosphere to discuss problems related to thesis.

I am grateful to Mr. Tandin Gyeltshen, Engineer, BPC and many other friends form BPC and DGPC for providing the data that required for my thesis. It would not have been possible to complete my thesis work on time without their help.

Last but not least I would like to express my deepest appreciation to my parents and my dear wife who gave me endless support and inspiration to continue with this study at abroad.

Protection Coordination study in Bhutan Transmission Network

Thinley Gyeltshen Department of Energy and Environment Division of Electric Power Engineering Chalmers University of Technology Göteborg, Sweden 2009

ABSTRACT

Since the effects of an unreliable power system transmission can be widespread and affect millions of people, as well as damage to life and equipment, therefore one of the most important requirements of protection system is to isolate and disconnect faulted parts of the system selectively and quickly. This purpose can be achieved by proper coordination of protective devices. One of the aims of this thesis was to make a general guideline from which proper coordination of transmission system protection can be developed in Bhutan network.

This thesis proposes a review of coordination of distance relays for transmission lines of a real network in Bhutan for study. The transmission network in Bhutan has under gone drastic change for past decade, due to growth of generation capacity with the target to generate 10,000MW by 2020 and target to achieve "Electricity for All by 2013". It has become very much necessary to study and analyze the protection system of existing transmission network.

This thesis also discusses the important of directional earth fault relay (67N) during the high impedance fault and its coordination. This thesis highlights the factor that effecting the operation of distance relay.

This study is carried out using Computer Aided Protection Engineering (CAPE) software by Electrocon International Incorporated, Michigan, USA.

Keywords: Coordination, CAPE software, Distance relays, Directional Erath fault, Power system, Protection system, Transmission network,

Table of Contents

Ackn	lowle	edgments	i
ABS	ГRAC	Ti	i
List o	of Fig	gures	7
List o	of Ta	blesvi	i
Abbr	evia	tionsvii	i
1.	INT	RODUCTION1	L
1.1.	Bac	kground and Motivation:	1
1.2.	Obj	ective of the thesis:	2
1.3.	Pro	cedure and Outcome:	2
1.4.	Sco	pe and Limitations:	2
1.5.	Out	line of the Thesis:	2
2.	DES	CRIPTION OF NETWORK UNDER STUDY	ŀ
2.1.	Ove	erview of Power Sector in Bhutan:	1
2.2.	Pro	blem Definition:	5
2.3.	Sys	tem Data:	5
2.4.	Exi	sting Protection System:	7
3.	STU	DY ASPECT FOR PROTECTION COORDINATION STUDIES	5
	10		
3.1.	Pro	tective Objective:)
3.2.	Rela	ay Characteristics:)
3.3.	Typ	bes of Distance relay:	1
3.	3.1.	Quadrilaterals relay Characteristics1	1
3.	3.2.	MHO Characteristics:	2
3.4.	Sys	tem Impedance:1	3
3.5.	Coo	ordination Study:	3
3.6.	Prir	nary and back-up Protection:	1
3.7.	Fac	tors Affecting Distance Relay Operation:	5
3.	7.1.	Fault Resistance:	5
3.	7.2.	Infeed Effect:	7
3.	7.3.	Branching-off effect:	3
3.	7.4.	Load Encroachment:	3

4.]	ME	THODOLOGY FOR PROTECTION COORDINATION	
S	ГUD	IES		20
	4.1.	Dis	cussion on classical receipt for zone settings:	20
	4.2.	Wh	at should detect distance protection?	21
	4.2	.1.	Fault Detection Criteria:	21
	4.2	.2.	Resistive reach of Quadrilateral characteristic:	22
	4.3.	Sce	narios to simulate:	23
	4.3	.1.	Scenarios to test sensitivity:	23
	4.3	.2.	Scenarios to test coordination and Sequential operation:	26
	4.4.	Imp	portance of Directional Earth Fault Relay (67N)	29
	4.4	.1.	Setting Criteria of 67N:	29
	4.4	.2	Simulation showing Importance of Directional earth fault relay:	30
	4.4	.2.	Scenarios to Test Coordination of 67N:	32
5.]	DIS	CUSSION ON SIMULATION RESULT	38
	5.1.	Intr	oduction:	38
	5.2.	Sof	tware used for the Thesis:	38
	5.3	Dis	tance Relay setting and coordination study:	38
	5.4.	Cal	culated/Proposal Impedance for Zones Settings:	39
	5.5.	Dire	ectional Earth fault relay (67N):	43
6.		COI	NCLUSION, RECOMMENDATIONS & FUTURE WORK	44
	6.1.	Cor	nclusion:	44
	6.2.	Rec	commendations:	44
	6.3.	Fut	ure Works:	45
R	EFE	RE	NCES:	46
A	PPE	NDI	IX:	47
	APPE	ENDE	X [A]: Single line diagram of Bhutan network	47
			X [B]: Fault MVA and Current for various system Buses during Three phases fault:	
	APPE	ENDE	X [C]: Simulation result for sensitivity test of Distance relay setting:	50
	APPE	ENDE	X [D]: Simulation result for Coordination & Sequential Operation of distance relay:	56
			X [E]: Simulation showing the important of 67N:	
	APPE	ENDE	X [F]: Simulation Result 67N (CDD21) Coordination:	64
	APPE	ENDE	X [G]: Propose relay setting details:	72

List of Figures

Figure 1: Organizational structure of Bhutan Power Sector
Figure 2: Network showing which is considered for study
Figure3: Distance relay protection zones9
Figure 4: Distance relay Characteristics [7]12
Figure 5: Mho impedance characteristic12
Figure 6: Impedance measured by distance relay13
Figure 7: Three Zone time-distance Characteristics14
Figure 8: Local back-up protection15
Figure 9: Tripping logic circuit of breaker back-up protection
Figure 10: Remote back-up protection16
Figure 11: Effect of fault Resistance17
Figure 12: Infeed effect on distance relays18
Figure 13: Branching-off effect18
Figure 14: Load Encroachment in Mho relays.[11]19
Figure 15: Visualizing the fault resistance (R_F) component [14]
Figure 16: Distance relay Characteristics of ABB (Fig. a) and Siemens (Fig. b) relay [15]
Figure 17: Relay under simulation24
Figure 18: Simulation result24
Figure 19: Characteristics plot of existing setting
Figure 20: Network showing the fault location and relay operation
Figure 21: Characteristics plot of new setting
Figure 22: Network showing the fault location and relay operation
Figure 23: Characteristics of combined IDMT and definite time over current relays30
Figure 24: Network showing the fault location and non operation of distance relay 31

Figure 25: Network showing the fault location and 67N operation
Figure 26: Final Network after fault is cleared and showing the relay operation
Figure 27: Network showing the fault location and relay operations for existing setting.
Figure 28: Normal inverse time-characteristic of 67N for existing setting
Figure 29: Network showing the fault location and relay operations for new setting 33
Figure 30: Normal inverse time-characteristic of 67N for new setting
Figure 31: Network showing 67N relay on 220kV feeder looking towards 220kV Rurichu Substation
Figure 32: Normal inverse time-characteristic of 67N for existing setting
Figure 33: Normal inverse time-characteristic of 67N for new setting with DEF
Figure 34: Relay Characteristics plot for existing setting:
Figure 35: System Simulation for existing relay setting SL-G fault at line end
Figure 36: System simulation for existing relay setting SL-G fault at 90% line
Figure 37: Relay Characteristics plot for new setting:41
Figure 38: System simulation for new relay setting SL-G fault at end of line

List of Tables

Table 1: Transmission line length	5
Table 2: Existing installed capacity power	5
Table 3: Status of future Hydropower development in Bhutan	5
Table 4 : Transmission line data considered for study	6
Table 5: Protections and CT & PT ratios considered in study	7
Table 6: Protection scheme used in Transmission line:	8
Table 7: Simulation Result for 5 ohms fault resistance 2	25
Table 8: Simulation Result for 5 ohms fault resistance for new settings:	26
Table 9: Fault current comparison between two locations:	35
Table 10: Operating of Relay during the fault	36
Table 11: Proposed Zone settings to be followed in Bhutan network	38
Table 12: Proposed New relay setting	12
Table 13: Proposed relay setting of relay type CDD21 for 132kV Eastern Grid	13
Table 14: Direction Earth fault relay setting:	13

Abbreviations

A	Ampere
ACSR	Aluminum Conductor Steel Reinforced
BPC	Bhutan Power Corporation Limited
BHP	Basochu Hydro Plant
СНР	Chukha Hydro Plant
CAPE	Computer Aided Protection Engineering
CB	Circuit Breaker
CT	Current Transformer
CTR	Current Transformer Ratio
DGPC	Druk Green Power Corporation
DL-G	Double phase to Ground
DoP	Department of Power
DPR	Detail Project Report
EF	Earth Fault
EHV	Extra High Voltage
GOI	Government of India
HV	High Voltage
IG	Inter Government
IDMT	Inverse Definite Minimum Time
JV	Joint Venture
kV	Kilo Volt
kVA	Kilo Volt Ampere
LV	Low Voltage
MVA	Mega Volt Ampere
MVAR	Mega Volt Ampere Reactive
MW	Mega Watt
NHPC	National Hydroelectric Power Corporation
NERC	North American Reliability Council
00	Over Current
PMP	Power System Master Plan
P.u	Per unit
PT	Potential Transformer
PTR	Potential Transformer Ratio
SL-G	Single Phase to ground
SJVNL	Satluj Jal Vidyut Nigam Limited
THDC	Tehri Hydro Development Corporation Limited
THP	Tala Hydro Plant
VT	Voltage Transformer
WAPCOS	Water And Power Consultancy Services
67N	Directional Earth fault relay

1. INTRODUCTION

1.1. Background and Motivation:

Transmission Network System in Bhutan is still in its infancy. As of now, it has two separate isolated grids called the Western and the Eastern grids. The Western grid comprises both 66 kV, 220 kV and 400kV transmission lines which are powered by the Basochu (64 MW), Chhukha (360MW) and the Tala (1020 MW) generation plants.

Currently, in the Eastern grid is powered by only the Kurichhu (60 MW) generation plant. The transmission voltage level of Eastern grid is 132 kV which is isolated from the Western grid. But somehow they are interconnected through Indian grid at different interface points to export any surplus electric energy to India. However, inter-connection of eastern and western grid and formation of national grid is in the process, due to upcoming hydro project namely Punatsangchu, Mangdechu, Dagachu, and etc. The construction of 220 kV interlinks between Tsirang and Jigmeling in Gelephu to connect the two grids is also under progress and schedule to complete by July 2011.[1]

With power demand growing rapidly at the rate of 8% per annum, Bhutan Power System Master Plan (PSMP) projected to increase the generating capacity and government committed to provide affordable and reliable electricity to all citizens by 2013 and committed to generate 10,000MW by 2020.[2] Due to which, power system network in Bhutan has been expanding rapidly for past few years, as a result the protection system setting/coordination were disturb and which may result abnormal tripping of the system.

Bhutan Power Corporation Limited (BPC), who is solely responsible for transmission of electric power in Bhutan. As of now BPC had no dedicate division who looks after the protection system study and also no software to carry out the study. Now BPC is high time to have dedicated units which will carry out the time to time protection coordination and system study in order to keep updating the protection system in network. The side benefits of coordination study the interrupting ratings of all protective equipment, conductors, and switches are checked for adequacy.

The objective of this thesis is to develop a standardized protection setting on Bhutan network and also maintain proper data base for the protection equipments. This will help to develop a maximum protection of equipment, transmission lines and a consistence statistical frame work for evaluating year-to-year variation of transmission service quality and stability performance indicators.

This thesis report is a small work based on the requirement, the power system analysis and protective device coordination for the safe and reliable power supply of the Bhutan network. In Bhutan, the generating stations are located at different parts of the country, which are interconnected by transmission networks and ultimately connected to Indian grid. In fact, this thesis work is not able to coordinate all protective devices for whole interconnected network, due to limited information from the Indian grid side and Generation Company. Therefore, this thesis report is a just starting and which is the beginning protection Bhutan future. for the study on network in

1.2. Objective of the thesis:

Power system network in Bhutan has been expanding rapidly for past few years and it is expected to further increase the network after completing the ongoing and upcoming transmission and hydro power projects, which is shown in table 3. With the rapid growth of transmission and distribution system, the power system is becoming more and more complex. Besides that, there is always small changes in loading conditions occur continuously in the power system. The power system must adjust to these changing conditions and continue to operate. Therefore, sometimes it has to upgrade the equipment and system protective devices. As the system becomes more complex, it is very important to carry out the detail power system studies and coordinate the protection system properly to maintain better system reliability. So far no coordination study is carried out after commission of the protective devices.

Therefore, the main goal of this thesis is to make general guidelines for protection coordination from which the transmission protection system will be improved in Bhutan. The main objectives are setting calculation, factors effecting the distance relay operations, recommendation for protection coordination proposal, coordination of existing systems, coordination curves, and justification of protective devices proposed for line such as earth fault protection (67N), tabulation of Coordination results and Analysis and recommendations.

1.3. Procedure and Outcome:

The load flow study and short circuit analysis has carried out with the help of CAPE Software and also with the DIgSLIENT. The result from the both software are same and it is attached in Appendix (B). For the protection relay coordination study is carried out only in CAPE, since CAPE software is the tools for protection engineering. While doing the simulation, the manufacturer's guidelines also followed for distance relay settings [3] [4] [5]. The outcome of the thesis has tabulated and written in the form of report. Recommendations were made for the best protection of the grid network in Bhutan. A general report provided to improve the protection system as well as to review the coordination of the system by implementing this information.

1.4. Scope and Limitations:

- The power system model is built in CAPE with data available from Bhutan. The hydro projects under construction and the transmission line expansion being carried out at present is not considered in the proposed model.
- Due to insufficient data and time constraint, only 75% of network where able to study the distance relay coordination. In addition to that, transformer protection and Bus bar protection were also not taken into consideration for the study. The network that is selected for study is the network which has under gone many changes in the system for past years.
- Equivalent relay model has to used in some of the line since relay used were very old model and its model are not available in the CAPE.

1.5. Outline of the Thesis:

Chapter 1 describes the background and motivation for carrying out thesis and its objectives. It also describes the scope and limitation of the work.

Chapter 2 presents the existing network protection system, describes the problem definition of the existing protection system for which the coordination study needs to be

done and also present the data that are required for this study. In this, present the overview of Power sector in Bhutan.

Chapter 3, describe the study aspects of the protection coordination studies and its objectives. It also high lights the factor affecting the performance of the relay operation and also describe the relay characteristics that are used in the existing network.

Chapter 4 discusses about methodology of the classical receipt for zone setting of the protection relay, coordination and same is verified by simulation. And also discuss the important of Directional earth fault relay (67N), definite time stage function and it is coordination.

Chapter 5 present the software used for the simulation and discussion of the simulation result. It also present the justifications of proposed settings and presenting the new setting table for distance relay and Directional earth fault relay.

Chapter 6, Conclusion and recommendations on the findings are made and suggestions for future studies on the work are proposed.

2. DESCRIPTION OF NETWORK UNDER STUDY

2.1. Overview of Power Sector in Bhutan:

Power sector in Bhutan plays a vital role in small economy like Bhutan. Electricity is major contributor towards total revenue in the country. This chapter gives an overview of hydro power plants and Transmission & Distribution system in Bhutan. The Organizational Structure of the Bhutan Power Sector is presented in figure (1).



Figure 1: Organizational structure of Bhutan Power Sector

Prior to July 2002, all electricity transmission and distribution encompassing the development and construction of Mini and Micro Hydropower Plants in Bhutan was catered by the Department of Power (DoP) under the Ministry of Trade and Industry. From July 2002, the erstwhile Department of Power under the Ministry of Trade and Industry was bifurcated into three separate entities namely Bhutan Power Corporation Limited (BPC)- a public utility, Department of Energy - A government department responsible for policy, planning and coordination activities for the energy sector and Bhutan Electricity Authority – Regulatory body under the Department of Energy. It can be seen from figure 1 that, BPC is the soul entity that purchases power from different generation station and sale it to its domestic customer and as well as export to India. BPC has mandate to provide electricity to all by 2013. At present, the transmission network covers 13 Dzongkhags with about 822.86 km of line length (66, 132, 220 & 400kV) and 22 substations (619.5MVA). The transmission grid of Bhutan consists of Western Grid and Eastern Grid which are yet to be interconnected. However, interconnection work are under progress and by end of year 2011 Bhutan will have one national grid. Detail of these lines is shown in table 1.

SI.No.	Particulars	Line length (km)	Remarks
1	400 kV Line	74	
2	220 kV (Double Circuit)	71.28	
3	220 kV (Single Circuit)	126.62	
4	132 kV	304.29	
5	66 kV	246.67	
	Total	822.86	

Table 1: Transmission line	length
----------------------------	--------

Source [1]

From 1st January 2008, all the generating stations are merged under single entity called 'Druk Green Power Corporation', which will be a government owned body, responsible for hydropower generation in the kingdom. The total installed capacity of DGPC currently stands at 1480MW. The detail installed capacity of hydro power plant is shown in the table 2. DGPC has mandate to increase the installed capacity to 10,000MW by 2020. In line with this, the Royal Government, has approved the following ten (10) hydroelectric projects in its 14th session of the Lhengye Zhungtshog held on 14th August 2008 are shown table3 below.[2]

Table 2: Existing installed capacity power

SI. No.	Name of Hydro Power Plant	Installed capacity in MW	Remarks
1	Chhukha	336	
2	Basochu (Upper & Lower)	64	
3	Kurichu	60	
4	Tala	1020	
5	Mini/Micro	8.062	Under BPC
	Total	1488.062	

Source [6]

Table 3: Status of future Hydropower development in	n Bhutan
---	----------

SI No	Project	Capacity (MW)	DPR Schedule	Indian PSU identified for DPR	Construction Schedule	Funding Model	
1	Dagachu	114	Under constr	ruction	•	•	
2	Punatsangchu-I	1200	Done	Completed	2008-2015	IG	
3	Punatsangchu-II	1000	2006-2008	WAPCOS	2009-2016	IG	
4	Mangdechu	720	2006-2008	NHPC	2009-2017	IG	
5	Sunkosh Storage(Main Dam)	4000	2009-2011	NHPC	2011-2020	IG	
6	Kuri Gongri	1800	2009-2011	NHPC	2012-2019	IG	
7	Amochu Storage	620	2009-2011	NTPC	2012-2018	IG	
8	Wangchuk Storage	900	2009	NTPC	2010-2017	JV	
9	Bunakha Storage	180	2009	SJVNL	2010-2016	JV	
10	Kholongchu	486	2009-2010	THDC	2011-2017	JV	
11	Chamkharchu-I	670	2009-2010	SJVNL	2011-2017	JV	
	Total	11,690	By 2020 10,000MW				

Source [2][6]

Note: Inter-Govt. (IG) - project will be between GOI & RGOB with 40% grant and 60% Loan from GOI. Joint venture (JV) - Public sector from Bhutan and India will participate.

2.2. Problem Definition:

In Bhutan, the transmission grid voltage levels are 400kV, 220kV, 132kV and 66kV. The single line diagram of the network is shown in appendix (A). The transmission lines are overhead lines with ACSR conductors and are supported on steel tower. All power transformers and equipment are out door type. The system mainly protected with distance relay, directional earth fault relay, over current relay, circuit breakers, etc.

With such a network, the problem is how to maintain a safe, reliable and efficient energy supply by ensuring that transmission line and equipment are well protected in the event of fault. Protection system must recognize the existence of a fault and initiate circuit breaker operation to disconnect faulted line of the system selectively and quickly. The actions required assure minimum disruption of electrical services and limit damage in the faulted equipment. This can only be achieved if the protective devices are well coordinated. Although, the existing network was coordinated when it was installing but it should be reviewed of coordination as causes described in chapter 1. The equipment has been upgraded in the network due to growing demand of power where in most cases it was not planned with protective device coordination in mind. Therefore, there is loss of selectivity between upstream and downstream protective devices. In this circumstance, this study needs to be done for proper coordination.

Another problem is that, most of the directional earth fault relay functions are disable. It is very important to enable this function. During the high impedance fault the distance relay will not detect the fault, so in that case the direction earth fault will sense the fault and cleared the fault. The detail simulation reports are discussed in the Chapter 4.

2.3. System Data:

The system data used to build the network are taking from the data base available with the BPC and DGPC. Power plants under considered for the study are shown in table 2. The positive and zero sequence impedance of the conductors are very necessary for the distance protection setting of transmission lines. The impedances of conductor which used in the existing network are given in table 4. Type of relay used in the network and CT & PT ratios are given in table 5.

SI.No	Description	Conductor parameter			
1	Conductor type	Twin Moose	Zebra	Panther	Dog
2	Voltage level (KV)	400	220	132	66
3	Positive sequence R (ohms/km)	0.0264	0.07	0.1695	0.217
4	Positive sequence X (ohms/km/)	0.309	0.4063	0.4298	0.417
5	Zero sequence R (ohms/km)	0.263	0.187	0.4227	0.99
6	Zero sequence X (ohms/km)	1.1325	1.087	1.35	2.27
7	Nominal Rating (MVA)	924.22	224.82	90.31	27.32

Table 4 : Transmission line data considered for study.

Source [1]

SI. No	Elemer	nt Nme	Type of Prote	ection Relay	CT Ratio(A	PT Ratio (kV/V)		
	From	То	From	То	From	То	(,	
1	КНР	Nangkhor	SPHM 101	EPAC3000	500/1	300/1	132/110	
2	Nagkhor	Nanglam	EPAC300	EPAC300	300/1	300/1	132/110	
3	Nanglam	Tingtibi	EPAC300	EPAC300	300/1	300/1	132/110	
4	Tingtibi	Gelephu	EPAC300	EPAC300	300/1	300/1	132/110	
5	Gelephu	Salakati	SPHM 101	*****	300/1	****	132/110	
6	Chukha	Chumdo	7SA6xxx	REL511	300/1	300/1	66/110	
7	Chumdo	Jemina	REL511	PD532	300/1	200/1	66/110	
8	Jemina	Olakha	PD532	7SA6xxx	200/1	600/1	66/110	
9	Olakha	Semtokha	7SA6xxx	REL511	600/1	300/1	66/110	
10	Semtokha	Lobeysa	REL511	REL511	300/1	300/1	66/110	
11	Lobeysa	Rurichu	REL511	REL511	300/1	400/1	66/110	
12	Rurichu	Basochu	REL511	REL511	400/1	400/1	66/110	
13	Rurichu	Semtokha	REL511	REL511	300/1	300/1	220/110	
14	Semtokha	Chukha	REL511	REL511	300/1	600/1	220/110	
15	Chukha	Malbesa	REL511	REL511	600/1	800/1	220/110	
16	Malbesa	Birpara	REL511	****	800/1	****	220/110	
17	Chukha	Birpara	REL511	****	800/1	****	220/110	
18	Tala	Malbesa	7SA6xxx	REL512	2000/1	1000/1	400/110	
19	Tala	Binaguri	7SA6xxx	****	2000/1	* * * * *	400/110	
20	Malbesa	Binaguri	REL512	****	1000/1	****	400/110	

Table 5: Protections and	СТ	& PT ratios	considered in study
	• •	a a	

Note: ***** Data not available, since the substation is under Indian grid.

Above are the lists of the line that we have considered for the study of relay coordination. Details simulations were discuss and presented in chapter 4 and 5.

2.4. Existing Protection System:

As describe in the Chapter 1, transmission network of Bhutan is divided into two, i.e Western and Eastern grid and ultimately connected with Indian grid. For my study, the networks are divided into three parts. They are 132kV system in Eastern grid and western grid into two parts, 220kV and 66kV system. The details networks are show in the figure below:



Figure 2: Network showing which is considered for study

Norms of protection system followed in Bhutan are shown in table below: The scheme used is Non-switched scheme for all transmission line

SI.No	Votage level		Protection Scheme	Remarks		
1		kV	Main-I Numerical Distance Scheme	Main-II, same		
	Line		Main-II Numerical Distance Scheme	manufacture but with different model.		
-	220					
2	220	kV	Main-I Numerical Distance Scheme	Main-II, same		
	Line		Main-II Numerical Distance Scheme	manufacture but with		
				different model		
3	132 kV		Main-I Numerical Distance Scheme	Backup protection is of		
	Line		Backup Protection: 3 Nos dir. IDMT OC	Electromechanical relay.		
			relays and 1 No. dir. EF relay.			
4	66 kV Line		Main-I Numerical Distance Scheme	Backup protection is of		
			Backup Protection: 3 Nos dir. IDMT OC	Electromechanical relay		
			relays and 1 No. dir. EF relay.			

Table 6: Protection scheme used in Transmission line:

The distance relays used are of numerical, which is of five stepped distance protection zone, with zone 1, 2&3 in forward direction and Zone 4 as reverse direction, Zone 5 is not used. The Classical method is used for Zone setting of the existing distance relay. Figure 3 below shows set up with the following criteria:

- \checkmark Zone 1: Forward direction, 80% of the line length with instantaneous trip.
- ✓ Zone 2: Forward direction, 100% of the protective line length, plus at least 20% of the shortest adjacent line and with time delay of 0.4 sec.
- ✓ Zone 3: Forward direction, 100% of the protective line length, plus 100% of adjacent line, plus 20% of the third line and with time delay of 0.8 sec.
- \checkmark Zone 4: Reverse direction, 10% of the protective line length, with time delay of 1 sec.



Figure 3: Distance relay protection zones

While going through the setting for different lines, it was found that some zone settings are under reach and some are over reach. This is checked and simulated with the help of CAPE software. The details simulation is discuss in chapter 4 and 5. And also it was found that Directional earth fault function is not used as backup protection. It is very important to used, since the distance protection will not detect the high impedance fault. There are two type of scheme in distance protection system are:

- ✓ Non-switched scheme: This scheme is faster and more accurate but is costly. There are 6 starters, 3 for phase faults and 3 for ground faults. There will be independent measuring units for both phase faults and earth fault for each phase, for all three zones, totaling to 18 units.
- ✓ Switched scheme: This scheme is relatively slow in operation and the risk of total scheme failure in the event of failure of the only one measuring unit available. In this scheme only one measuring unit will be used for all types of faults. This single measuring unit is switched to the correct fault loop impedance by switching-in the respective voltages and currents by the starter.

3. STUDY ASPECT FOR PROTECTION COORDINATION STUDIES

3.1.Protective Objective:

The objective of the protection is to quickly isolate section from both ends so that the rest of the system can function satisfactorily. More fundamental, however, is that the power system should operate in a safe manner at all the times. But no matter how well designed, faults will always occur on a power system and these faults may represent a risk to life or property. The main four functional requirements of the relays are: [7]

- ✓ Reliability
- ✓ Selectivity
- ✓ Sensitivity
- ✓ Speed

Reliability:

The most important requisite of protective relay is reliability since they supervise the circuit for a long time before a fault occurs, if a fault then occurs, the relays must respond instantly and correctly. When protective relays fails to function properly, the allied mitigation features are largely ineffective. Therefore, it is essential that protective-relaying equipment be inherently reliable, and that its application, installation, and maintenance be such as to assure that its maximum capabilities will be realized.

Selectivity:

The relay must be able to discriminate (select) between those conditions for which prompt operation is required and those for which no operation, or time delayed operation is required. The property of selectivity tripping is also called 'discrimination' and is achieved by two general methods.

- ✓ Time Grading
- ✓ Unit system

Sensitivity:

The relaying equipment must be sufficiently sensitive so that it operates reliably when required under the actual conditions that produces least operating tendency. This is a term frequently used when referring to the minimum operating level (current, voltage, power etc.) of relays or complete protection scheme. The relay or scheme is said to be sensitive, if the parameters of the primary operating setting is low.

Speed:

The function of protection systems is to isolate faults on the power system as rapidly as possible. Therefore, the relay must operate at the required speed. It should neither be too slow which may result in damage to the equipment nor should it be too fast which may result in undesired operation.

3.2. Relay Characteristics:

The distance relays need to have a characteristic that will ensure correct operation when a short circuit fault occurs within the zone of protection and at the same time avoid miss-operation under no-fault conditions. The protected transmission line is in the impedance plane with the area of arc resistance that has to be covered by the protection element.

Some of the numerical relays measure the absolute fault impedance and then determine whether operation is required according to impedance boundaries defined on the R/X diagram. Whereas, in traditional distance relays they compare the measure voltage with a replica voltage derived from the fault current and the zone impedance setting to determine whether the fault is within zone or out-of-zone. The common types compare either the relative amplitude or phase of two input quantities to obtain operating characteristics that are either straight lines or circles when plotted on an R/X diagram.

3.3. Types of Distance relay:

The conventional distance relay uses three distance measuring units. Distance relays can be classified into phase relay and ground relays. Phase relays are used to protect the transmission line against phase fault (3phase, L-L) and the ground relays are used to protect against ground faults (SL-G, DL-G). The most important and versatile family of relays is the distance relay group. It includes the following major types-

- ✓ Impedance relays
- ✓ Reactance relays
- ✓ MHO relays
- ✓ Angle impedance relays
- ✓ Quadrilateral relays etc.

At present distance relay used in Bhutan network are mostly of ABB make and few of AREVA and SIEMENS make. The relay characteristics used in the Bhutan network are of Quadrilateral and MHO. Therefore, the characteristics of MHO relay and Quadrilateral relay are discussed only in this study.

3.3.1. Quadrilaterals relay Characteristics.

A quadrilateral relay is suitable for long lines and as well as for short lines. This relay characteristics would allow the ground fault resistive reach to be increased or decreased independently of the forward reach and source impedance behind relay so that the required ground fault resistive coverage can be achieved. It is therefore provides better resistive coverage than any mho-type characteristic for short lines. To avoid excessive errors in the zone reach accuracy, it is common to impose a maximum resistive reach in terms of the zone impedance reach. Obviously, the characteristic needs to have a shape and be wide enough to provide this coverage. At the same time the characteristic should have a shape and be narrow enough so that the dynamically changing load impedance does not enter inside the characteristic, which will result in undesired tripping of the protected line at the time fault. The characteristics are shown in the figure below:



Figure 4: Distance relay Characteristics [7]

3.3.2. MHO Characteristics:

The MHO characteristic, as seen on the impedance polar diagram, is a circle whose diameter is the relay impedance setting vector, such that the characteristic passes through the origin of the impedance diagram, as shown in Figure 5. This demonstrates that the impedance element is inherently directional and such that it will operate only for faults in the forward direction along the protective line. Therefore MHO relay is directional. In this impedance reach varies with fault angle. As the line to be protected is made up of resistance and inductance, its fault angle will be dependent upon the relative values of R and X at the system operating frequency. Therefore, Relay Characteristic Angle (RCA) φ is set less than the line angle, so that it is possible to accept a small amount of fault resistance without causing under-reach.



Figure 5: Mho impedance characteristic

3.4. System Impedance:

The impedance of the power system is divided into two parts. Firstly, the impedance behind the relaying point, including the generators, feeders, transformers, etc., forms the source impedance. The second part is the impedance to the fault in front of the relaying point, which is governed by the geometrical arrangement, size, shape, spacing and material of the conductors. Generally, this impedance data are provided by manufacturers. Both of this impedance must be known to determine the faults levels and setting of the relays. However, all the setting calculations are in terms of secondary impedance. Therefore, the relation between secondary and primary are presented below:



Figure 6: Impedance measured by distance relay

$$Z_{R} = \frac{V_{R}}{I_{R}} = \frac{V_{FP} \times \frac{V_{2}}{V_{1}}}{I_{FP} \times \frac{I_{2}}{I_{1}}}$$
$$Z_{R} = \frac{V_{FP}}{I_{FP}} \times \frac{\frac{I_{1}}{I_{2}}}{\frac{V_{1}}{V_{2}}} = Z_{P} \times \frac{C.T.ratio}{V.T.Ratio} = \text{Secondary impedance } (Z_{S})$$

Where, Z_R is the relay impedance, V_{FP} is the fault voltage at the fault point, I_{FP} is the fault current at the fault point, Z_p is the positive sequence impedance of the line on primary side and Z_S is the secondary positive sequence impedance of the line on secondary side.

3.5. Coordination Study:

The basic role of the protection scheme is to sense faults and isolate these faults by opening all incoming current paths. However, the protection scheme must be selective so that only faulted element is removed i.e. isolated. Therefore, a coordination study maximizes power system selectivity by isolating faults to the nearest protective device, as well as helping to avoid mal-operations. The other upstream devices must be designed to operate in sequence to provide back-up protection, if any device fails to respond, this is called selective coordination. One of the main topics of concern protection engineers is the proper coordination behavior of different relay units so as to avoid relay maloperation. In fact, for proper coordination, it is better to follow the relay manual guides which are provided by manufacturers. A new or revised coordination study should be made when the available short-circuit current from the power supply is increased; when new large loads are added or existing equipment is replaced with larger equipment; or when protective devices are upgraded. Typical three zone time-distance characteristics of distance relay is shown in sketch below figure 7.



Figure 7: Three Zone time-distance Characteristics

3.6. Primary and back-up Protection:

A power system is divided into various zones for its protection. There is a suitable protective scheme for each zone; it is the duty of the primary relays of that zone to isolate the faulty element. The primary protection is the first line to defense. If the primary protection fails to operate, there is a back-up protective scheme to clear the fault as a second line to defense.

The causes of failures of primary protection could be due to failure of the CT/VT or relay, or failure of the circuit breaker. The back-up protection should also preferably be located at a place different from where the primary protection is located. Further, the back-up protection must wait for the primary protection to operate, before using the trip command to its associated circuit breakers. In other words, the operating time of the back-up protection must be delayed by an appropriate amount over that of the primary protection. Thus the operating time of the backup protection should be equal to the operating time of primary protection plus the operating time of the primary circuit breaker. In general, there are three types of back-up Protection scheme in power system protection: [7]

- ✓ Local back-up protection
- ✓ Breaker back-up protection
- ✓ Remote back-up protection

Local back-up protection:

This is kind of a local back-up in which an additional relay is provided for back-up protection. This is achieved by protection which detect an un-cleared primary system fault at its own location and which then trip its own circuit breakers, shown in figure 8 below:





Breaker back-up protection:

This is also kind of a local back- up is necessary for a bus bar system where a number of circuit breakers are connected to it. When a protective relay operates in response to a fault but the circuit breaker fails to trip, the fault is treated as a bus bar fault. In such a situation, it becomes necessary that all other circuit breakers on that bus bar should trip. Figure 9 below shows tripping logic circuit of Breaker back-up protection.



Figure 9: Tripping logic circuit of breaker back-up protection.

Remote back-up Protection:

When back-up relays are located at a neighboring station, they backup the entire primary protective scheme which includes the relay, circuit breaker, PT, CT and other elements. This is provided by protection that detects an un-cleared primary system fault at a remote location and then issue a local trip command e.g. the second or third zones of distance relay. It is the cheapest and simplest form of back-up protection and is widely used back-up protection for transmission line.



Figure 10: Remote back-up protection

3.7. Factors Affecting Distance Relay Operation:

Proper settings of protective relays are essential for the reliable operation of electrical power systems, during both fault and normal system operating conditions. The setting of distance relays should ensure that they are not going to operate when not required (security) and will operate to trip when necessary (dependability). The ideal relay operating characteristics can also be influenced by parasitic phenomena, such as CT saturation. The main factors that affecting the operation of distance relay are discuss as below:

3.7.1. Fault Resistance:

Ground resistance and Arc resistance are the two components of the fault resistance. If a flashover from phase to phase or phase to ground occurs, an arc resistance is introduced into the fault path which is appreciable at higher voltages. The arc resistance is added to the impedance of the line and hence, the resultant impedance which is seen by distance relays is increased. In case of ground faults, the earth resistance is also introduced into the fault path. The arc resistance is treated as pure resistance in series with the line impedance, where reactive component is negligible. The arc resistance has little effect on accuracy of zone-1 unit as it operates instantaneously before the arc can stretch appreciably length. Therefore, arc resistance will have greater impact on accuracy of backup zones (time delayed) as the arc stretches appreciably.

Figure 11 shows the effect of fault resistance on distance relay impedance reach. For a fault at the point F, the actual line impedance up to fault is Zf but due to the presence of the fault resistance, the impedance measured by the relay is (Zf + R). That is why, this shows that arc resistance causes under-reach and relay fails to operate.

Due to the physical nature of an arc, there is a non-linear relationship between arc voltage and arc current, which result in a non-linear resistance. Using the empirical formula derived by A.R Van C. Warrington, [9] the approximate value of arc resistance can be assessed as:



Figure 11: Effect of fault Resistance.

3.7.2. Infeed Effect:

The effect of intermediate current source between relay location and fault point is termed as infeed effect. This will cause a distance relay to under reach. Distance relay is said to under reach when the impedance presented to it is apparently greater than the impedance to the fault. Consider the sketch indicated in figure 12. A fault at \mathbf{F} on the line **BC** is at a distance of **Z1+Z2** for the relay at station **A**. But when current **I2** flows from bus **D**, the impedance to the fault as seen by the relay at **A** is:

$$Z_1 + \frac{I_1 + I_2}{I_1} \times X \times Z_2$$

So for relay balance:

$$Z_1 + Z_2 = Z_1 + \frac{(I_1 + I_2)}{I_1} \times X \times Z_2$$

Therefore the effective reach is

$$Z_1 + \left(\frac{I_1}{I_1 + I_2}\right) Z_2$$

Thus the fault is seen by the relay A as farther than what it really is, it is clear from above equations that relay will under reach due to the infeed effect. The effect of infeed becomes more pronounced with more interconnections at station B.[9]



Figure 12: Infeed effect on distance relays.

3.7.3. Branching-off effect:

This effect will cause distance relay to over reach. It is said over reach, when the apparent impedance presented to it is less than the impedance to the fault. In the figure 13, a fault at **F** is at the distance of Z_1+Z_2 for the relay at station **A**. But when current I_1 gets distributed as $I_2 \& I_3$ at station **B**, the impedance to fault seen by the relay at station **A** will be $(Z_1 + I_3/I_1 * Z_2)$ which is less than (Z_1+Z_2) .

Then the fault is seen by the relay as nearer than what it really is i.e. distance relay overreaches due to branching-off effect. This overreaching tendency will cause the relay to lose its selectivity.



Figure 13: Branching-off effect.

3.7.4. Load Encroachment:

While protecting long lines the necessary reach may be so large that the minimum service impedance (or load impedance) falls within the region of the starter. This would result in tripping without there being any fault. The two conditions i.e. operation at heavy load and short circuit differ by virtue of phase angle between voltage and current. For the load

impedance, the phase angle will be within +30 to -30 Deg. While during short circuits, the fault impedance has a phase angle of 60 to 80 deg. (i.e. line angle).[9]

Load encroachment problem is more pronounced in case of under impedance starters and gets lessened in case of mho, elliptical, lens etc and type of starters. Relays with suitable characteristic on R-X diagram have to be carefully chosen to protect long and heavily loaded lines, and this becomes easily possible with microprocessor based numerical relays. Figure 14 below shows the load encroachment in mho relay.

The load resistance vectors area can be represented as the distance protection tripping zone encroachment. This load area settings calculation until now had been made with the determination of the maximal angle of the load vectors (φ_{max}) and the minimal load resistance value (R_{Load}).

$$Z_{Load} = \frac{U_{min}}{\sqrt{3} X I_{Max}}, \qquad R_{Load} = Z_{load} \times \cos \varphi_{Load}$$



Figure 14: Load Encroachment in Mho relays.[11]

4. METHODOLOGY FOR PROTECTION COORDINATION STUDIES

4.1. Discussion on classical receipt for zone settings:

In Chapter 3, we have presented the definition of distance protection and the idea of parameters (settings) that define operating zones. However, the objective of protection coordination study is to determine those parameters so that protections fulfill the objective of protection systems. It has been mention in Chapter 3 that the objective of protection systems: speed, selectivity, sensitivity and reliability are dependent on the protection coordination study, while speed and reliability are more in the field of manufacturers. In order to achieve the objectives mentioned above, the classical receipt to set the zone settings are:

Zone 1: The protected zone of the first unit is called the Zone 1 of protection. It is high speed unit and is used for the primary protection of the protected line. Instantaneous Zone 1 might not be applicable if the line is too short (less than 10 miles) because short lines have very low impedance that can be within the measurement error of distance relay. Due to this, miss-coordination may occur. Generally the reach of Zone 1 is set between 80% - 90% of the whole length of the protected line with no time delay. The reach of this zone is not 100 % the length of the protected line to avoid the errors due to measurement.[12]

Zone 2: The Zone 2 reach is set to protect the remaining portion of the line left unprotected by Zone 1 and provides an adequate margin. At the same time it provides backup protection for the bus bar in the remote end substation. To coordinate with zone 1 of the relays at remote bus, time delays of 20-30 cycles are typically added to Zone 2 setting, although temporization may vary depending on the circumstances. Usually Zone 2 is set to 120% - 150% of primary protective line. This provides sufficient margin to account for measurement errors.

It is very important to note that Zone 2 also provides back up protection to a part of the adjacent line. In this case, Zone 2 reach can be set one of the following two ways:

- ✓ Zone 2 is set to reach 50% of the shortest back up line provided that $Z_P + 0.5Z_B >$ 1.2 Z_P ; where Z_P and Z_B are the positive sequence impedance of primary and shortest back up line respectively.
- ✓ If $Z_P + 0.5Z_B < 1.2 Z_P$, which means that the shortest back up line is too short. In such case Zone 2 is set to 1.2 Z_P

However, in both cases, the calculated reach needs to be checked to ensure that it does not cover beyond the Zone 1 of the next line section.

Zone 3: Even though the transmission line is fully protected with Zone 1 and Zone 2 relays, a third forward reaching zone is often employed.[13] This Zone 3 reach is calculated to act as a backup for Zone 2 and may be applied as remote backup for relay or station failures at the remote terminal. Generally Zone 3 reach is set to 200% of the line impedance, Z_L with a time delay of 60 cycles or the reach of Zone 3 is set to 100% of Z_L plus 120% of next longest adjacent line Z_{L2} . This is shown in mathematical form as below:

Zone $3 = Z_{L} + 1.2Z_{L2}$

The resistive reach of the Zone 3 setting should check that the reach does not limit the load carrying capability of the line. Therefore it is very important to verify that the Zone 3 relay does not trip on load under extreme conditions.

4.2. What should detect distance protection?

It is wrongly believed that distance protection should cover every fault inside its operating zone, that is, zone 1 should operate for every fault up to the 80% of the line, zone 2 up to 120 of the line and so on. The setting of distance relays should ensure that they are not going to operate when not required (security) and will operate to trip when necessary (dependability). However, the influence of infeed effect makes that distance relay to be less sensitive, especially in the case of resistive faults. It is wrong to say, for instance, that if the distance relay is 50 ohms, then it should detect faults with resistances of 50 ohms, as was shown in chapter 3. Due to infeed effect, distance relay's performance is poor for high impedance faults. Therefore, a reasonable goal has to be set for the faults detection. This will be shown in the following sections.

4.2.1. Fault Detection Criteria:

As discuss earlier that operating of distance relay has so many factors and one of the most critical factors is fault resistance, since it has different components of fault resistance for overhead transmission line faults. The different components of fault resistance are the tower structure, insulator chains, ground wire and the different impedances to the flow of fault current and are shown in figure 15.[14]



Figure 15: Visualizing the fault resistance (R_F) component [14]

Depending on the nature of the fault, the fault resistance can vary. If the fault is due to a flashover, then the fault resistance would be the sum of the tower footing resistance plus the arc resistance. According to data collected from the transmission utility of Bhutan, the maximum tower footing resistance is around 3.5 ohms.

If the fault is due to an object, like a tree, the fault resistance would be the sum of the tower footing resistance and the object resistance, which is typically very high, in the range of 20 to 50 ohms or even more. It is not practical, then, to set the fault detection criteria so that the relay would detect high impedance due to the reasons stated before. Therefore, in this thesis, the fault detection criteria would be so that the distance relay would detect at least 5 ohm in the whole length of the protected line. This corresponds to faults due to flashovers. As will be shown later, directional earth over current will take the task of detecting high impedance faults. Simulations will be performing to verify that this criterion is fulfilled and simulation result is shown in appendix [C].

4.2.2. Resistive reach of Quadrilateral characteristic:

For adjusting the resistive reach for different zones, as we have seen, a general criterion to select a value for all the different zones of distance protection, allow to establish the coordination between the tripping time of each zone and achieving selectivity. The resistive reach setting for Zone 3 is very important. There is a compromise in the selection of the resistive reach; a large resistive reach would allow the distance relay to detect more faults. However, the larger the resistive reach the more possible that distance relay operates due to load conditions. The figure below shows the quadrilateral characteristic plot for ABB and Siemens make distance relay: [5] [4]



Figure 16: Distance relay Characteristics of ABB (Fig. a) and Siemens (Fig. b) relay [15]

The resistive reach does not have to limit the load carrying capability of the line. The resistive ranges are calculated as 45% of the minimum load impedance load. As per the NERC, [11] this ensure that distance relay won't operate when the line is overloaded to 150% of rated thermal current during emergency operation for maximum of 15 minute and considering that the voltage is around 0.85 p.u and the power factor may be greater than 30 degrees, this is considering for the extreme system conditions. During this condition of overloading the system, the protections system should not operate. The minimum load impedance is calculated through the following expression:

$$Z_L = \frac{V_L^2}{MVA_{Rated}}$$

Where: Z_L :Minimum Load impedance in ohms V_L :Rated Line to Line Voltage in KV MVA_{Rated} :Thermal rating of Line in MVA

Therefore maximum resistance reach is calculated as: $R_{Reach} = 0.45 \text{ x } Z_L$ By using these criteria maximum resistive reach is calculated and checked with simulations for all distance relay and results presented in chapter 5.

4.3. Scenarios to simulate:

The simulation was carried by using CAPE software in different scenarios. For the initial simulation setting provided by the BPC is used and then be derived the new setting values where ever required by taking the consideration of the above mention factors. The simulations were carried out in different scenarios as below:

4.3.1. Scenarios to test sensitivity:

The sensitivity simulation of the distance relay is carried out on two different scenarios, i.e during the peak season and lean seasons of generation. In this simulation effects of infeed on distance relay is also checked and verify the maximum fault resistance that the distance relay could detect the fault. The simulation for one relay is discussed below and details report of the simulation is shown in appendix [C]. The same simulation procedures are followed for rest of the relays.

Case 1: When all the Generator are running with full capacity and without infeed:

The simulation is carried out for the exsiting relay setting with single phase to gorund fault at fault resistance of 5 ohms. In the simulation report as shown below we can see that Zone 1 is detecting the fault of 20%, Zone 2 only 60% and Zone 3 up to 80% of the protective line. Even though zone 1 is set to cove the 80% of the protective line and Zone 2 to cove the 100% of the protective line plus the 20% of the next line. The main reason for non operating of the distance relay is due to low setting coverage of resistive reach for the respective Zone and also due to infeed affect. Hence, the reach of the distance relay varies as a function of fault current distribution, as well as fault location.



Figure 17: Relay under simulation



Relay Name: EFAC_Nang-R Style: EF01111BCD Model: EFAC 3136-3 Scheme: Demo Sanufacturer: ALSTOM Substation: Nangkhor Local ZOP: *132kV Nang	536	
elay Common Taps: Tap Name	Tap Setting	Find Original Setting
1105 ZON T2	0.4	0.4
1106 ZON Z3	4.16	4.16
1107 ZON T3	0.8	0.8
1108 ZON Dir, Z3	Forwards	Forwards
1109 ZON Z4	4.16	4,16
110A ZON T4	0.8	0,8
110B ZON 25	1.04	1.04
110C ZON T5	0.8	0,8
110D ZON T>>	0	0
110E ZON T>	0	0
110F ZON Ph/God RZ1	10	10
1110 ZON Ph Ph RZ1	10	10
1111 ZON RLim Z2	16	16
1112 ZON RLim Z3	25	25
1113 ZON RLim Starter	25	25
1114 ZON I>> Acti.	Yes	Yes
		<u>ب</u>

.

Simulation Result for 5 ohms fault resistance

		42 Bu 45 Bu	s_132Na s_132KB							
Curve: 1 Su Re				111BCD	HF DIST "21	" Zone	1			
Curve: 2 Su Re				111BCD	HF DIST "22	" Zone	2			
Curve: 3 Su Re				111BCD	HF DIST "Z3	" Zone	3			
Fault					CUR					
Location			APP	. IMP	OPER. SEC	APP	. IMP	OPER. SEC	APP	. IMP
Close in	A.	0.030	11.30	-8.2	0.430	11.30	-8.2	0.830	11.30	-8.2
0.100	B	0.030	12.78	-2.6	0.430	12.78	-2.6	0.830	12.70	-2.6
0.200	с	0.030	14.30	1.6	0.430	14.30	1.6	0.830	14.30	1.6
0.300	D	99999.898	16.00	4.8	0.430	16.08	4.8	0.830	16.00	4.8
0.400	E	99999.898	17.80	7.2	0.430	17.80	7.2	0.830	17.80	7.2
0.500	F	99999.898	19.78	8.9	0.430	19.78	8.9	0.830	19.70	8.9
0.600	G	99999.898	21.68	10.2	0.430	21.68	10.2	0.830	21.60	10.2
0.700	н	99999.898	23.70	10.9	99999.898	23.70	10.9	0.830	23.70	10.9
0.800	т	99999.898	25.80	11.3	99999.898	25.88	11.3	0.830	25.80	11.3
0.900			28.08	11.3	99999.898	28.08	11.3	99999.898	28.00	11.3
			17.88	59.4	0.430	17.88	59.4	0.830	17.88	59.4
Line_End	K	99999.898	11.06	35.4	0.400	30.38			21100	

Figure 18: Simulation result

Case 2: When all the Generator are running with full capacity and with infeed:

In this case the simulation is run with the infeed from the Deothang feeder. Now we can see that the zone coverage of the all the Zones has improved as compare to the case 1. This is happen due to the current contribution from the Deothang feeder, due to which short circuit current flowing through the relay has increased. From this simulation result, we can see that there is no need to increase the resistive reach setting. But we should keep in mind that relay should operate even for the worst system condition. And also resistive reach can increase unless that the reach does not limit the load carrying capability of the line. The simulation result is presented below:

Table 7: Simulation Result for 5 ohms fault resistance

Study Line: From: Nangkhor 42 Bus_132Nang To: Kurichhu 45 Bus_132KHP											
Curve: 1 Substation: Nangkhor Relay: EPAC_Nang-KHP EP311111BCDHF DIST "Z1" Zone 1											
Curve: 2 Substation: Nangkhor Relay: EPAC_Nang-KHP EP311111BCDHF DIST "Z2" Zone 2											
Curve: 3 Substation: Nangkhor Relay: EPAC_Nang-KHP EP311111BCDHF DIST "Z3" Zone 3											
Fault FaultCURVE 1CURVE 2CURVE 3 Location Code OPER. SEC APP. IMP OPER. SEC APP. IMP OPER. SEC APP. IMP											
		0.030				6.120	-4.3	0.830	6.120	-4.3	
0.100	в	0.030	7.240	7.0	0.430	7.240	7.0	0.830	7.240	7.0	
					0.430	8.590			8.590		
0.300	D	0.030	10.10	20.2	0.430	10.10	20.2	0.830	10.10	20.2	
0.400	E	0.030	11.70		0.430	11.70	23.9	0.830	11.70	23.9	
0.500	F	0.030	13.40		0.430	13.40	26.5	0.830	13.40	26.5	
0.600	G	0.030	15.10	28.2	0.430	15.10	28.2	0.830	15.10	28.2	
0.700		0.030	16.90	29.3		16.90	29.3		-		
	I	0.030		29.8		18.70			18.70	29.8	
		99999.898	-			-			-		
_		99999.898									
Remote_Bus	L	99999.898	22.50	29.6	0.430	22.50	29.6	0.830	22.50	29.6	

The resistive reach setting for all the Zones are increased and check that the reach does not limit the load carrying capability of the line. The maximum resistive reach of the line is calculated and compare with the new resistive reach setting of the zone and same is presented in tabular form in Chapter 5. The simulation result after new setting is presented below:
0.500

0.600

0.700

0.800

Remote_Bus L

0.900

Line End

F

G

Н

I

J

15.10

20.60

28.2

29.9

59.4

0.830 13.40 26.5 99999.898 13.40 26.5

0.830 16.90 29.3 99999.898 16.90 29.3

0.830 18.70 29.8 99999.898 18.70 29.8

0.830 22.50 29.6 99999.898 22.50 29.6

99999.898

99999.898

59.4 99999.898 17.80

0.830 15.10 28.2

0.830 20.60 29.9

0.830 17.80

Table 8: Simulation Result for 5 ohms fault resistance for new settings:

Study Li	ne:												
From:	Nangkhor	42 Bu	s_{132Na}	ng									
To:	Kurichhu	. 45 Bu	s_132KH	<u>,</u>									
Curve: 1		on: Nangkho PAC_Nang-KH		EP31	L1111BCDHF I)IST "Z1	." Zone	: 1					
Curve: 2		on: Nangkho											
	Relay: E	PAC_Nang-KH	P(New 2	EP31	L1111BCDHF I	DIST "Z2	" Zone	2					
Curve: 3		on: Nangkho PAC_Nang-KH		EP31	L1111BCDHF I)IST "Z3	" Zone	: 3					
Curve: 4		on: Nangkho PAC_Nang-KH		EP31	L1111BCDHF I)IST "ZS	5″ Zone	: 5					
Fault	Fault	CIIR	VE 1		CIIE	RVE 2		CIIR	VE 3		CUR	VE 4	
											OPER. SEC		
Close_	in A	0.030					-4.3	0.830	6.120	-4.3	99999.898	6.120	-4.3
0.1	00 B	0.030	7.240	7.0	0.430	7.240	7.0	0.830	7.240	7.0	99999.898	7.240	7.0
0.2	00 C	0.030	8.590	14.8	0.430	8.590	14.8	0.830	8.590	14.8	99999.898	8.590	14.8
0.3	00 D	0.030	10.10	20.2	0.430	10.10	20.2	0.830	10.10	20.2	99999.898	10.10	20.2
0.4	00 E	0.030	11.70	23.9	0.430	11.70	23.9	0.830	11.70	23.9	99999.898	11.70	23.9

4.3.2. Scenarios to test coordination and Sequential operation:

59.4

0.030 13.40 26.5

0.030 15.10 28.2

0.030 16.90 29.3

0.030 18.70 29.8

0.030 20.60 29.9

0.030 22.50 29.6

K 99999.898 17.80

Proper settings of protective relays are essential for the reliable operation of electrical power systems, during both fault and normal system operating conditions. Therefore, it is very much necessary to check the analysis of the ideal operation of protective relays and the identification of incorrect relay operation by Computer simulations. With the help of CAPE software Zone reach of the existing distance relay setting were checked by using Coordination Graphics module and cross checked by running the System Simulation module. The primary intension in conducting this study was to verify the accuracy of the line protection setting. The following study were carried out in three different voltage level, 66kV, 132kV and 220kV as shown in the figure 2 of Chapter 2. The total of 35 numbers of distance relays where carried out system simulation and coordination checking of the relay setting. For the discussion we have selected 66kV Jemina feeder distance relay which is located at Olakha substation. The simulations are carried out in two cases, in case 1 with the existing setting and case 2 with the new settings. The same procedure is followed to all the 35 relays. The simulation result for only 66kV Jemina feeder is presented only, due to large number of pages that attaching the whole simulations results would have meant.

0.430 13.40 26.5

0.430 15.10 28.2

0.430 16.90 29.3

0.430 18.70 29.8

0.430 22.50 29.6

29.9

59.4

0.430 20.60

0.430 17.80

Case 1: Characteristics diagram of Distance relay of the existing setting is plotted by using Coordination Graphics, in which reach of the relay is checked and is shown in the figure 19. The operation of the relay is checked by using the System Simulation, shown in figure 20. During the system simulation the network is consider in normal operation and created single line to ground (SL-G) on the 66kV line between Jemina and Chumdo, which is next adjacent line. During this fault it is found that Distance relay of Olakha end

operated in Zone 1 tripping. This shows that improper setting/coordination of distance relay, which can be also seen from the coordination graphics characteristics. The details simulation report is shown in appendix [D].



Figure 19: Characteristics plot of existing setting



Figure 20: Network showing the fault location and relay operation.

From the above system simulation we can see that there is problem with the setting of the relays. As per the logic, when there is fault between the bus 9bus_66Jemina and 6Bus_66SChum, the relay at 10Bus_66Olakha operate in Zone 1 as shown in figure above. Therefore necessary relay setting changes are made, with help of Coordination Graphics and same is simulated by using System simulation.

Case 2: Characteristics diagram of Distance relay of the new setting is plotted by using Coordination Graphics, after making necessary change in setting is shown in figure 21. Before coming to conclusion the same is checked with the System simulation, by creating the same fault on same line as above, as shown in figure 22.



Figure 21: Characteristics plot of new setting



Figure 22: Network showing the fault location and relay operation.

Now we see that both the relay at fault end operated and the relay at Olakha are not operated. The details simulation reports are shown in appendix [D] and new setting table are presented in tabular from in chapter 5.

4.4. Importance of Directional Earth Fault Relay (67N)

In the previous paragraphs it has been shown that the performance of the distance protection is badly affected when the fault resistance increases due to infeed effect. Because of this reason, we have determined that the objective of distance protection is mainly to detect low impedance faults. High impedance faults mainly are the case when an object approaches to the transmission line. Those faults are typically single-phase to ground faults or two-phase to ground faults.

In this type of faults, zero sequence current appears and therefore, directional earth fault relays can be used. Directional earth fault is very sensitive to earth faults and can easily detect those high impedance faults that distance protection can't detect.

Therefore, directional earth fault function is very important since it detects faults that distance relay can't detect. The considerations to set the directional earth fault function are presented in the following sections.

4.4.1. Setting Criteria of 67N:

The coordination of directional earth fault protection is very important to get selective relay operation. Therefore, appropriate criteria to set directional earth fault relays should be determined by the protection engineer as part of an overall protection coordination study. Settings that are too sensitive or too fast may result in non-selective relay operation. If the settings are not sensitive enough, the relay may not detect some faults and this could cause excessive damage in the protection transmission line.

The typical settings for earth fault relays are 30% to 40% of the full load current or minimum earth fault current on the part of the system being protected. However, in Bhutan we usually set earth fault relays 10% to 20% and time delay less than Zone 2 timing. Some time we have to use definite time stage protection setting combined with normal directional earth fault protection for fast tripping. This is use when the fault close to relay, current is much greater than when the fault at the end of the line. This makes a reduction in the tripping time at high fault levels possible, as shown in figure below. The setting of the definite time stage is usually done as:

$$I_{\text{Setting}} = \frac{I_A + I_B}{2},$$

Where: I_A : Fault current when the fault near to relay.

 I_B : Fault current when the fault at end of line.



Figure 23: Characteristics of combined IDMT and definite time over current relays.

4.4.2 Simulation showing Importance of Directional earth fault relay:

The single to ground fault apply at 50% of transmission line with a fault resistance of 20 ohms. The simulation was done in two cases, case 1 without Direction earth fault protection and case 2 with directional Earth fault protection. Details simulation results are presented in appendix [E].

Case 1: Without directional earth fault protection:

Fault is not cleared after 1.0 cycles 0.020 seconds



Figure 24: Network showing the fault location and non operation of distance relay.

From above simulation we see that all the distance relay were not detecting the high impedance fault, therefore it is not operated and the fault is not able to clear.





Figure 25: Network showing the fault location and 67N operation.

Now the 67N at source end has operated first and still the fault is not clear in first cycle of simulation. Now the network has become radial feeder and distance zone 1 has operated because Zone 1 is set with no time delay, as u can see in the figure below.



Figure 26: Final Network after fault is cleared and showing the relay operation.

4.4.2. Scenarios to Test Coordination of 67N:

As we all know that basic function of the protection is to detect the faults and isolate these faults by giving trip command to circuit breaker to disconnect the faulty line from the system. However, the protection scheme must be selective so that only faulted element is isolated. Therefore, proper coordination between different relays units is very much necessary, in order to fulfill the selectivity, which is one of the principle function requirement of protection system. Simulation of relay coordination is presented below and details simulation reports are shown in appendix [F].



Figure 27: Network showing the fault location and relay operations for existing setting.



CURRENT in P.Amps

Figure 28: Normal inverse time-characteristic of 67N for existing setting.

Curve	Current	Operating	Source/Total line
	—		rom"42Bus_132Nang"
	t temporary bus 99 e on "42 Bus 132Na	_	-
Fault: D			

(+ seq	(+ seq SIR)		Operating	Source/Total line			
	Primary A	A/Pickup	Seconds				
1	428.02	7.13	1.746	2.77 @ 8.0			
2	136.28	2.27	2.158	6.44 @ 3.4			
3	136.28	2.27	0.846	1.66 @ 6.3			
4	136.28	2.27	1.270	2.09 @ 8.4			

From the above simulation and from the plots it is found that the relay R3 is not properly coordinate with the other relays. Due to which R3 is operating faster than the other relays, even the fault is not in it jurisdictions. This is also seen in the system simulation too. We see that R3 is operating at this fault (As shown in the network). Therefore, necessary coordination were done by using the coordination graphics module, this is one of main advantage of using CAPE software. In this we don't need to recalculate whole setting, what we can do here just play with the characteristics curve by dragging up and down till we get proper coordination result. After making necessary relay coordination with help of CAPE software, system simulation and relay characteristics plot are shown below and details simulation result in appendix [F].



Figure 29: Network showing the fault location and relay operations for new setting



CURRENT in P.Amps

Figure 30: Normal inverse time-characteristic of 67N for new setting Fault: D

SLG_A_R30 at temporary bus 999001 Bus_132Nang (NEWBUS1)
Midline node on "42 Bus_132Nang" to "45 Bus_132KHP" Ckt 1
"999001Bus_132Nang" (NEWBUS1) distant0.501
from"42Bus 132Nang"

Curve SIR)	Current	Operating Source/Total line (+ seq					
	Primary A	A/Pickup	Seconds				
1	428.02	7.13	0.544	2.77 @ 8.0			
(operat	ed)						
2	136.28	2.27	1.150	6.44 @ 3.4			
3	136.28	2.27	0.992	1.66 @ 6.3			
4	136.28	2.27	0.846	2.09 @ 8.4			

From the above figure now we can see it is properly coordinated and it is found that only R1 has operated. In this coordination we have not make use of definite time stage function of 67N, since the short circuit current close to relay and at far end are almost equal. The simulation below shows coordination of 67N using the definite time stage function in 220kv feeders:



Figure 31: Network showing 67N relay on 220kV feeder looking towards 220kV Rurichu Substation.

Table below show the fault current comparison between the fault close to relay and far away from relay and its operating time.

Relays	Fault current in Amps		Operatii	ng time in sec	Remarks
	Close to relay	Far from relay	T-close	T- Far away	-
R1	2201.49	1318	0.220	0.260	Fault current difference between
R2	7484.61	2349.92	0.256	0.296	close to relay and far end is large.
R3	4784.30	1578.26	0.298	0.448	Therefore it takes longer time to isolate the fault.

Table 9: Fault current comparison between two locations:



Figure 32: Normal inverse time-characteristic of 67N for existing setting.

From the above coordination graph and simulation, it is found that there is large difference in fault current for the two fault location as mention above. If we only use the normal inverse time characteristics of 67N, then the fault close to relay will take more time to isolate the fault. In this case we need to make use of definite time stage function, so that it will isolate the fault of high current with no time delay. The setting are done as explain above. Now new coordination graph of 67N relay with definite time stage are presented below:

Relays	Fault current in Amps		Operati	ng time in sec	Remarks
	Close to relay	Far from relay	T-close	T- Far away	
R1	2234.99	1318.5	0.196	0.220	DEF operated
R2	8002.08	2312.98	0.216	0.230	instantaneously when
R3	4954.75	1481.26	0.236	0.326	the fault current is
DEF1	2234.99	1318.5	0.020	Infinite	close to relay.
DEF2	8002.08	2312.98	0.020	Infinite	
DEF3	4954.75	1481.26	0.020	Infinite	

Table 10: Operating of Relay during the fault.

Note: R1, R2 & R3 are of normal inverse time earth fault relay (67N), DEF is definite time earth fault relay



Figure 33: Normal inverse time-characteristic of 67N for new setting with DEF.

5. DISCUSSION ON SIMULATION RESULT

5.1. Introduction:

In this thesis, I have concentrate more on to check the existing relay setting and its coordination. I have tried my best to check as many relay as possible, but due to limited data and time, not able to cover all relay setting checking and coordination. However, I have covered the maximum number of relays and most critical line. Before, I start the relay coordination study; load flow and short circuit study were carried out. The load flow and short circuit study has performed mainly for coordination study of the existing network, in addition to calculate the present load flow and fault levels. Therefore, in this project work, the main discussion has done about coordination analysis of distance relays.

5.2. Software used for the Thesis:

The software tool used is Computer Aided Protection Engineering (CAPE) software. This software is mainly for protection of high voltage transmission system and distribution system within power utilities. This software offers a wide spectrum of applications of protection coordination. In this thesis mainly used One-Line diagram, Short circuit, Power Flow, Coordination Graphics and System Simulator modules.[16]

5.3 Distance Relay setting and coordination study:

In this thesis, detail distance relay coordination and setting were checked with the help of Coordination Graphic and System Simulation module. During the simulation, it was found most of the relays were not coordinate properly and same is rectified. The relay which required major change in setting is presented in table 12 below:

During the collecting the data from Bhutan, it was noted that the method used for zone reach settings are followed different from region to region, base on the relay manufacturer, and the reach are not uniform. Therefore, in order to ensure proper coordination between distance relays in power system, it is customary to choose relay ohms setting should be same. And also in line with Protection and Control Philosophies and IEEE standards for Protection of EHV Lines, the strategy proposed to BPC to adopt distance relay settings is summarized as follows: [13]

SI.No	Zones	Impedance Reach	Time	Direction
1	Zone-1	80% of ZL	Instantaneous	Forward
2	Zone-2	100% of ZL + 40-50% of ZSL	0.3 to 0.4 seconds	Forward
3	Zone-3	100% of ZL + 120% of ZSL	0.6 to 0.8 seconds	Forward
4	Zone-4	100% of ZL + 120% of ZLL	0.9 to 1.5 seconds	Forward/Reversed

Table 11: Proposed Zone settings to be followed in Bhutan	network
---	---------

Where: ZL = Positive sequence impedance of line to be protected.

ZSL = Positive sequence impedance of adjacent shortest line.

ZLL =Positive sequence impedance of adjacent longest line.

The zone-1 reach is limited to 80% of ZL to provide a reasonable margin against a possible overreach due to errors in CTs, PTs, relay measurement, line parameters, etc. The zone-2 reach is set to cover up to 40-50% of adjoining line so that this will definitely cover the balance 20% of main line (after zone-1 reach) and provides backup to adjoining line relay. Zone-2 setting shall be not less than 120% of ZL in order to ensure definite coverage of 100% of main line. The zone-3 & zone-4 reaches will be suitably set to provide backup for relays on adjacent lines with proper time gradation.

5.4. Calculated/Proposal Impedance for Zones Settings:

Distance relay setting were revised for most of the relay, after it has simulated and checked by Coordination graph. The maximum resistive reach is also checked with the resistive reach limit, which is calculated using equation that is discuss in chapter 4. The detail discussion and justification of existing and proposed settings are explain for one feeder only, since it is follow same principle and also to explain all will be a lot in report. Therefore, I have taken 132 kV Nangkhor feeder for the discussion and justification. For rest of the relays final relay setting are presented in the table 12 below.

132kV Nangkhor feeder at Kurichu Switchyard:

The existing zone settings of the relays for this feeder are not accurate as can be seen from the relay characteristics plot. The zone settings are all set under reach, which means it is not even covering the full line length. This has been justified by system simulation and it is shown below:



Figure 34: Relay Characteristics plot for existing setting:



During the single phase to ground fault at end of the line, we can see that relay at other end is not able to clear the fault.

Figure 35: System Simulation for existing relay setting SL-G fault at line end

With existing setting, relay was able to detect the single phase to ground fault at 90% of line length only, which is also by Zone 3 protection, system simulation is shown below. As per the setting criteria Zone 1 should cove 80% of line and zone 2 & 3 are to be cover 100% of line plus backup protection for the next line.



Figure 36: System simulation for existing relay setting SL-G fault at 90% line.

From the above characteristics and system simulation we can clearly see that Zone 1 protection has not set to 80% of the line length and also Zone 2 and Zone 3 are just within the protective line. Due to which it is not able to clear the fault which is close to line end. Therefore new settings have made and relay characteristics plot and system simulation shown below:



Figure 37: Relay Characteristics plot for new setting:



Figure 38: System simulation for new relay setting SL-G fault at end of line.

Name of	Name of feeder	Zone s	setting in	Ω (Sec.)	Resistive Reach in Ω (Sec)			Resistive	
S/S		Z1	Z2	Z3	R1	R2	R3	reach limit in Ω(Sec	
Nangkhor	132kV KHP	3.33	4.74	5.31	16.79	19.21	20.07	()	
Nangkiloi	132kV Nanglam	2.86	4.47	14.67	10.0	15.0	15.0		
Nanglam	132kV Nangkhor	3.14	5.18	6.6	7.5	9.2	9.65		
Nungium	132kV Tingtibi	7.81	12.2	17.09	4.6	5	5	21.71	
Tingtibi	132kV Nanglam	7.68	11.39	13.44	7.53	8.36	8.36		
11190101	132kV Gelephu	4.11	8.15	10.92	4	5	5		
Gelephu	132kV Tingtibi	4.11	7.1	12.95	4	5	5		
Chukha	66kV Chumdo	6.95	10.06	16.54	13.21	14.64	17.64		
Chumdo	66kV Chukha	6.96	11.16	13.63	22.23	26.79	26.84		
	66kV Jemina	2.19	4.93	7.13	12.71	14.30	15.03	35.9	
	66kV Paro	4.52	12.41	14.12	5.00	5.63	6.51		
	66kV Haa	6.28	17.27	19.63	5.20	6.38	6.38		
Jemina	66kV Chumdo	1.58	4.07	7.60	4.65	4.65	4.65	23.91	
Jennia	66kV Olakha	2.27	3.80	4.94	6.50	7.38	7.90		
Olakha	66kV Jemina	7.03	11.17	14.14	11.90	14.96	16.92	71.80	
Clairia	66kV Semtokha	3.01	4.76	7.38	24.47	24.47	24.47	, 100	
Semtokha	66kV Olakha	0.39	1.37		4.00	4.00			
Sentokia	66kV D/ling	2.22	3.34	5.01	3.85	3.85	3.85	35.90	
Lobeysa	66kV Semtokha	4.89	7.34	12.22	19.52	25.00	25.00	55.50	
	66kV Rurichu	3.97	8.41	12.36	12.5	14.00	15.00		
Rurichu	66kV Basochu	0.75	2.50		6.01	7.80		47.89	
Basochu	66kV Rurich	0.75	5.24		12.80	14.18			
Semtokha	220kV Rurichu	1.52	2.72	3.54	6.73	7.5	7.5	15.52	
Rurichu	220kV Semto	1.85	4.18	5.51	8.52	8.97	9.00	13.32	

Table 12: Proposed New relay setting

Where time setting for Zone 1 is instantaneous and Zone 2 & 3 is set 0.4 sec and 0.8 sec respectively.

5.5. Directional Earth fault relay (67N):

During the simulation of high impedance fault, it was found that distance relay are not able to detect the fault, for this protection directional earth fault (67N) is required to isolate the fault. Detail simulation and coordination are presented in chapter 4. In this chapter the summarized relay setting of 67N is shown in the table below.

				New se	tting		
Name of S/S	Feeder	Relay type	CT ratio	Pickup in Amps(sec)	Time dial	Characteristics	
КНР	132kV Nangkhor	CDD 21	500/1A	0.20	0.224		
Neveliker	132kV KHP	CDD 21	300/1A	0.20	0.156		
Nangkhor	132kV Nanglam	CDD 21	300/1A	0.20	0.199		
Negalaria	132kV Nangkhor	CDD 21	300/1A	0.20	0.136		
Nanglam	132kV Tingtibi	CDD 21	300/1A	0.20	0.164	Normal Inverse	
Tinatihi	132kV Nanglam	CDD 21	300/1A	0.20	0.117	inverse	
Tingtibi	132kV Gelephu	CDD 21	300/1A	0.20	0.129	·	
Calanhu	132kV Tingtibi CDD 2		300/1A	0.20	0.10		
Gelephu	132kV Salakati 🕴	CDD 21	300/1A	0.20	0.10		

Table 13: Proposed relay setting of relay type CDD21 for 132kV Eastern Grid.

Note: Arrow above indicates the direction of coordination and colour represent the relay group for coordination.

The table below shows the setting of directional earth fault and definite time earth fault function that is inside the REL511 relay: The detail discussion is explained in chapter 4 and settings are presented in appendix [G].

		REL 511 type		Relay set	ting	
Substation	Feeder	function enable	CT ratio	Pickup in Amps(sec)	Time dial	Characteristics
Semtokha	220kV Rurichu	OC IN>_TEF	300/1A	0.20	0.10	Standard Inverse
		Inst. OC IN1>	300/1A	4.46		Definite
Chukha	220kV	OC IN>_TEF	600/1A	0.15	0.11	Standard Inverse
	Semtokha	Inst. OC IN1>	600/1A	3.87		Definite
Malbesa	220kV	OC IN>_TEF	800/1A	0.15	0.12	Standard Inverse
	Chukha	Inst. OC IN1>	800/1A	3.95		Definite

Table 14: Direction Earth fault relay setting:

6. CONCLUSION, RECOMMENDATIONS & FUTURE WORK

6.1. Conclusion:

The distance relay coordination study was carried out for a real transmission network of Bhutan, but due to lack of data and information not able to carry out for whole network, anyhow tried to cover the maximum relay and most critical transmission line. Since these activities are highly data intensive and it has always been a tedious and time consuming task to collect this data among many users.

In this thesis, the traditional rules applied for choosing the reach of distance relays are reviewed. These rules fail to cover all the different cases of network topology. False operations are experienced in the case of high impedance faults and with high in-feed. They are also experience in the case of a long line followed by a short one or vice versa. All these studies were carried out by using the Coordination Graphics and same is verified by System Simulation.

Therefore this thesis presents an adaptive setting procedure to avoid the above mentioned problems. Base on this, the proposed new zone settings and justifications are discussed and tabulated in chapter 4&5 for those relays. In cases where long feeders are followed by short feeders, it has taken care to ensure discrimination between the zones of back-up protection on adjacent feeders. The operating time settings of zone 2 and zone 3 are made long enough to be selective with zone 2 and zone 3 of adjacent line section and basic principle are considered to ensure selectivity for proper coordination.

In this thesis also discuss the important of directional earth fault relay as back up protection and its coordination. Details simulation were presented in chapter 4 and proposed new setting were shown in table 13 & 14.

After scrutinizing, it is recommended that, existing relay settings should be set according to proposed settings thereby it would be possible to get optimum protection by using the existing relay. This study proposes the proper coordination of relay thereby relay maloperation will not be happened during the fault. It will be increased the availability of power in terms of reliability of the network.

6.2. **Recommendations:**

Based on the simulation studies conduct it is recommended to do the following for existing network of protection system:

- ✓ It is very important to follow the same standard zone reach setting criteria for the entire region in Bhutan Network. The proposed zone reach setting standard is presented in table 11.
- ✓ Most of the distances relay setting need to revised, however the author highly recommended to changing the distance relay setting of 132kV Nangkhor feeder at Kurichu end, 66kV Jemina feeder and 66kV Semtokha feeder at Olakha Substation.
- ✓ Directional earth fault relay (67N), type CDD21 are used as back up protection in 132kV Eastern grid is not properly coordinate and need to revise the setting. The proposed new settings are presented in Chapter 5.

- ✓ Directional earth fault function in distance relay are not enable, the author highly recommend enabling the function in the view of its usefulness at detecting the high impedance fault and backup protection at no additional cost. Detail justifications are explained in Chapter 4. Before enabling, it is recommended to do proper coordination study. Due to limited of time, author could carry out coordination for the 220kV feeder only and settings are presented in table 14.
- ✓ New relay settings are prepared for the 66kV feeders which are soon going to install new numerical distance relay of M/S ABB make. Details setting are presented in Appendix [G].
- ✓ All transmission system in Bhutan should have optimal level of protective device with trained personals especially on its applications and its related studies (short circuit and coordination study). It is very high time to have protection software tools for analysis of protection system and system studies. As the Bhutan network is expanding and soon going to have National Grid.

6.3. **Future Works:**

In order to have a complete picture and a deeper understanding of the protection system of the Bhutan network, it would be appropriate to continue the study as and when the system is upgraded. The study carry out in this thesis can serve as a basis for the establishment of protection system study in Bhutan network. In this thesis, the author had just carried out the study of distance relay setting coordination and important of directional earth fault relay. This is first steep for study of protection system of Bhutan network.

The model built in this thesis can be upgraded by including the upcoming generating plants and new transmission lines. Thus, there is huge scope of future work and improvement of protection system in Bhutan network. Therefore, following studies were recommended for future works.

- ✓ Continue the distance relay coordination studies and directional earth fault relay for the remaining relays.
- ✓ Coordination study of directional over current relay, which are used as back up protection for transmission lines.
- ✓ Study the important of carrier inter trip scheme in distance protection system, such as Permissive Under-reach Transfer Tripping (PUP) Scheme and Permissive Over-Reach Transfer tripping (POP) Scheme.
- ✓ Study the important of mutual impedance affect on distance relay operation for parallel transmission line.
- ✓ Carry out the complete relay coordination for HV substation. Which will includes 11kV out going feeder, transformer feeder, bus bar and HV incomer.

REFERENCES:

- [1] <u>http://www.bpc.bt/</u>
- [2] <u>http://www.dgpc.bt/</u>
- [3] "Numerical Distance relay Commissioning and Maintenance Guide", for EPAC 3100/EPAC 3500, Version V5E & V6, Alstom, 1999.
- [4] "SIPROTEC distance Protection", Manual for the 7SA6, V 4.2, Siemens.
- [5] "Application Manual REL 511*2.3, Line distance protection terminal", ABB, 2001
- [6] Royal Government of Bhutan, Department of Energy, "Power Progress Data", 2005
- [7] C. Rusell Mason, "The Art & Science of Protection relaying".
- [8] Alstom T&D Energy Automation and Information, "Network Protection and Automation Guide" first edition July 2002, Le Sextant, 3 avenue Andre Malraux, 92300 Levallois-Perret. France
- [9] Walter. A. Elmore, "Protective Relaying Theory and application", Second Edition
- [10] M.V.S Birinchi and C.Venkateswarlu, *"Handbook for Protection Engineers"*, Engineering Staff College of India.
- [11] "Increase Line Loadability by Enabling Load Encroachment Functions of Digital Relays", North American Electric Reliability Council (NERC) Prepared by the System Protection and Control Task Force, December 7, 2005.
- [12] M.I. Gilany, Bahaa Eldeen Hasan and O.P. Malik, "The Egyptian Electricity Authority Strategy for distance relay setting: problems and solutions", ELSEVIER Electric Power System Research 56 (2000) 89-94.
- [13] "Power Systems Relay committee (PSRC)" of IEEE C37.113-1999, IEEE guide for "Protective Relay applications to Transmission Lines"
- [14] Fernando Calero, Armando Guzman and Gabriel Benmouyal, "Adaptive Phase and ground Quadrilateral distance Elements" Schweitzer Engineering Laboratories, Inc. 2009.
- [15] K.Brinkis, and D.Drozds, "Load encroachment resistance settings calculation features for digital distance protection" CIGRE Relay Protection and Substation Automation of Modern Power Systems, Cheboksary, September 9-13, 2007.
- [16] <u>http://www.electrocon.com</u>

APPENDIX:



APPENDIX [A]: Single line diagram of Bhutan network

APPENDIX [**B**]: Fault MVA and Current for various system Buses during Three phases fault:

CAPE Result:

3PH SC MVA = ABS (sqrt(3)*basekv/1000 *
Phase A TPH Fault Current)

		TPH	3ph
Fai	ulted Bus	IFA	MVA
1	Bus 11CHP	173302.6	3301.86
2	Bus 220CHP	8771.8	3342.50
3	Bus 66CHP	4375.1	500.14
4	Bus 220Semto	4098.4	1561.71
5	Bus 66Semto	3847.6	439.84
6	Bus 66Chum	2627.2	300.33
7	Bus 66Paro	1432.0	163.70
8	Bus 66Haa	1227.6	140.33
9	Bus_66Jemina	2758.0	315.28
10	Bus_660lakha	3680.4	420.72
11	Bus_66Decho	2482.5	283.79
12	Bus_66Lobe	3353.0	383.30
13	Bus_220Ruri	3230.7	1231.07
14	Bus_66Ruri	5006.7	572.35
15	Bus_66Baso	4534.4	518.36
16	Bus_11Baso	18582.0	354.03
17	Bus_11Ruri	35444.7	675.31
18	Bus_66Gedu	3886.5	444.28
19	Bus_66Pling	5754.9	657.87
20	Bus_220Malbe	8080.0	3078.91
21 22	Bus_66Malbes	10556.3	1206.74 5732.12
22	Bus_Malbes Bus 33Malbe	8273.6 6856.4	391.90
23	Bus 220Singa	7867.5	2997.94
24	Bus 66Singa	3828.2	437.62
26	Bus 66Singa	8917.6	1019.42
20	Bus 66Paskha	9740.5	1113.48
28	Bus 66Gomtu	1852.6	211.78
29	Bus 66Semtsh		153.98
	Bus 13.8Tala		11114.86
	Bus 400Tala	9909.8	6865.69
32	_	8501.6	5890.08
33	Bus 220Bina	9718.9	3703.40
34	Bus 220Birpa	8609.4	3280.63
35	Bus_Indian G	7931.3	5494.99
36	Bus_220Salak	2610.3	994.64
37	Bus_132Salak	5093.3	1164.48
38	Bus_132Rangi	2818.8	644.47
39	Bus_132Gelp	2575.4	588.81
40	Bus_132Ting	2070.9	473.47
41	Bus_132Nglam	2192.4	501.24
42	Bus_132Nang	2701.0	617.53

43	Bus_132Deo	2515.5	575.12
44	Bus_132Motan	2490.5	569.40
45	Bus_132KHP	2606.2	595.86
46	Bus_132Kili	2254.3	515.40
47	Bus_132Kangl	1607.0	367.42
48	Bus_11KHP	39316.2	749.07
49	Bus_66Tserin	1388.3	158.70

Fault MVA for three phase short circuit in DIgSILENT

	Name	Grid	lk" kA	Sk'' MVA	ip kA	lb kA	Sb MVA	lth kA	Rk, Re(Zk) Ohm	Xk, Im(Zk) Ohm
▶ • • •	Basochu11	Part 1	19,07018	363,3358	50,67169	14,94839	284,805	17,78406	0,01868673	0,3325006
-+- v	Basochu6	Part 1	4,407064	503,7951	10,72158	4,059315	464,042	4,473712	1,401932	8,532703
- + - /	Binaguri4	Part 1	11,09483	7686,721	28,69446	10,05833	6968,613	11,38639	1,564797	21,07743
- + /	Biniguri2	Part 1	11,41728	4350,57	30,186	10,82735	4125,773	11,8185	0,6513231	11,2486
- - - /	Birpara2	Part 1	10,80486	4117,205	28,17022	10,30707	3927,521	11,11787	1,118705	11,60721
- -	CHP Gen Bus11	Part 1	191,5375	3649,279	521,3746	164,7748	3139,382	187,7708	0,00143167	0,03312631
- -	Chukha2	Part 1	9,664408	3682,634	25,25397	8,967534	3417,089	9,952636	1,361595	13,0111
- + •	Chukha6	Part 1	3,194363	365,1648	8,334424	3,166223	361,9479	3,287775	1,686888	11,72634
- + v	Chumdo6	Part 1	2,419906	276,6324	4,915104	2,399041	274,2472	2,434447	5,513881	14,13357
- + •	Dechencholing6	Part 1	2,244082	256,533	4,608012	2,219759	253,7525	2,258167	6,050753	15,04052
- + v	Deothang3	Part 1	3,103454	709,5448	6,703138	2,967883	678,5492	3,127604	6,097391	23,8563
- + v	Gedu6	Part 1	1,58557	181,255	3,214522	1,578644	180,4633	1,595028	8,114956	22,25167
- 	Gelephu3	Part 1	2,782622	636,1929	5,740341	2,756502	630,221	2,800418	7,621966	26,576
- - - v	Gen bus 13.8	Part 1	441,7714	10559,35	1204,919	311,5195	7446,033	414,1245	0,00052011	0,0180277
- -	Gomtu6	Part 1	1,457049	166,563	2,588325	1,453771	166,1884	1,462387	11,77371	19,49907
- +	Наа6	Part 1	1,193563	136,4426	2,223718	1,188514	135,8654	1,198752	12,98512	27,71714
- + - v	Jemina6	Part 1	2,547125	291,1755	5,191854	2,521522	288,2486	2,562647	5,188532	13,4387
- + v	Kanglung3	Part 1	1,715961	392,3213	3,658514	1,597951	365,3406	1,728567	11,95053	43,60215
- + v	Kurichu 11	Part 1	40,45857	770,8392	100,0909	31,67958	603,5772	36,77916		0,1559558
- + - v	Kurichu3	Part 1	2,833371	647,7956	6,615969	2,527867	577,948	2,866198	5,244877	26,92869
- + v	Loebsa6	Part 1	3,007893	343,8484	5,992318	2,923294	334,1774	3,024655	4,380975	11,53981
- +	Malbase2	Part 1	8,300515	3162,921	21,1761	7,919558	3017,757	8,48858	2,17611	14,87596
+ /	Malbase4	Part 1	9,335758	6468,003	24,55519	8,277462	5734,794	9,639947	1,509822	25,09667
+ /	Malbesa 3	Part 1	5,474814	312,9276	15,17285	5,442463	311,0785	6,103495	0,1930095	3,323081
+ •	Malbesa6	Part 1	10,34492	1182,584	27,91255	10,16843	1162,407	10,86712	0,5010593	3,502545
+ /	Motanga3	Part 1	3,192596	729,9255	6,940072	3,076197	703,3131	3,218176	5,787854	23,14831
+ /	Nangkhor3	Part 1	3,124672	714,3961	6,826324	2,922976	668,282	3,15029	5,900753	23,92355
+ / + /	Nanglam3	Part 1	2,415362	552,226	4,957332	2,321525	530,7719	2,430493	9,186419	30,58119
+ /	Olakha6	Part 1	3,593213	410,7594	8,494527	3,531314	403,6834	3,638069	2,725978	9,861304
+ /	Paro6 Pasakha6	Part 1 Part 1	1,399458 9,580409	159,9795	2,64194 24,95478	1,392566 9,429105	159,1917 1077,892	1,40584 9,854736	11,10375 0,6560701	23,39987 3,744458
+ /	Phuntsholing6	Part 1	4,066988	1095,188 464,9192		4,039928	461,8258	4,087857	3,478869	7,948891
+ /	Rangia3	Part 1	4,000300	1115,664	7,927702	4,033320	1098,446	5,074073	1,029178	15,51499
+ /	Ririchu6	Part 1	4,970392	568,1921	12,72916	4,604437	528,6653	5,074073	0,9597611	7,576024
+ /	Rurichu 11	Part 1	34,40635	655,5291	92,66511	29,94368	570,5038	33,31728		0,1844375
+ 1	Rurichu2	Part 1	3,152567	1201,289	7,6098	2,96913	1131,39	3,197678	6,016384	39,76477
+ /	Salakati2	Part 1	2,856297	1088,395	7,101401	2,832172	1079,202	2,907391	4,809365	44,23219
-+- v	Salakati3	Part 1	6,034975	1379,781	15,88031	5,983369	1367,982	6,232833	0,8533529	12,63008
-+- v	Samtshe6	Part 1	1,048563	119,8668	1,840293	1,046867	119,673	1,052241	16,16958	27,13203
+ -	Semtokha2	Part 1	4,289545	1634,536	10,35896	4,069084	1550,529	4,351117	4,541445	29,11712
+ -	Semtokha6	Part 1	3,763529	430,2292	9,070367	3,69466	422,3564	3,816805	2,489796	9,452454
+ -	Singhagon2	Part 1	8,099005	3086,136	20,59088	7,736005	2947,814	8,276274	2,264409	15,23463
+ -	Singhgon6	Part 1	4,024573	460,0704	11,18843	3,998281	457,0649	4,565548	1,462995	8,95074
+ -	Singhgon6(1)	Part 1	8,797424	1005,681	22,34259	8,66927	991,0306	8,987965	0,7568677	4,074948
- - - /	Slack Bus 4	Part 1	17,44244	12084,48	43,8187	16,80825	11645,1	17,78367	1,255427	13,18047
- - - /	Tala4	Part 1	11,21979	7773,301	29,88131	9,603966	6653,823	10,71208	1,065472	20,89031
- -	Tintibi3	Part 1	2,204822	504,0901	4,404379	2,172269	496,6474	2,217237	10,79976	33,25121
- + - /	Tsering6	Part 1	1,411355	161,3396	3,271257	1,381877	157,9697	1,427022	4,772557	26,39906
+	kilikhar3	Part 1	2,445616	559,143	5,517317	2,213791	506,1406	2,469074	6,95989	30,9944
		1					-			

APPENDIX [C]: Simulation result for sensitivity test of Distance relay setting:

Fault Type: SL-G of 5 ohms fault resistance: Voltage level: 132kV

Simulation Result: Existing setting

	1 45 Bus_132Ki 42 Bus_132Na				
Curve: 1 Substati Relay: S	ion: Kurichhu 5HPM101_KHP-Nangkho	or SHPM101_1A DIS	T "21" Zone 1		
Curve: 2 Substati Relay: S	ion: Kurichhu SHPM101_KHP-Nangkho	or SHPM101_1A DIS	T "Z2" Zone 2		
Curve: 3 Substati Relay: S	ion: Kurichhu SHPM101_KHP-Nangkho	or SHPM101_1A DIS	T "Z3" Zone 3		
	COPER. SEC API				
Close in A	0.020 3.940	3.4 0.420	3.940 3.4	0.820	3.940 3.4
0.100 B	0.020 4.850	22.4 0.420	4.850 22.4	0.820	4.850 22.4
0.200 C	0.020 6.110	34.7 0.420	6.110 34.7	0.820	6.110 34.7
0.300 D	99999.898 7.560	42.6 0.420	7.560 42.6	0.820	7.560 42.6
0.400 E	99999.898 9.110	47.9 0.420	9.110 47.9	0.820	9.110 47.9
0.500 F		51.8 0.420			
0.600 G			-		-
0.700 H	-		-		-
0.800 I			-		-
	99999.898 17.50		-		17.50 60.0
	99999.898 17.70				
—	99999.898 19.20				

Simulation Result: New setting:

	1 45 Bus_132Ki 42 Bus_132Na			
Curve: 1 Substati Relay: S	on: Kurichhu HPM101_KHP-Nag(Neu	w) SHPM101_1A DIS	T "Z1" Zone 1	
Curve: 2 Substati Relay: S	on: Kurichhu HPM101_KHP-Nag(Nev	w) SHPM101_1A DIS	T "Z2" Zone 2	
Curve: 3 Substati Relay: S	on: Kurichhu HPM101_KHP-Nag(Ne)	w) SHPM101_1A DIS	T "Z3" Zone 3	
				CURVE 3 OPER. SEC APP. IMP
Close in A	0.020 5.340	3.9 0.420	5.340 3.9	0.820 5.340 3.9
0.100 B				0.820 6.570 23.0
0.200 C	0.020 8.280	35.2 0.420	8.280 35.2	0.820 8.280 35.2
0.300 D				0.820 10.20 43.1
0.400 E	0.020 12.30	48.5 0.420	12.30 48.5	0.820 12.30 48.5
0.500 F	99999.898 14.50	52.3 0.420	14.50 52.3	0.820 14.50 52.3
0.600 G	99999.898 16.70	55.1 0.420	16.70 55.1	0.820 16.70 55.1
0.700 H	99999.898 19.00	57.3 0.420	19.00 57.3	0.820 19.00 57.3
0.800 I	99999.898 21.20	59.0 0.420	21.20 59.0	0.820 21.20 59.0
0.900 J			-	0.820 23.50 60.4
	99999.898 22.80			0.820 22.80 61.9
—	99999.898 25.90			0.820 25.90 61.6

Simulation Result for Existing setting without infeed:

Study Line: From: Nganl To: Nang}		41 Bu 42 Bu								
Curve: 1 Subst Relay		-		11111B	CDHF DIST "	'Z1" Zone	1			
Curve: 2 Subst Relay		-		11111B	CDHF DIST "	Z2" Zone	2			
Curve: 3 Subst Relay		-		11111B	CDHF DIST "	'Z4" Zone	2 4			
					CUR					
Location Co										
Close in					0.430					
					0.430					
0.200	С	99999.898	10.40	8.6	0.430	10.40	8.7	0.830	10.40	8.7
0.300	D	99999.898	11.80	13.3	99999.898	11.80	13.3	99999.898	11.80	13.3
0.400	Ε	99999.898	13.30	16.8	99999.898	13.30	16.9	99999.898	13.30	16.9
0.500	F	99999.898	14.80	19.5	99999.898	14.80	19.6	99999.898	14.80	19.6
0.600	G	99999.898		21.6	99999.898	16.40	21.6	99999.898	16.40	21.6
0.700	H		-		99999.898	-		99999.898	-	
0.800		99999.898	-		99999.898	-			-	
	J	99999.898	-		99999.898	-		99999.898	-	
Line_End									16.90	
Remote_Bus					99999.898					

Simulation Result for Existing setting with infeed:

To: Nangkhor 42 Bus_132Nang
Curve: 1 Substation: Nganlam Relay: Main Protection EP311111BCDHF DIST "Z1" Zone 1
Curve: 2 Substation: Nganlam Relay: Main Protection EP311111BCDHF DIST "Z2" Zone 2
Curve: 3 Substation: Nganlam Relay: Main Protection EP311111BCDHF DIST "24" Zone 4
Fault FaultCURVE 1CURVE 2CURVE 3 Location Code OPER. SEC APP. IMP OPER. SEC APP. IMP OPER. SEC APP. IMP
Close in & 99999.898 10.10 -4.8 0.430 10.10 -4.7 0.830 10.10 -4.7
$0.\overline{1}00$ B 99999.898 11.10 2.6 99999.898 11.10 2.7 99999.898 11.10 2.7
0.200 C 99999.898 12.40 8.5 99999.898 12.40 8.6 99999.898 12.40 8.6
0.300 D 99999.898 13.80 13.2 99999.898 13.80 13.3 99999.898 13.80 13.3
0.400 E 99999.898 15.30 16.9 99999.898 15.30 17.0 99999.898 15.30 17.0
0.500 F 99999.898 16.90 19.8 99999.898 16.90 19.9 99999.898 16.90 19.9
0.600 G 99999.898 18.60 22.1 99999.898 18.60 22.1 99999.898 18.60 22.1
0.700 H 99999.898 20.40 23.8 99999.898 20.40 23.8 99999.898 20.40 23.8
0.800 I 99999.898 22.30 25.0 99999.898 22.30 25.1 99999.898 22.30 25.1
0.900 J 99999.898 24.20 25.8 99999.898 24.20 25.9 99999.898 24.20 25.9
Line_End K 99999.898 16.90 58.9 0.430 16.90 59.0 0.830 16.90 59.0
Remote_Bus L 99999.898 26.30 26.3 99999.898 26.30 26.4 99999.898 26.30 26.4

Simulation Result for New setting with infeed:

	anlam	41 Bu : 42 Bu								
		on: Nganlam PAC_Nanglam		o EP31	.1111BCDHF D	IST "Z1	" Zone	1		
		on: Nganlam PAC_Nanglam		O EP31	.1111BCDHF D	IST "Z2	" Zone	2		
		.on: Nganlam :PAC_Nanglam	-Nangkh							
		OPER. SEC	VE 1			VE 2		CUR		
Close in	 A	0.030	10.10		0.430	10.10			 10.10	
					0.430			0.830		
0.200	с	0.030	12.40	8.5	0.430	12.40	8.6	0.830	12.40	8.6
0.300					0.430			0.830		
0.400	E	0.030	15.30	16.9	0.430	15.30	17.0	0.830	15.30	17.0
0.500			16.90		0.430			0.830		19.9
0.600	G	0.030	18.60	22.1	0.430	18.60	22.1	0.830	18.60	22.1
0.700	н	99999.898	20.40	23.8	0.430	20.40	23.8	0.830	20.40	23.8
0.800	I	99999.898	22.30	25.0	0.430	22.30	25.1	0.830	22.30	25.1
0.900	J	99999.898	24.20	25.8	99999.898	24.20	25.9	0.830	24.20	25.9
Line_End	K	99999.898	16.90	58.9	0.430	16.90	59.0	0.830	16.90	59.0
Remote_Bus	L	99999.898	26.30	26.3	99999.898	26.30	26.4	99999.898	26.30	26.4

Fault Type: SL-G of 5 ohms fault resistance: Voltage level: 66kV

Simulation Result for Existing setting:

Fault Study Results Fault type: SLG_A_RS
No network changes specified
Status of Remote End Breaker: CLOSED Remote End Breaker Always OPEN for Line-End Fault
Internal supervision of elements is included
Study Line: From: Rurichu 14 Bus_66Ruri To: Basochu 15 Bus_66Baso
Curve: 1 Substation: Rurichu Relay: Main Protection REL511 V2.3 1A DIST "ZM1 GND" Zone 1
Curve: 2 Substation: Rurichu Relay: Main Protection REL511_V2.3_1A DIST "2M2_GND" Zone 2
Fault FaultCURVE 1CURVE 2 Location Code OPER. SEC APP. IMP OPER. SEC APP. IMP
Close in A 99999.898 2.830 -2.8 0.420 7.320 3.8
0.100 B 99999.898 2.960 -0.9 0.420 7.540 5.8
0.200 C 99999.898 3.100 0.9 0.420 7.760 7.7
0.300 D 99999.898 3.240 2.4 0.420 7.990 9.3
0.400 E 99999.898 3.380 3.8 0.420 8.210 10.8
Close_in A 99999.898 2.830 -2.8 0.420 7.320 3.8 0.100 B 99999.898 2.960 -0.9 0.420 7.540 5.8 0.200 C 99999.898 3.100 0.9 0.420 7.540 7.7 0.300 D 99999.898 3.240 2.4 0.420 7.990 9.3 0.400 E 99999.898 3.380 3.8 0.420 8.210 10.8 0.500 F 99999.898 3.530 5.0 0.420 8.430 12.2 0.600 G 99999.898 3.690 6.0 0.420 8.640 13.4 0.700 H 99999.898 3.850 6.9 0.420 8.860 14.5 0.800 I 99999.898 4.010 7.7 0.420 9.060 15.5 0.900 J 99999.898 4.180 8.3 0.420 9.260 16.3
0.600 G 99999.898 3.690 6.0 0.420 8.640 13.4
0.700 Н 99999.898 3.850 6.9 0.420 8.860 14.5
0.800 I 99999.898 4.010 7.7 0.420 9.060 15.5
0.900 J 99999.898 4.180 8.3 0.420 9.260 16.3
Line_End K 99999.898 2.850 21.9 0.420 8.160 30.7 Remote_Bus L 99999.898 4.350 8.8 0.420 9.450 17.1
Remote_Bus L 99999.898 4.350 8.8 0.420 9.450 17.1

Simulation Result for New setting:

		14 Bu 15 Bu	_					
Curve: 1 Su Re		ion: Rurichu REL511_Baso) RELS	11_V2.3_1A	DIST "Z	M1_GND"	Zone 1
Curve: 2 Su Re		lon: Rurichu EL511_Baso		r) REL5	11_V2.3_1A	DIST "Z	M2_GND"	Zone 2
		CUR	 VF 1		CUP	 VF 2		
Location	Code	OPER. SEC	APP	. IMP	OPER. SEC	APP	. IMP	
		0.020						
0.100	в	0.020	2.960	-0.9	0.420	7.050	15.3	
0.200	С	0.020	3.100	0.9	0.420	7.260	16.9	
0.300	D	0.020	3.240	2.4	0.420	7.480	18.3	
0.400		0.020						
0.500	F	0.020	3.530	5.0	0.420	7.910	20.6	
0.600	G	0.020	3.690	6.0	0.420	8.120	21.6	
0.700	н	0.020	3.850	6.9	0.420	8.330	22.4	
0.800	I	0.020	4.010	7.7	0.420	8.530	23.1	
0.900	J	0.020	4.180	8.3	0.420	8.720	23.7	
Line_End	К	99999.898	2.850	21.9	99999.898	7.440	41.7	
Remote_Bus	L 	0.020	4.350	8.8	0.420	8.910 	24.2	

Simulation Result for Existing setting:

	emtokha	i 5 Bus 10 Bu								
		.on: Semtokh REL511_Sem-C		EL511_	V2.3_1A DIS	T "ZM1_	GND" Z	one 1		
		.on: Semtokh REL511_Sem-C		EL511_	V2.3_1A DIS	т "zm2_	GND" Z	one 2		
Re	elay: F	_	lakha R		V2.3_1A DIS					
Fault Location	Fault Code	OPER. SEC	VE 1 APP	. IMP	CUR OPER. SEC	VE 2 APP	. IMP	CUR OPER. SEC	VE 3 APP	. IMP
					0.420					
					0.420					
0.200	С	99999.898	3.590	5.9	0.420	3.550	5.5	0.820	3.560	5.6
					0.420					
0.400	Е	99999.898	3.730	9.1	0.420	3.690	8.7	0.820	3.700	8.8
0.500	F	99999.898	3.800	10.6	0.420	3.760	10.3	0.820	3.770	10.4
0.600	G	99999.898	3.880	12.1	0.420	3.830	11.7	0.820	3.840	11.8
					0.420					13.2
	I				0.420					14.6
					0.420					15.9
_			-		0.420	-			-	
Remote_Bus	L	99999.898	4.200	17.4	0.420	4.160	17.1	0.820	4.170	17.1

Simulation Result for New setting:

		a 5 Bus 10 Bu						
Curve: 1 Sub Rel		ion: Semtokh REL511_Sem-O		REL51	1_V2.3_1A D	IST "ZM	1_GND"	Zone 1
Curve: 2 Sub Rel		lon: Semtokh REL511_Sem-O		REL51	1_V2.3_1A D	IST "ZM	2_GND"	Zone 2
		OPER. SEC						
		0.020				2.300		
		0.020						
0.200		0.020						
		0.020						
		0.020						
		0.020						
		0.020						
		0.020						
0.800	I	0.020	2.670	12.4	0.420	2.680	11.7	
0.900	J	99999.898	2.720	13.7	0.420	2.740	13.0	
Line End	к	99999.898	2.470	14.9	0.420	2.490	14.3	
Remote_Bus	L	99999.898	2.780	14.9	0.420	2.800	14.3	

Simulation result for three Phase fault of 5 ohms fault resistance for existing setting:

From: S	Semtokha	5	Bus 66	Semto							
Curve: 1 S	Substati	on: Semi	tokha								
				ha REL5:	11_V2.3_1	A DIST	"ZM1_PH	" Zon	e 1		
Curve: 2 S	Substati	on: Semi	tokha								
				ha REL5:	11_V2.3_1	A DIST	"ZM2_PH	" Zon	.e 2		
		_					_				
Curve: 3 S				ha DRIS	11 V2.3 1	л ртст	77 173 DH	" 7on	a 3		
1	Kelay: F	ELSII_S	em-orak	na RELD.	11_02.3_1	A DISI	~2M3_PH	201	e s		
Location	n Code	OPER. S	SEC	APP. II	MP OPER.	SEC	APP.	IMP	OPER. SE	C A	PP. IM
Location Close_in	n Code n A	OPER. 99999.8	SEC 898 6.	APP. II 	MP OPER.	SEC .898 6	APP. 	IMP 3.3	OPER. SE 99999.89	с А 8 6.28	.PP. IM 0 3.
Location Close_in	n Code n A	OPER. 99999.8	SEC 898 6.	APP. II 	MP OPER.	SEC .898 6	APP. 	IMP 3.3	OPER. SE 99999.89	с А 8 6.28	.РР. IM
Location Close_in 0.100	n Code n A D B	OPER. 99999.8	SEC 898 6. 898 6.	APP. I) 280 3 330 4	MP OPER.	SEC .898 6 .898 6	APP. .280 .330	IMP 3.3 4.0	OPER. SE 99999.89 99999.89	C A 8 6.28 8 6.33	.PP. IM 0 3. 0 4.
Location Close_in 0.100 0.200	n Code n A D B D C	OPER. 5 99999.8 99999.8 99999.8	SEC 898 6. 898 6. 898 6.	APP. II 280 3 330 4 380 4	MP OPER. .3 99999 .0 99999	SEC .898 6 .898 6 .898 6	APP. .280 .330 .380	IMP 3.3 4.0 4.6	OPER. SE 99999.89 99999.89 99999.89	C A 8 6.28 8 6.33 8 6.38	.PP. IM
Location Close_in 0.100 0.200	n Code n A D B D C D D	OPER. 5 99999.8 99999.8 99999.8 99999.8	SEC 898 6. 898 6. 898 6. 898 6.	APP. I) 280 3 330 4 380 4 440 5	MP OPER. 	SEC .898 6 .898 6 .898 6 .898 6	APP. .280 .330 .380 .440	IMP 3.3 4.0 4.6 5.2	OPER. SE 99999.89 99999.89 99999.89 99999.89	C A 8 6.28 8 6.33 8 6.38 8 6.44	.PP. IM
Location Close_in 0.100 0.200 0.300	n Code n A D B D C D D D E	OPER. 5 999999.8 99999.8 99999.8 99999.8 99999.8	SEC 898 6. 898 6. 898 6. 898 6. 898 6.	APP. I) 280 3 330 4 380 4 440 5 490 5	MP OPER. 	SEC .898 6 .898 6 .898 6 .898 6 .898 6	APP. .280 .330 .380 .440 .490	IMP 3.3 4.0 4.6 5.2 5.9	OPER. SE 99999.89 99999.89 99999.89 99999.89 99999.89	C A 8 6.28 8 6.33 8 6.38 8 6.44 8 6.49	PP. IM
Location Close_in 0.100 0.200 0.300 0.300	n Code n A D B D C D D D E D F	OPER. 9 99999.8 99999.8 99999.8 99999.8 99999.8 99999.8	SEC 898 6. 898 6. 898 6. 898 6. 898 6.	APP. I) 280 3 330 4 380 4 440 5 490 5 550 6	MP OPER. 	SEC .898 6 .898 6 .898 6 .898 6 .898 6 .898 6 .898 6	APP. .280 .330 .380 .440 .490 .550	IMP 3.3 4.0 4.6 5.2 5.9 6.5	OPER. SE 99999.89 99999.89 99999.89 99999.89 99999.89 99999.89	C A 8 6.28 8 6.33 8 6.38 8 6.44 8 6.49 8 6.55	PP. IM 0 3. 0 4. 0 4. 0 5. 0 5. 0 5. 0 6.
Location Close_in 0.100 0.200 0.300 0.400 0.500	n Code n A D B D C D D D E D F D G	OPER. 5 99999.1 99999.1 99999.1 99999.1 99999.1 99999.1 99999.1	SEC 898 6. 898 6. 898 6. 898 6. 898 6. 898 6. 898 6.	APP. II 280 3 330 4 380 4 440 5 490 5 550 6 600 7	MP OPER. .3 99999 .0 99999 .6 99999 .2 99999 .9 99999 .5 99999	SEC .898 6 .898 6 .898 6 .898 6 .898 6 .898 6 .898 6 .898 6	APP. .280 .330 .380 .440 .490 .550 .600	IMP 3.3 4.0 4.6 5.2 5.9 6.5 7.1	OPER. SE 99999.89 99999.89 99999.89 99999.89 99999.89 99999.89 99999.89	C A 8 6.28 8 6.33 8 6.38 8 6.44 8 6.49 8 6.55 8 6.60	PP. IM 0 3. 0 4. 0 4. 0 5. 0 5. 0 5. 0 5. 0 7.
Location Close_in 0.100 0.200 0.300 0.400 0.500 0.600 0.600	n Code n A D B D C D D D E D F D G D H	OPER. 5 99999.1 99999.1 99999.1 99999.1 99999.1 99999.1 99999.1	SEC 898 6. 898 6. 898 6. 898 6. 898 6. 898 6. 898 6. 898 6.	APP. I) 280 3 330 4 380 4 440 5 490 5 550 6 600 7 660 7	MP OPER. .3 99999 .0 99999 .6 99999 .2 99999 .9 99999 .5 99999 .1 99999	SEC .898 6 .898 6 .898 6 .898 6 .898 6 .898 6 .898 6 .898 6	APP. .280 .330 .380 .440 .550 .600 .660	IMP 3.3 4.0 4.6 5.2 5.9 6.5 7.1 7.6	OPER. SE 99999.89 99999.89 99999.89 99999.89 99999.89 99999.89 99999.89 99999.89	C A 8 6.28 8 6.33 8 6.38 8 6.44 8 6.49 8 6.55 8 6.60 8 6.66	PP. IM 0 3. 0 4. 0 4. 0 5. 0 5. 0 5. 0 5. 0 7. 0 7.
Location Close_in 0.100 0.200 0.300 0.400 0.500 0.600 0.700 0.800	n Code n A D B D C D D D E D F D G D H D H	OPER. 5 99999.1 99999.1 99999.1 99999.1 99999.1 99999.1 99999.1 99999.1	SEC 898 6. 898 6. 898 6. 898 6. 898 6. 898 6. 898 6. 898 6. 898 6.	APP. I) 280 3 330 4 380 4 440 5 490 5 550 6 600 7 660 7 720 8	MP OPER. .3 99999 .0 99999 .6 99999 .2 99999 .9 99999 .5 99999 .1 99999 .6 99999	SEC .898 6 .898 6 .898 6 .898 6 .898 6 .898 6 .898 6 .898 6 .898 6	APP. .280 .330 .380 .440 .550 .600 .600 .720	IMP 3.3 4.0 4.6 5.2 5.9 6.5 7.1 7.6 8.2	OPER. SE 99999.89 99999.89 99999.89 99999.89 99999.89 99999.89 99999.89 99999.89	C A 8 6.28 8 6.33 8 6.38 8 6.44 8 6.49 8 6.55 8 6.60 8 6.60 8 6.72	PP. IM 0 3. 0 4. 0 4. 0 5. 0 5. 0 5. 0 5. 0 7. 0 7. 0 8.
Location Close_in 0.100 0.200 0.300 0.400 0.500 0.500 0.700 0.800 0.900	n Code n A D B D C D E D F D F D F D G D H D I J J	OPER. 5 99999.1 99999.1 99999.1 99999.1 99999.1 99999.1 99999.1 99999.1	SEC 898 6. 898 6. 898 6. 898 6. 898 6. 898 6. 898 6. 898 6. 898 6. 898 6.	APP. I) 280 3 330 4 380 4 440 5 550 5 550 6 600 7 720 8 780 8	MP OPER. .3 99999 .0 99999 .6 99999 .2 99999 .9 99999 .5 99999 .1 99999 .6 99999 .2 99999	SEC .898 6 .898 6 .898 6 .898 6 .898 6 .898 6 .898 6 .898 6 .898 6 .898 6	APP. .280 .330 .380 .440 .550 .600 .660 .720 .780	IMP 3.3 4.0 4.6 5.2 5.9 6.5 7.1 7.6 8.2 8.8	OPER. SE 99999.89 99999.89 99999.89 99999.89 99999.89 99999.89 99999.89 99999.89	C A 8 6.28 8 6.33 8 6.38 8 6.44 8 6.49 8 6.55 8 6.66 8 6.66 8 6.72 8 6.78	PP. IM 0 3. 0 4. 0 4. 0 5. 0 5. 0 5. 0 7. 0 7. 0 8. 0 8.

Simulation result for three Phase fault of 5 ohms fault resistance for new setting:

Curve: 1 Substation: Semtokha Relay: REL511_Sem-Ola(New) REL511_V2.3_1A DIST "ZM1_PH" Zone 1 Curve: 2 Substation: Semtokha Relay: REL511_Sem-Ola(New) REL511_V2.3_1A DIST "ZM2_PH" Zone 2 	Study Line: From: Semtokha 5 Bus_66Semto To: Olakha 10 Bus_66Olakha													
Relay: REL511_Sem-Ola(New) REL511_V2.3_1A DIST "ZM2_PH" Zone 2 Fault Fault CURVE 1 CURVE 2 Location Code OPER. SEC APP. IMP OPER. SEC APP. IMP Close_in A 0.020 3.770 3.3 0.420 3.770 3.3 0.100 B 0.020 3.810 4.4 0.420 3.810 4.4 0.200 C 0.020 3.810 4.4 0.420 3.860 5.4 0.300 D 0.020 3.910 6.5 0.420 3.960 7.5 0.400 E 0.020 3.960 7.5 0.420 3.960 7.5 0.500 F 0.020 4.010 8.4 0.420 4.010 8.4 0.600 G 0.020 4.070 9.4 0.420 4.070 9.4 0.700 H 99999.898 4.120 10.3 0.420 4.120 10.3 0.800 I 99999.898 4.180 11.2 0.420 4.230 12.1														
LocationCodeOPER. SECAPP. IMPOPER. SECAPP. IMPClose_inA0.0203.7703.30.4203.7703.30.100B0.0203.8104.40.4203.8104.40.200C0.0203.8605.40.4203.8605.40.300D0.0203.9106.50.4203.9106.50.400E0.0203.9607.50.4203.9607.50.500F0.0204.0108.40.4204.0108.40.600G0.0204.0709.40.4204.0709.40.700H99999.8984.12010.30.4204.12010.30.800I99999.8984.23012.10.4204.23012.1Line_EndK99999.8983.44011.90.4203.44011.9														
0.100 B 0.020 3.810 4.4 0.420 3.810 4.4 0.200 C 0.020 3.860 5.4 0.420 3.860 5.4 0.300 D 0.020 3.910 6.5 0.420 3.910 6.5 0.400 E 0.020 3.960 7.5 0.420 3.960 7.5 0.500 F 0.020 4.010 8.4 0.420 4.010 8.4 0.600 G 0.020 4.070 9.4 0.420 4.070 9.4 0.700 H 99999.898 4.120 10.3 0.420 4.120 10.3 0.800 I 99999.898 4.180 11.2 0.420 4.180 11.2 0.900 J 99999.898 3.440 11.9 0.420 3.440 11.9														
0.200 C 0.020 3.860 5.4 0.420 3.860 5.4 0.300 D 0.020 3.910 6.5 0.420 3.910 6.5 0.400 E 0.020 3.960 7.5 0.420 3.960 7.5 0.500 F 0.020 4.010 8.4 0.420 4.010 8.4 0.600 G 0.020 4.070 9.4 0.420 4.070 9.4 0.700 H 99999.898 4.120 10.3 0.420 4.120 10.3 0.800 I 99999.898 4.180 11.2 0.420 4.180 11.2 0.900 J 99999.898 4.230 12.1 0.420 4.230 12.1 Line_End K 99999.898 3.440 11.9 0.420 3.440 11.9	Close in	 A	0.020	3.770	3.3	 0.420	 3.770	3.3						
0.200 C 0.020 3.860 5.4 0.420 3.860 5.4 0.300 D 0.020 3.910 6.5 0.420 3.910 6.5 0.400 E 0.020 3.960 7.5 0.420 3.960 7.5 0.500 F 0.020 4.010 8.4 0.420 4.010 8.4 0.600 G 0.020 4.070 9.4 0.420 4.070 9.4 0.700 H 99999.898 4.120 10.3 0.420 4.120 10.3 0.800 I 99999.898 4.180 11.2 0.420 4.180 11.2 0.900 J 99999.898 4.230 12.1 0.420 4.230 12.1 Line_End K 99999.898 3.440 11.9 0.420 3.440 11.9	0.100	в	0.020	3.810	4.4	0.420	3.810	4.4						
0.300 D 0.020 3.910 6.5 0.420 3.910 6.5 0.400 E 0.020 3.960 7.5 0.420 3.960 7.5 0.500 F 0.020 4.010 8.4 0.420 4.010 8.4 0.600 G 0.020 4.070 9.4 0.420 4.070 9.4 0.700 H 99999.898 4.120 10.3 0.420 4.120 10.3 0.800 I 99999.898 4.180 11.2 0.420 4.180 11.2 0.900 J 99999.898 4.230 12.1 0.420 4.230 12.1 Line_End K 99999.898 3.440 11.9 0.420 3.440 11.9														
0.500 F 0.020 4.010 8.4 0.420 4.010 8.4 0.600 G 0.020 4.070 9.4 0.420 4.070 9.4 0.700 H 99999.898 4.120 10.3 0.420 4.120 10.3 0.800 I 99999.898 4.180 11.2 0.420 4.180 11.2 0.900 J 99999.898 4.230 12.1 0.420 4.230 12.1 Line_End K 99999.898 3.440 11.9 0.420 3.440 11.9														
0.600 G 0.020 4.070 9.4 0.420 4.070 9.4 0.700 H 99999.898 4.120 10.3 0.420 4.120 10.3 0.800 I 99999.898 4.180 11.2 0.420 4.180 11.2 0.900 J 99999.898 4.230 12.1 0.420 4.230 12.1 Line_End K 99999.898 3.440 11.9 0.420 3.440 11.9	0.400	E	0.020	3.960	7.5	0.420	3.960	7.5						
0.700 H 99999.898 4.120 10.3 0.420 4.120 10.3 0.800 I 99999.898 4.180 11.2 0.420 4.180 11.2 0.900 J 99999.898 4.230 12.1 0.420 4.230 12.1 Line_End K 99999.898 3.440 11.9 0.420 3.440 11.9	0.500	F	0.020	4.010	8.4	0.420	4.010	8.4						
0.800 I 99999.898 4.180 11.2 0.420 4.180 11.2 0.900 J 99999.898 4.230 12.1 0.420 4.230 12.1 Line_End K 99999.898 3.440 11.9 0.420 3.440 11.9	0.600	G	0.020	4.070	9.4	0.420	4.070	9.4						
0.900 J 99999.898 4.230 12.1 0.420 4.230 12.1 Line_End K 99999.898 3.440 11.9 0.420 3.440 11.9	0.700	н	99999.898	4.120	10.3	0.420	4.120	10.3						
Line_End K 99999.898 3.440 11.9 0.420 3.440 11.9	0.800	I	99999.898	4.180	11.2	0.420	4.180	11.2						
-	0.900	J	99999.898	4.230	12.1	0.420	4.230	12.1						
Remote_Bus L 99999.898 4.290 12.9 0.420 4.290 12.9	Line_End	К	99999.898	3.440	11.9	0.420	3.440	11.9						
	Remote_Bus	L 	99999.898	4.290	12.9	0.420	4.290	12.9						

APPENDIX [D]: Simulation result for Coordination & Sequential Operation of distance relay:

Creating a SL fault between the 66kV line Jemina- Chumdo at 50% of the protective line and assuming the fault resistance of 5 ohms. The simulation result is as below:

*** Starting event # 1
Fault 1 of 1:
Midline node on "6 Bus_66Chum" to "9 Bus_66Jemina" Ckt 1
"999001 Bus 66Chum" (NEWBUS1) distant 0.500 from "6 Bus 66Chum"

SLG A R5 at temporary bus "999001 Bus 66Chum" (NEWBUS1)

Number of fault buses: 1 Bus & phase pair Fault current Amps @ deg 999001A - 00 1338.345 @ -77.30

LZOP Summary Report

LZOP Operating Times (s) predicted at 0.080 seconds from start:

S/S ID LZOP Name Type P/B Trip LZOP Breaker Total Olakha 45 66kVOlakha-Jemina LINE Bkp Z1 0.020 0.060 0.080 Op in event1 3ph Jemina 48 66kVChumdo feeder LINE Pri Z1 0.030 0.060 0.090 Bkr opening:3ph Chumdo 49 66kVJemina feeder LINE Pri Z1G 0.040 0.060 0.100 Bkr opening:3ph Chukha 51 66kV Chumdo feeder LINE Bkp Z3 0.820 0.060 0.880 Predicted

LZOP	Breaker type & location	Bkr	openi	ing cyc	Tripp	ed by
45 LINE	E Branch"10Bus_660lakha"to"9Bus_66Jem	ina"C	kt1 3.	.0 Opened	3pole	210LD
48 LINE	E Branch"9Bus_66Jemina"to"6Bus_66Chum	"Ckt1	3.0	Opening	3pole	210LD
49 Brar	nch"6Bus_66Chum"to"9Bus_66Jemina"Ckt1	3.0	Open	ing 3pole	21_TRI	P_OLD

Network changes now in effect:

Midline node on "6 Bus_66Chum" to "9 Bus_66Jemina" Ckt 1 "999001 Bus_66Chum" (NEWBUS1) distant 0.500 from "6 Bus_66Chum" Open breaker on "10 Bus_66Olakha" to "9 Bus_66Jemina" Ckt 1 at "10 Bus_66Olakha"; New bus "999002 Bus_66Olakha" (NEWBUS2) SLG_A_R5 at temporary bus "999001 Bus_66Chum" (NEWBUS1)

Fault is not cleared after 4.0 cycles 0.080 seconds

*** Starting event # 2 Fault 1 of 1: Number of fault buses: 1 Bus & phase pair Fault current Amps @ deg 999001A - 00 582.088 @ -84.25

With breakers open in event # 1 continue to event # 2

Network changes now in effect:

Midline node on "6 Bus_66Chum" to "9 Bus_66Jemina" Ckt 1 "999001 Bus_66Chum" (NEWBUS1) distant 0.500 from "6 Bus_66Chum" Open breaker on "10 Bus_66Olakha" to "9 Bus_66Jemina" Ckt 1 at "10 Bus_66Olakha"; New bus "999002 Bus_66Olakha" (NEWBUS2) SLG_A_R5 at temporary bus "999001 Bus_66Chum" (NEWBUS1)

After event 2 4.5 c 0.090 s

LZOP Summary Report

LZOP Operating Times (s) predicted at 0.090 seconds from start:

S/S ID LZOP Name Type P/B Trip LZOP Breaker Total Olakha 45 66kV Olakha-Jemina LINE Bkp Z1 0.020 0.060 0.080 Op in event1 3ph Jemina 48 66kVChumdo feeder LINE Pri Z1 0.030 0.060 0.090 Op in event2 3ph Chumdo 49 66kV Jemina fdr. LINE Pri Z1G 0.040 0.060 0.100 Bkr opening: 3ph Chukha 51 66kV Chumdo fdr. LINE Bkp Z3 0.820 0.060 0.880 Predicted Logical breakers for all LZOPS asserted in previous steps:

LZOPBreaker type & locationBkr opening cycTripped by45 Branch"10Bus_660lakha"to"9Bus_66Jemina"Ckt1 3.0Opened 3pole 210LD48 Branch"9Bus_66Jemina"to"6Bus_66Chum"Ckt13.0Opened 3pole 210LD49 Branch"6Bus 66Chum"to"9Bus 66Jemina"Ckt13.0Opening 3pole 21 TRIP 0LD

Network changes now in effect:

Midline node on "6 Bus_66Chum" to "9 Bus_66Jemina" Ckt 1 "999001 Bus_66Chum" (NEWBUS1) distant 0.500 from "6 Bus_66Chum" Open breaker on "10 Bus_66Olakha" to "9 Bus_66Jemina" Ckt 1 at "10 Bus_66Olakha"; New bus "999002 Bus_66Olakha" (NEWBUS2) Open breaker on "9 Bus_66Jemina" to "6 Bus_66Chum" Ckt 1 at "9 Bus_66Jemina"; New bus "999003 Bus_66Jemina" (NEWBUS3) SLG A R5 at temporary bus "999001 Bus 66Chum" (NEWBUS1)

Fault is not cleared after 4.5 cycles 0.090 seconds Network after Events 2:



*** Starting event # 3
Fault 1 of 1:
Bus & phase pair Fault current Amps @ deg
999001A - 00 586.254 @ -83.75

With breakers open in event # 2 continue to event # 3

Network changes now in effect:

Midline node on "6 Bus_66Chum" to "9 Bus_66Jemina" Ckt 1 "999001 Bus_66Chum" (NEWBUS1) distant 0.500 from "6 Bus_66Chum" Open breaker on "10 Bus_66Olakha" to "9 Bus_66Jemina" Ckt 1 at "10 Bus_66Olakha"; New bus "999002 Bus_66Olakha" (NEWBUS2) Open breaker on "9 Bus_66Jemina" to "6 Bus_66Chum" Ckt 1 at "9 Bus_66Jemina"; New bus "999003 Bus_66Jemina" (NEWBUS3) SLG_A_R5 at temporary bus "999001 Bus_66Chum" (NEWBUS1)

After event 3 5.0 c 0.100 s

LZOP Summary Report

LZOP Operating Times (s) predicted at 0.100 seconds from start:

S/S ID LZOP Name Type P/B Trip LZOP Breaker Total Olakha 45 66kVOlakha-Jemina LINE Bkp Z1 0.020 0.060 0.080 Op in event1 3ph Jemina 48 66kV Chumdo fdr. LINE Pri Z1 0.030 0.060 0.090 Op in event2 3ph Chumdo 49 66kV Jemina fdr. LINE Pri Z1G 0.040 0.060 0.100 Op in event3 3ph Chukha 51 66kV Chumdo fdr. LINE Bkp Z3 0.820 0.060 0.880 Predicted

Logical breakers for all LZOPS asserted in previous steps: LZOP Breaker type & location Bkr opening cyc Tripped by 45 Branch"10Bus_660lakha"to"9Bus_66Jemina"Ckt1 3.0 Opened 3pole 210LD 48 Branch"9Bus_66Jemina"to"6Bus_66Chum"Ckt1 3.0 Opened 3pole 210LD 49 Branch"6Bus 66Chum"to"9Bus 66Jemina"Ckt1 3.0 Opened 3pole 21 TRIP OLD

Network changes now in effect:

Midline node on "6 Bus_66Chum" to "9 Bus_66Jemina" Ckt 1 "999001 Bus_66Chum" (NEWBUS1) distant 0.500 from "6 Bus_66Chum" Open breaker on "10 Bus_66Olakha" to "9 Bus_66Jemina" Ckt 1 at "10 Bus_66Olakha"; New bus "999002 Bus_66Olakha" (NEWBUS2) Open breaker on "9 Bus_66Jemina" to "6 Bus_66Chum" Ckt 1 at "9 Bus_66Jemina"; New bus "999003 Bus_66Jemina" (NEWBUS3) Open breaker on "6 Bus_66Chum" to "9 Bus_66Jemina" Ckt 1 at "6 Bus_66Chum"; New bus "999004 Bus_66Chum" (NEWBUS4) SLG A R5 at temporary bus "999001 Bus 66Chum" (NEWBUS1)

Fault is cleared after 5.0 cycles 0.100 seconds

58

Check By Simulation: open breakers in successive steps Simulation Area:

LZOP Summary Report

LZOP Operating Times (s) predicted at 0.090 seconds from start:

S/S ID LZOP Name Type P/B Trip LZOP Breaker Total Jemina 48 66kV Chumdo LINE Pri Z1N 0.030 0.060 0.090 Op in event1 3ph Chumdo 49 66kV Jemina LINE Pri Z1G 0.040 0.060 0.100 Bkr opening:3ph Chukha 51 66kV Chumdo LINE Bkp Z3 0.820 0.060 0.880 Predicted Olakha 45 66kV Olakha LINE Bkp Z3N 0.820 0.060 0.880 Predicted

Logical breakers for all LZOPS asserted in this step:

 LZOP
 Breaker type & location
 Bkr opening cyc
 Tripped by

 48 Branch"9Bus_66Jemina"to"6Bus_66Chum"Ckt1
 3.0
 Opened 3-pole
 21NEW

49 Branch"6Bus 66Chum"to"9Bus 66Jemina"Ckt1 3.0 Opening 3pole 21 TRIP OLD

Network changes now in effect:

Midline node on "6 Bus_66Chum" to "9 Bus_66Jemina" Ckt 1 "999001 Bus_66Chum" (NEWBUS1) distant 0.500 from "6 Bus_66Chum" Open breaker on "9 Bus_66Jemina" to "6 Bus_66Chum" Ckt 1 at "9 Bus_66Jemina"; New bus "999002 Bus_66Jemina" (NEWBUS2) SLG A R5 at temporary bus "999001 Bus 66Chum" (NEWBUS1)

Fault is not cleared after 4.5 cycles 0.090 seconds



*** Starting event # 2 Fault 1 of 1: Midline node on "6 Bus_66Chum" to "9 Bus_66Jemina" Ckt 1 "999001 Bus_66Chum" (NEWBUS1) distant 0.500 from "6 Bus_66Chum" Open breaker on "9 Bus_66Jemina" to "6 Bus_66Chum" Ckt 1 at "9 Bus_66Jemina"; New bus "999002 Bus_66Jemina" (NEWBUS2) SLG A R5 at temporary bus "999001 Bus 66Chum" (NEWBUS1)

Number of fault buses: 1 Bus & phase pair Fault current Amps @ deg 999001A - 00 586.175 @ -83.77

With breakers open in event # 1 continue to event # 2 After event 2 5.0 c 0.100 s

LZOP Summary Report

LZOP Operating Times (s) predicted at 0.100 seconds from start:

S/S IDLZOPNameTypeP/BTripLZOPBreakerTotalJemina4866kVChumdoLINEPriZ1N0.0300.0600.090Op in event13-phChumdo4966kVJeminaLINEPriZ1G0.0400.0600.100Op in event23-phChukha5166kVChumdoLINEBkpZ30.8200.0600.880Predicted

LZOPBreaker type & locationBkr opening cycTripped by48 Branch"9Bus 66Jemina"to"6Bus 66Chum"Ckt13.0Opened 3-pole21NEW

49 Branch"6Bus 66Chum"to"9Bus 66Jemina"Ckt1 3.0 Opened 3pole 21 TRIP OLD

Network changes now in effect:

Midline node on "6 Bus_66Chum" to "9 Bus_66Jemina" Ckt 1 "999001 Bus_66Chum" (NEWBUS1) distant 0.500 from "6 Bus_66Chum" Open breaker on "9 Bus_66Jemina" to "6 Bus_66Chum" Ckt 1 at "9 Bus_66Jemina"; New bus "999002 Bus_66Jemina" (NEWBUS2) Open breaker on "6 Bus_66Chum" to "9 Bus_66Jemina" Ckt 1 at "6 Bus_66Chum"; New bus "999003 Bus_66Chum" (NEWBUS3) SLG_A_R5 at temporary bus "999001 Bus_66Chum" (NEWBUS1)

Fault is cleared after 5.0 cycles 0.100 seconds Final Network after fault is cleared and showing the relay operation:



APPENDIX [E]: Simulation showing the important of 67N:

Case1: Simulation Result for without 67N:

```
Bus & phase pair Fault current Amps @ deg
999001A - 00 1325.622 @ -43.94
```

Check_By Simulation: open breakers in successive steps Simulation Area:



From above simulation we see that all the distance relay were not detecting the high impedance fault, therefore it is not operated and the fault is not able to clear.

Case2: Simulation Result for with 67N:

```
SS FAULT COMMAND: APPLY SILENT FAULT SLG A R20 NEWBUS1
******
*** Starting event # 1
Fault 1 of 1:
Midline node on "12 Bus 66Lobe" to "14 Bus 66Ruri" Ckt 1
"999001 Bus 66Lobe" (NEWBUS1) distant 0.500 from "12 Bus 66Lobe"
SLG A R20 at temporary bus "999001 Bus 66Lobe" (NEWBUS1)
Number of fault buses: 1
Bus & phase pair Fault current Amps @ deg
999001A -
            00
                  1325.622 @ -43.94
Check By Simulation: open breakers in successive steps
After event 1 21.7 c 0.434 s
Fastest Primary:
Primary LZOP:
             28 66kV Rurichu-Lobeysa feeder at Rurichu; 3-pole
LZOP 0.374; Breaker 0.060; LZOP+Bkr 0.434 sec
Trip path TEF REL511 Lobeysa Fdr. 28 TOC IN> TEF 0.374 sec from
start
Fastest Backup:
```

Backup LZOP: 43 66kV Semtokha-Lobeysa feeder at Semtokha; 3-pole LZOP 1.04; Breaker 0.060; LZOP+Bkr 1.10 sec
Trip path TEF REL511_Sem-Lobeysa 46 TOC IN>_TEF 1.042 sec from start CTI: Min desired 0.300 Max desired 9999.00 Predicted 0.668 seconds CTI Defn: TB LZOP - TP LZOP

LZOP Summary Report

S/S IDLZOPNameType P/B TripLZOPBreakerTotalRurichu 2866kVRurichu LINEPriTEF0.3740.0600.434Op in event1 3phLobeysa 4166kVLobeysa LINEPriTEF0.6040.0600.664PredictedSemtokha 4366kVSemtokha LINEBkpTEF1.0420.0601.102PredictedLZOPBreaker type & locationBkr opening cycTripped by28Branch"14Bus66Ruri-12Bus66Lobe"Ckt13.00pened3poleGENTRIPNEWABB

Network changes now in effect:

Midline node on "12 Bus_66Lobe" to "14 Bus_66Ruri" Ckt 1 "999001 Bus_66Lobe" (NEWBUS1) distant 0.500 from "12 Bus_66Lobe" Open breaker on "14 Bus_66Ruri" to "12 Bus_66Lobe" Ckt 1 at "14 Bus_66Ruri"; New bus "999002 Bus_66Ruri" (NEWBUS2) SLG_A_R20 at temporary bus "999001 Bus_66Lobe" (NEWBUS1) Fault is not cleared after 21.7 cycles 0.434 seconds



*** Starting event # 2

Fault 1 of 1: Number of fault buses: 1 Bus & phase pair Fault current Amps @ deg 999001A - 00 572.147 @ -70.82

Midline node on "12 Bus_66Lobe" to "14 Bus_66Ruri" Ckt 1 "999001 Bus_66Lobe" (NEWBUS1) distant 0.500 from "12 Bus_66Lobe" Open breaker on "14 Bus_66Ruri" to "12 Bus_66Lobe" Ckt 1 at "14 Bus_66Ruri"; New bus "999002 Bus_66Ruri" (NEWBUS2) SLG A R20 at temporary bus "999001 Bus 66Lobe" (NEWBUS1)

After event 2 25.7 c 0.514 s

Fastest Primary:

Primary LZOP: 41 66kV Lobeysa-Basochu feeder at Lobeysa; 3-pole LZOP 0.454; Breaker 0.060; LZOP+Bkr 0.514 sec Trip path Z1G_N REL511_Lob-Ruri(New) 73 DIST ZM1_GND 1 0.454 sec from start

Fastest Backup:

Backup LZOP: 43 66kV Semtokha-Lobeysa feeder at Semtokha; 3-pole LZOP 0.854; Breaker 0.060; LZOP+Bkr 0.914 sec Trip path Z2G REL511_Sem-Lobeysa 46 TIMER t2PE 1 0.854 sec from start

CTI: Min desired 0.300 Max desired 9999.00 Predicted 0.400 seconds CTI Defn: TB LZOP - TP LZOP

LZOP Summary Report

S/S ID LZOP Name Type P/B Trip LZOP Breaker Total Rurichu 28 66kV Rurichu-Lobeysa Pri TEF 0.374 0.060 0.434 Op in event1 3ph Lobeysa 41 66kV Lobeysa-Basochu Pri Z1G_N 0.454 0.060 0.514 Op event2 3ph Semtokha 43 66kV Semtokha-Lobeysa Bkp Z2G 0.854 0.060 0.914 Predicted

LZOP Breaker type & location Bkr opening cyc Tripped by

28LINE Branch"14Bus 66Ruri-12Bus 66Lobe"Ckt13.0 Opened3-pole GENTRIPNEW ABB

41LINE Branch"12Bus 66Lobe-14Bus 66Ruri"Ckt13.0 Opened3-pole GENTRIPNEW ABB

Network changes now in effect:

Midline node on "12 Bus_66Lobe" to "14 Bus_66Ruri" Ckt 1 "999001 Bus_66Lobe" (NEWBUS1) distant 0.500 from "12 Bus_66Lobe" Open breaker on "14 Bus_66Ruri" to "12 Bus_66Lobe" Ckt 1 at "14 Bus_66Ruri"; New bus "999002 Bus_66Ruri" (NEWBUS2) Open breaker on "12 Bus_66Lobe" to "14 Bus_66Ruri" Ckt 1 at "12 Bus_66Lobe"; New bus "999003 Bus_66Lobe" (NEWBUS3) SLG A R20 at temporary bus "999001 Bus 66Lobe" (NEWBUS1)

Fault is cleared after 25.7 cycles 0.514 seconds

APPENDIX [F]: Simulation Result 67N (CDD21) Coordination:

132kV line between Kurichu to Gelephu, Relay coordination checking for the relay looking towards the Gelephu:

Case 1: Existing setting simulation result:



Fault: A

SLG_A_R30 at temporary bus 999001 Bus_132Gelp (NEWBUS1) Midline node on "39 Bus_132Gelp" to "37 Bus_132Salak" Ckt 1 "999001 Bus 132Gelp" (NEWBUS1) distant 0.500 from "39 Bus_132Gelp"

Curve	Current		Operating	Source/Total line (+ seq SIR)		
	Primary A	A/Pickup	Seconds			
-	100 64	1 0 4		2 24 2 21 5		
T	183.64	1.84	4.526	3.24 @ 21.5		
2	231.89	3.86	1.404	2.67 @ 14.0		
3	231.89	3.86	1.022	1.51 @ 10.7		
4	231.89	3.86	0.766	4.63 @ 5.9		
5	231.89	3.86	0.510	5.01 @ 4.9		

Fault: B

SLG_A_R30 at temporary bus 999001 Bus_132Gelp (NEWBUS1) Midline node on "39 Bus_132Gelp" to "40 Bus_132Ting" Ckt 1 "999001 Bus 132Gelp" (NEWBUS1) distant 0.499 from "39 Bus 132Gelp"

Curve	Cur Primary A	rent A/Pickup	Operating Seconds	Source/Total line (+ seq SIR)
1	273.84	2.74	2.720	3.24 @ 21.5
2	381.21	6.35	1.022	2.45 @ 14.3
3	381.21	6.35	0.744	1.42 @ 10.7
4	381.21	6.35	0.558	4.46 @ 5.7
5	781.25	13.02	Infinite	1.70 @-171.7

Fault: C

SLG_A_R30 at temporary bus 999001 Bus_132Ting (NEWBUS1) Midline node on "40 Bus_132Ting" to "41 Bus_132Nglam" Ckt 1 "999001 Bus_132Ting" (NEWBUS1) distant 0.500 from "40 Bus_132Ting"

Curve	Curr Primary A		Operating Seconds	Source/Total line (+ seq SIR)
	400 50	4 0 0	1 074	2. 04. 0. 01. 5
T	428.79	4.29	1.874	3.24 @ 21.5
2	618.83	10.31	0.806	2.35 @ 14.1
3	618.83	10.31	0.586	1.38 @ 10.5
4	454.34	7.57	Infinite	2.95 @-174.4
5	454.34	7.57	Infinite	1.74 @-171.5

Fault: D

SLG_A_R30 at temporary bus 999001 Bus_132Nglam (NEWBUS1) Midline node on "41 Bus_132Nglam" to "42 Bus_132Nang" Ckt 1 "999001 Bus_132Nglam" (NEWBUS1) distant 0.500 from "41 Bus_132Nglam"

Current		Operating	Source/Total line (+ seq SIR)		
Primary A	A/Pickup	Seconds			
739.98	7.40	1.356	3.24 @ 21.5		
1082.77	18.05	0.646	2.31 @ 14.0		
262.21	4.37	Infinite	2.63 @-176.1		
262.21	4.37	Infinite	3.03 @-174.3		
262.21	4.37	Infinite	1.82 @-171.4		
	739.98 1082.77 262.21 262.21	Primary A A/Pickup 739.98 7.40 1082.77 18.05 262.21 4.37 262.21 4.37	Primary A A/Pickup Seconds 739.98 7.40 1.356 1082.77 18.05 0.646 262.21 4.37 Infinite 262.21 4.37 Infinite		

Fault: E

SLG_A_R30 at temporary bus 999001 Bus_132Nang (NEWBUS1)
Midline node on "42 Bus_132Nang" to "45 Bus_132KHP" Ckt 1
"999001 Bus_132Nang" (NEWBUS1) distant 0.501 from "42 Bus_132Nang"

Curve	Curr Primary A		Operating Seconds	Source/Total line (+ seq SIR)
1	1197.90	11.98	1.088	3.24 @ 21.5
2	136.28	2.27	Infinite	7.44 @-177.1
3	136.28	2.27	Infinite	2.66 @-176.1
4	136.28	2.27	Infinite	3.08 @-174.3
5	136.28	2.27	Infinite	1.86 @-171.6



From the above simulation the operation sequence are fine but operating time of the relay is longer than Zone 2 timing for all the cases. Therefore time coordination is required.

Case 2: New setting simulation result:

Case 2. IN	ew setting sim	ulation result	•					
<pre>Fault: A SLG_A_R30 at temporary bus 999001 Bus_132Gelp (NEWBUS1) Midline node on "39 Bus_132Gelp" to "37 Bus_132Salak" Ckt 1 "999001 Bus_132Gelp" (NEWBUS1) distant 0.500 from "39 Bus_132Gelp"</pre>								
Curve	Current		Operating	Source/Total line (+ seq				
SIR)	Primary A	A/Pickup	Seconds					
1	183.64	1.84	2.562	3.24 @ 21.5				
2	231.89	3.86	1.016	2.67 @ 14.0				
3	231.89	3.86	0.838	1.51 @ 10.7				
4	231.89	3.86	0.658	4.63 @ 5.9				
5	231.89	3.86	0.510	5.01 @ 4.9				

Fault: B

SLG_A_R30 at temporary bus 999001 Bus_132Gelp (NEWBUS1) Midline node on "39 Bus_132Gelp" to "40 Bus_132Ting" Ckt 1 "999001 Bus_132Gelp" (NEWBUS1) distant 0.499 from "39 Bus_132Gelp"

Curve	Cu Primary A	rrent A/Pickup	Operating Seconds	Source/Total line (+ seq SIR)	
1 2 3 4 5	273.84 381.21 381.21 381.21 381.21 781.25	2.74 6.35 6.35 6.35 13.02	1.540 0.740 0.610 0.480 Infinite	3.24 @ 21.5 2.45 @ 14.3 1.42 @ 10.7 4.46 @ 5.7 1.70 @-171.7	

Fault: C

SLG_A_R30 at temporary bus 999001 Bus_132Ting (NEWBUS1) Midline node on "40 Bus_132Ting" to "41 Bus_132Nglam" Ckt 1 "999001 Bus 132Ting" (NEWBUS1) distant 0.500 from "40 Bus 132Ting"

Curve	Current		Operating	Source/Total	line (+ seq SIR)
	Primary A	A/Pickup	Seconds		
1	428.79	4.29	1.060	3.24 @ 21.5	
2	618.83	10.31	0.584	2.35 @ 14.1	
3	618.83	10.31	0.480	1.38 @ 10.5	
4	454.34	7.57	Infinite	2.95 @-174.4	
5	454.34	7.57	Infinite	1.74 @-171.5	
3 4	618.83 454.34	10.31 7.57	0.480 Infinite	1.38 @ 10.5 2.95 @-174.4	

Fault: D

SLG_A_R30 at temporary bus 999001 Bus_132Nglam (NEWBUS1) Midline node on "41 Bus_132Nglam" to "42 Bus_132Nang" Ckt 1 "999001 Bus_132Nglam" (NEWBUS1) distant 0.500 from "41 Bus_132Nglam"

Curve	Cu Primary A	rrent A/Pickup	Operating Seconds	Source/Total	line (+ seq SIR)
1	739.98	7.40	0.768	3.24 @ 21.5	
2	1082.77	18.05	0.468	2.31 @ 14.0	
3	262.21	4.37	Infinite	2.63 @-176.1	
4	262.21	4.37	Infinite	3.03 @-174.3	
5	262.21	4.37	Infinite	1.82 @-171.4	

Fault: E

SLG_A_R30 at temporary bus 999001 Bus_132Nang (NEWBUS1) Midline node on "42 Bus_132Nang" to "45 Bus_132KHP" Ckt 1 "999001 Bus 132Nang" (NEWBUS1) distant 0.501 from "42 Bus 132Nang"

Curve	Cu Primary A	rrent A/Pickup	Operating Seconds	Source/Total	line (+ seq SIR)
1 2 3 4	1197.90 136.28 136.28 136.28	11.98 2.27 2.27 2.27	0.616 Infinite Infinite Infinite	3.24 @ 21.5 7.44 @-177.1 2.66 @-176.1 3.08 @-174.3	
5	136.28	2.27	Infinite	1.86 @-171.6	



Coordination curve for new setting:

220kV feeders, 67N coordination of REL 511 simulation result:

Existing Setting:

Fault: A

SINGLE_LINE_GROUND at temporary bus 999001 Bus_220Semto (NEWBUS1) Midline node on "4 Bus_220Semto" to "13 Bus_220Ruri" Ckt 1 "999001 Bus_220Semto" (NEWBUS1) distant **0.951 from "4 Bus_220Semto**"

Curve	Curr Primary A		Operating Seconds	Source/Total line (+ seq SIR)
1	1318.50	17.58	0.260	2.64 @ 3.1
2	1293.31	10.78	0.374	0.72 @ 7.8
3	200.50	1.25	4.642	3.43 @ 10.1

Fault: B

SINGLE_LINE_GROUND at temporary bus 999001 Bus_220Semto (NEWBUS1) Midline node on "4 Bus_220Semto" to "13 Bus_220Ruri" Ckt 1 "999001 Bus_220Semto" (NEWBUS1) distant **0.066 from "4 Bus_220Semto**"

Curve	Current		Operating	Source/Total line (+ se			seq SIR)
	Primary A	A/Pickup	Seconds				
1	2201.49	29.35	0.220	2.58 @	3.0		
2	2106.03	17.55	0.308	0.71 @	7.7		
3	326.10	2.04	1.464	3.41 @	9.8		

Fault: C

SINGLE_LINE_GROUND at Midline node on "4 Bus_220Semto" to "2
Bus_220CHP" Ckt 1
"999001 Bus 220Semto" (NEWBUS1) distant 0.074 from "4 Bus_220Semto"

Curve	Curr	ent	Operating	Source/Total line (+ seq SIR)
	Primary A	A/Pickup	Seconds	
1	1281.61	17.09	Infinite	13.06 @-171.7
2	2349.92	19.58	0.296	0.71 @ 7.6
3	363.09	2.27	1.270	3.40 @ 9.7

Fault: D

SINGLE_LINE_GROUND at Midline node on "4 Bus_220Semto" to "2
Bus_220CHP" Ckt 1
"999001 Bus_220Semto" (NEWBUS1) distant 0.918 from "4 Bus_220Semto"

Curve	Curr Primary A		Operating Seconds	Source/Total line (+ seq SIR)
1	672.50	8.97		13.58 @-170.4
2 3	7484.61 1138.50	62.37 7.12	0.256 0.524	0.69 @ 7.4 3.30 @ 9.0

Fault: E

SINGLE_LINE_GROUND at temporary bus 999001 Bus_220CHP (NEWBUS1) Midline node on "2 Bus_220CHP" to "20 Bus_220Malbe" Ckt 1 "999001 Bus 220CHP" (NEWBUS1) distant **0.052 from "2 Bus_220CHP**"

Curve	Curr Primary A		Operating Seconds	Source/Total line (+ seq SIR)
1	515.03	6.87		13.72 @-170.1
2 3	521.57 1578.26	4.35 9.86	Infinite 0.448	9.20 @-172.0 3.12 @ 8.9

Fault: F

SINGLE_LINE_GROUND at temporary bus 999001 Bus_220CHP (NEWBUS1) Midline node on "2 Bus_220CHP" to "20 Bus_220Malbe" Ckt 1 "999001 Bus 220CHP" (NEWBUS1) distant 0.928 from "2 Bus 220CHP"

Curve	Curr Primary A		Operating Seconds	Source/Total line (+ seq SIR)
1	243.14	3.24	Infinite	13.92 @-169.4
2	242.22	2.02	Infinite	9.73 @-170.4
3	4784.30	29.90	0.298	1.90 @ 7.5

New Setting Simulation Result:

Fault: A SINGLE_LINE_GROUND at temporary bus 999001 Bus_220Semto (NEWBUS1) Midline node on "4 Bus_220Semto" to "13 Bus_220Ruri" Ckt 1 "999001 Bus_220Semto" (NEWBUS1) distant 0.951 from "4 Bus_220Semto"

Curve	Curr	ent	Operating	Source/Total line (+ seq SIR)
	Primary A	A/Pickup	Seconds	
_				
1	1318.50	21.97	0.220	2.64 @ 3.1
2	1293.31	14.37	0.282	0.72 @ 7.8
3	200.50	1.67	1.628	3.43 @ 10.1
4	1318.50	0.99	Infinite	Unavailable
5	1293.31	0.56	Infinite	Unavailable
6	200.50	0.06	Infinite	Unavailable

Fault: B

SINGLE_LINE_GROUND at temporary bus 999001 Bus_220Semto (NEWBUS1) Midline node on "4 Bus_220Semto" to "13 Bus_220Ruri" Ckt 1 "999001 Bus 220Semto" (NEWBUS1) distant 0.041 from "4 Bus 220Semto"

Curve	Curr Primary A		Operating Seconds	Source/Total	line (+ seq SIR)
1	2234.99	37.25	0.196	2.57 @ 3.0	
2	2137.06	23.75	0.236	0.71 @ 7.7	
3	330.90	2.76	0.820	3.41 @ 9.8	
4	2234.99	1.67	0.020	Unavailable	
5	2137.06	0.92	Infinite	Unavailable	
6	330.90	0.10	Infinite	Unavailable	

Fault: C

SINGLE_LINE_GROUND at temporary bus 999001 Bus_220Semto (NEWBUS1) Midline node on "4 Bus_220Semto" to "2 Bus_220CHP" Ckt 1 "999001 Bus_220Semto" (NEWBUS1) distant 0.058 from "4 Bus_220Semto"

Curve	Curr Primary A		Operating Seconds	Source/Total line (+ seq SIR)
1	1291.63	21.53	Infinite	13.06 @-171.7
2	2312.98	25.70	0.230	0.71 @ 7.7
3	357.54	2.98	0.762	3.40 @ 9.7
4	1291.63	0.97	Infinite	Unavailable
5	2312.98	1.00	Infinite	Unavailable
6	357.54	0.11	Infinite	Unavailable

Fault: D SINGLE_LINE_GROUND at temporary bus 999001 Bus_220Semto (NEWBUS1) Midline node on "4 Bus_220Semto" to "2 Bus_220CHP" Ckt 1 "999001 Bus_220Semto" (NEWBUS1) distant 0.947 from "4 Bus_220Semto"

Curve	Curr Primary A		Operating Seconds	Source/Total line (+ seq SIR)
	<u> </u>	10 57		12 62 0 170 2
\perp	634.43	10.57	Infinite	13.62 @-170.3
2	8002.08	88.91	0.216	0.69 @ 7.4
3	1216.77	10.14	0.354	3.30 @ 9.0
4	634.43	0.47	Infinite	Unavailable
5	8002.08	3.45	0.020	Unavailable
6	1216.77	0.39	Infinite	Unavailable

Fault: E

SINGLE_LINE_GROUND at temporary bus 999001 Bus_220CHP (NEWBUS1) Midline node on "2 Bus_220CHP" to "20 Bus_220Malbe" Ckt 1 "999001 Bus_220CHP" (NEWBUS1) distant 0.024 from "2 Bus_220CHP"

Curve	Curr Primary A		Operating Seconds	Source/Total line (+ seq SIR)	
1	534.31	8.91	Infinite	13.71 @-170.2	-
2	541.25	6.01	Infinite	9.19 @-172.1	
3	1481.06	12.34	0.326	3.21 @ 8.9	
4	534.31	0.40	Infinite	Unavailable	
5	541.25	0.23	Infinite	Unavailable	
6	1481.06	0.47	Infinite	Unavailable	

Fault: F

SINGLE_LINE_GROUND at temporary bus 999001 Bus_220CHP (NEWBUS1) Midline node on "2 Bus_220CHP" to "20 Bus_220Malbe" Ckt 1 "999001 Bus 220CHP" (NEWBUS1) distant 0.959 from "2 Bus 220CHP"

Curve	Curr Primary A		Operating Seconds	Source/Total line (+ seq SIR)
1	239.00	3.98	Infinite	13.93 @-169.3
2	237.84	2.64	Infinite	9.76 @-170.3
3	4954.75	41.29	0.236	1.88 @ 7.5
4	239.00	0.18	Infinite	Unavailable
5	237.84	0.10	Infinite	Unavailable
6	4954.75	1.57	0.020	Unavailable

APPENDIX [G]: Propose relay setting details:

66kV feeders:

Location	of strength and		narks ction Data					1	Devic Relay	e Tag: Tag:		69 67	I	- Archived
Substa		Chu			C	han	je LZOP		Devic	e Nam	0	78	4612_Chun	ndo fdr.
LZOPI	D	LINE	1.0.17		-							-	-	
Name		-	/ Chumdo feeder				LZOP		Activ	e Grou	P			
LZOPI	Raink	1	_			view	1200							<< Advanced
Group Dis	played				ž]	Rename		Сору		Delete		Compare	Move
Relay St	yle	7SAE	6xx_V4.3_1A	•	Sel	lect	Style		mufact	10000	SIEMEN	s		
Scher	ne	MAIN	1	٠					day Mod degory:		7SA6 Digital P	acka	ige	
Relay Info	Elem	ents	Common Taps	Miscella	neous	Me	mos							
Find	Tap Na	me			Find		Filter 1	Taps	by Fun	ction				*
Type	Numb	Tap	Name		Setting			Re	marks	Rang	,		Tap Descri	ption
TEXT	110	1301	1 Op. mode Z1		Forwar	đ				Fwd/R	tev/Non/Ir	nact		
NUM	111	1302	2 R(Z1) ph-ph		5,71					0.05-8	00 ohms			
NUM	112	1303	3 X(Z1)		6,168					0.05-6	00 ohms	È.		
NUM	113	1304	4 RE(Z1) ph-E		13,21					0.05-8	00 ohms	6		
BOTH	114	1305	5 T1-1phase		0					0-30 s	ec			
BOTH	115	1308	6 T1-multi-phase		0					0-30 s	ec			
NUM	116	1307	7 Zone Reduction	1	0					0-45 0	leg.			
TEXT	117	1311	1 Op. mode Z2		Forward	d				Fwd/F	tewNon/In	nact		
NUM	118	1312	2 R(Z2) ph-ph		7,142					0.05-6	00 ohms	i		
NUM	119	1313	3 X(Z2)		8,92					0.05-8	00 ohms	£1.		
NUM	120	1314	4 RE(Z2) ph-E		14,642					0.05-8	00 ohms	È		
BOTH	121	1315	5 T2-1phase		0,4					0-30 5	ec			
BOTH	122	1318	6 T2-multi-phase		0					0-30 \$	ec			
TEXT	123	1317	7 Trip 1 pole Z2		NO					NOME	S			
TEXT	124	1321	1 Op. mode Z3		Forwar	d				Fwd/F	ew/Non/In	hact		
NUM	125	1322	2 R(Z3) ph-ph		10,136					0.05-6	00 ohms	8		
NUM		1323	3 X(Z3)		14,674					0.05-8	00 ohms			
NUM	127	1324	4 RE(Z3) ph-E		17,636					0.05-6	i00 ohms	1		
BOTH	128	1325	5 T3 DELAY		0,8			-		0-30 \$				
TEXT		1331	1 Op. mode Z4		Revers	8				Fwd/F	ev/Non/Ir	nact		
NUM		1332	2 R(Z4) ph-ph		0,571			1		0.05-6	00 ohms	6		
NUM			3 X(Z4)		0,617			-			i00 ohms			
NUM	132	1334	4 RE(Z4) ph-E		1,321					0.05-6	i00 ohms	É.,		
NUM	69	1118	RE/RL(Z1)		1,19			T		-0.33-1	1			
NUM			XEML(Z1)		1,48			1		-0.33-1				
NUM			REAL(ZIB Z5)		1,19					-0.33-1				
NUM		1000	XE/XL(Z18		1,48			-		-0.33-1				

location Devic	e Remarks Protection Data			Device Relay	0.000	68 66		Archived
Substation	Chumdo	Cha	nge LZOP		Name	-	11_CHP Fo	4
LZOP ID	LINE 10					Press.		
Name	66kV Chukha feeder			Active	Group	1		
LZOP Rask	1	Vie	WLZOP					< Advanced
Group Displayed			Rename	Copy	Delet	e (Compare	Move
Relay Style	REL511_V2.3_1A	• Selec	t Style	Manufactu Relay Mode				
Scheme	MAN	2		Category:		Packag	6	
Relay Info Elen	ents Common Taps	Aiscellaneous M	emos					
Find Tap Na	ene -	Find	Filter T	aps by Func	tion			1
Type Numb	Tap Name	Setting		Remarks	Range	T	o Descripti	00
TEXT 107	Operation_Z1	Forward			Zone Operat	ion		
TEXT 108	OperationPP_21	ON		-	OFFION			
NUM 109	X1PP_Z1	6,17			0 1-400 ohm	6		
NUM 110	R1PP_Z1	3,21			0.1-400 ohm	5		
NUM 111	RFPP_Z1	22,23			0.1-400 ohm	8		
TEXT 112	Timer t1PP	ON			OFF/ON			
NUM 113	tIPP	0			0-60 sec	_		
TEXT 114	OperationPE_Z1	ON			OFFION			
NUM 115	X1PE_Z1	6,17			0.1-400 ohm			
NUM 116	R1PE_21	3,21			0.1-400 ohm			
NUM 117	XOPE_Z1	33,50			0.1-1200 city			
NUM 118	ROPE_Z1	14,64			0.1-1200 oh			
NUM 119	RFPE_Z1	18,75			0.1-400 ohm	5		
TEXT 120	Timer ti PE	ON			OFF/ON			
NUM 121	tiPE	0			0-60 sec	inia.		
TEXT 122	Operation_Z2	Forward			Zone Operat	ion		
TEXT 123 NUM 124	OperationPP_Z2	9,9			0FF/0N 0.1-400 ohm			
NUM 125	X1PP_Z2 R1PP_Z2	5,15			0.1-400 ohm			
NUM 126	RFPP_Z2	26,79			0.1-400 ohm			
TEXT 127	Timer 12PP	ON			OFFION			
NUM 128	12PP	0,4			0-60 sec			
TEXT 129	OperationPE_Z2	ON			OFF/ON	_		
							_	
NUM 130	X1PE_Z2	9,9			0 1-400 ohm			
NUM 131 NUM 132	R1PE_Z2 X0PE_Z2	5,15			0.1-400 ohm			
NUM 133		53,89			0.1-1200 ohr 0.1-1200 ohr			
NUM 134	ROPE_ZZ RFPE_ZZ	23,5 20,15			0.1-400 ohm			
TEXT 135	Timer t2PE	20,15 ON			OFFION			
NUM 136	12PE	0,4			0-60 sec			
TEXT 137	Operation_Z3	Forward			Zone Operati	on		
TEXT 138	OperationPP_Z3	ON			OFFION			
NUM 139	X1PP_Z3	12,09			0.1-400 ohm	6		
NUM 140	R1PP_Z3	6,29			0.1-400 ohm			
NUM 141	REPP_Z)	26,84			0 1-400 ohm	_		
TEXT 142	Timer t3PP	ON			OFF/ON			
NUM 143	13PP	0,8			0-60 sec			
TEXT 144	OperationPE_Z3	ON			OFF/ON			
NUM 145	X1PE_Z3	12,09			0 1-400 ohm	5		
NUM 146	R1PE_23	6,29			0.1-400 ohm	6		
NUM 147	XOPE_ZJ	65,81			0 1-1200 ohr			
NUM 148	ROPE_Z)	28,7			0.1-1200 ohr			
NUM 149	RFPE_Z3	23,44			0.1-400 ohm	\$		
TEXT 150	Timer t3PE	ON			OFF/ON			
NUM 151	13PE	0,8			0-60 sec			
TEXT 152	Operation_Z4	Reverse			Zone Operati	20		

ocation Devic	Second			Device		0.50	Archived
	Protection Data	1 Annual		Relay 1	lag	68	
Substation LZOP ID	Chumdo LINE 11	Cha	nge LZOP	Device	Name	RELS11_Paro F	dr.
Name	66kV Paro Feeder			Active	Group		
LZOP Rank	1	Vie	W LZOP				<< Advanced
aroup Displayed	[-	Rename	- Com	Delete	Compare	Move
woodp Unspeayed	-		Pottsatie	Copy		Compare	anove
Relay Style	REL511_V2.3_1A MAIN	· Selec	t Style	Manufactur Relay Mode	C RELSIS	And the second se	
Scheme	ents Common Taps	iii Incollectore 11	1	Category:	Digital F	аскаде	
Find Tap Na		Find	Cold Street St	aps by Funct	ion		
Type Numb	Tan Name	Setting		Remarks	Rance	Tap Descript	A REAL PROPERTY AND INCOME.
TEXT 107	Operation_Z1	Forward			Zone Operatio		ENCL.
TEXT 108	OperationPP_Z1	ON			OFFION		
NUM 109	X1PP_Z1	4,01			0.1-400 ohms		
NUM 110	R1PP_Z1	2,08			1-400 ohms		
NUM 111	REPP_Z1	5			1-400 ohms		
TEXT 112	Timer ti PP	ON			OFFION		
NUM 113	tIPP	0		()-60 sec		
TEXT 114	OperationPE_Z1	ON		(OFF/ON		
NUM 115	X1PE_Z1	4,01		(0.1-400 ohms		
NUM 116	R1PE_Z1	2,08		(0.1-400 ohms		
NUM 117	X0PE_Z1	21,8			0.1-1200 ohm		
NUM 118	ROPE_Z1	9,51) 1-1200 ohm	8	
NUM 119	RFPE_Z1	4,35			0.1-400 ohms		
TEXT 120	Timer t1PE	ON			DEEJON	_	
NUM 121	TIPE	0			3-60 sec		
TEXT 122	Operation_Z2	Forward			Zone Operatio	n	
TEXT 123	OperationPP_Z2	ON			OFF/ON	-	
NUM 124 NUM 125	X1PP_Z2	11,01			0.1-400 ohms	_	
NUM 125	R1PP_Z2 RFPP_Z2	5,73			0.1-400 ohms 0.1-400 ohms		
TEXT 127	Timer t2PP	ON			OFFION		
NUM 128	t2PP	0,4			3-60 sec		
TEXT 129	OperationPE_Z2	ON			OFF/ON		
						10 E	
NUM 130 NUM 131	X1PE_Z2	11,01			0.1-400 ohms		
NUM 131	R1PE_Z2 X0PE_Z2	5,73 59,95			0.1-400 ohms 0.1-1200 ohm		
NUM 133	ROPE_Z2	26,15			0.1-1200 ohm		
NUM 134	RFPE_Z2	5,01			0.1-400 ohms		
TEXT 135	Timer t2PE	ON			OFFION		
NUM 136	12PE	0,4			0-60 sec		
TEXT 137	Operation_Z3	Forward			Zone Operatio	n	
TEXT 138	OperationPP_Z3	ON			OFF/ON		
NUM 139	X1PP_Z3	12,52			0.1-400 ohms		
NUM 140	R1PP_Z3	6,51		3	0.1-400 ohms		
NUM 141	RFPP_Z3	5,65			0.1-400 ohms		
TEXT 142	Timer t3PP	ON			OFFION		
NUM 143	13PP	0,8			0-60 sec		
TEXT 144	OperationPE_Z3	ON		1	OFF/ON		
NUM 145	X1PE_Z3	12,52			0.1-400 ohms		
NUM 146	R1PE_Z3	6,51			0 1-400 ohms		
NUM 147	X0PE_Z3	68,13			0.1-1200 ohm		
NUM 148	ROPE_Z3	29,71			0.1-1200 ohm		
NUM 149	RFPE_Z3	5,02			0.1-400 ohms	1	
TEXT 150	Timer t3PE	ON		1. 3	OFFION		
NUM 151	13PE	0,8			0-60 sec		

	Comparison of the	e Remarks Protection Data			Device Tag: Relay Tag:	71 69			
Subst		Citumdo	Cha	nge LZOP	Device Name	R	EL511_Haa fee	der	
LZOPI	D	LINE 12 66kV Haa feeder			Active Group	- in		200	
LZOPI		1	Vie	WLZOP	weave or outp				
Law		er.		weiter]			<	< Advance	
oup Dis	playe	1		Rename	Copy D	elete	Compare	Move	
Relay S	tyle	REL511_V2.3_1A	. Selec	t Style	Manufacturer: AB Relay Model: RE	0 L511_V			
Schee	ne	MAIN	-			gital Pack			
Any heto	Elen	nents Common Taps	Miscellaneous M	lemos					
and the second se	Tap Na	1000	Find	Contraction of the second	aps by Function	-			
	0.000	Tap Name	Setting		Remarks Range	-	Tap Description		
TEXT		Operation_Z1	Forward		Zone Op	eration .	THE CHARGE	20.	
TEXT		OperationPP_Z1	ON		OFFJON				
NUM	Contraction of the second	X1PP_Z1	5,57		0.1-400 0	hms			
NUM		R1PP_Z1	2,9		0.1-400 0				
NUM		RFPP_Z1	5,2		0.1-400 c	hms			
TEXT	112	Timer tt PP	ON		OFFION				
NUM	113	t1PP	0		0-60 sec				
TEXT	114	OperationPE_Z1	ON		OFF/ON				
NUM		X1PE_Z1	5,57		0.1-400 c	ihms			
NUM	and the second	R1PE_Z1	2,9		0.1-400 ¢	nms			
NUM	1.11.1	X0PE_Z1	30,33		0.1-1200	ohms			
NUM		ROPE_Z1	13,23		0.1-1200				
NUM		RFPE_Z1	4,15		0.1-400 0	hms			
TEXT		Timer ti PE	ON		OFFION				
NUM		tIPE	0		0-60 sec				
TEXT	Contraction of the local distance of the loc	Operation_22	Forward		Zone Op	eration			
TEXT		OperationPP_Z2	ON		OFFION		-		
NUM		X1PP_Z2	15,32		0.1-400 0				
NUM	1.	R1PP_Z2	7,97		0.1-400 0				
NUM		RFPP_Z2	6,38		0.1-400 (hms			
TEXT	1.	Timer 12PP 12PP	ON		OFFION				
NUM		144.1.7	0,4 ON		0-60 sec OFF/ON				
TEXT	_	OperationPE_Z2							
NUM	100 million	X1PE_Z2	15,32		0.1-400 (
NUM	0.000	R1PE_Z2	7,97		0.1-400 (-		
NUM		X0PE_Z2	83,4		0.1-1200		-		
NUM		ROPE_Z2	36,37		0.1-1200				
NUM		RFPE_Z2	5,2 ON		0.1-400 0	mins	-		
NUM	and the second sec	Timer t2PE t2PE	0,4		OFF/ON 0-60 sec		-		
TEXT		Operation_Z3	Forward		Zone Op		-		
TEXT		OperationPP_Z3	ON		OFFION	er aurori			
NUM		X1PP_Z3	17,41		0.1-400 0	hms	1		
NUM		R1PP_Z3	9,06		0.1-400 0				
NUM		RFPP_Z3	6,32		0.1-400 0				
TEXT		Timer t3PP	ON		OFF/ON	1111			
NUM		13PP	0,8		0-60 sec				
TEAT		OperationPE_Z3	ON		OFF/ON				
NUM		X1PE_Z3	17,41		0.1-400 0	hms			
NUM		R1PE_Z3	9,05		0.1-400 0				
NUM		XOPE_Z3	94,77		0.1-1200	ohms			
NUM		ROPE_Z3	41,33		0.1-1200	ohms			
NUM	149	RFPE_Z)	5,2		0.1-400 0	hmis			
TEXT	150	Timer t3PE	ON		OFF/ON				
NUM	151	13PE	0,8		0-60 sec				
Conception and	152	Operation_Z4	Reverse		Zone Op	and Maker			

Local Zone of	e Remarks Protection Data			Device 1 Relay Ta	0.70	67 F 65	Archived
Substation LZOP ID	Chumdo LINE 13	Cha	nge LZOP	Device	Lame	REL511_Jenim	a feèder
Name	66kV Jemina feeder			Active G	roup		
LZOP Rank	1	Vie	WLZOP				<< Advanced
Group Displayed			Rename	Copy	Delete	Compare	Move
Relay Style	REL511_V2.3_1A	• Selec	t Style	Manufacture Relay Model		1/2	
Scheme	MAIN	-		Category:	Digital P		
Contraction of Contract	sents Common Taps	and the second se	1111110				
Find Tap Na	and the second se	Find	1 Patter 1	aps by Functio			-
Type Numb		Setting	-	Remarks R		Tap Descrip	tion
TEXT 107	Operation_Z1	Forward			one Operatio	n	
TEXT 108	OperationPP_Z1	ON			FFION		
NUM 109	X1PP_Z1	1,94		1.	1-400 ohms		
NUM 110	R1PP_Z1	1,01			1-400 ohms		
NUM 111	RFPP_Z1	12,71			1-400 ohms	_	
TEXT 112	Timer tIPP tIPP	0			FFJON		
NUM 113 TEXT 114	OperationPE_Z1	ON			60 sec FFJON		
NUM 115	X1PE_Z1	1,94			1-400 ohms		
NUM 116	RIPE_ZI	1,01			1-400 ohms		
NUM 117	XOPE_Z1	10,53			1-1200 ohm		
NUM 118	ROPE_Z1	4,59		1	1-1200 ohm:		
NUM 119	RFPE_Z1	10,85			1-400 ohms		
TEXT 120	Timer ti PE	ON			FFJON		
NUM 121	tIPE	0			60 sec		
TEXT 122	Operation_Z2	Forward			one Operation	n	
TEXT 123	OperationPP_Z2	ON			FFION		
NUM 124	X1PP_Z2	4,37		12	1-400 ohms		
NUM 125	R1PP_Z2	2,27			1-400 ohms		
NUM 126	RFPP_Z2	14,3			1-400 ohms		
TEXT 127	Timer 12PP	ON		0	FFJON		
NUM 128	t2PP	0,4		0-	60 sec		
TEXT 129	OperationPE_Z2	ON		0	FFION		
NUM 130	X1PE_Z2	4,37		0	1-400 ohms		
NUM 131	R1PE_Z2	2,27			1-400 ohms		
NUM 132	XOPE_Z2	23,78			1-1200 ohm		
NUM 133	ROPE_Z2	10,37			1-1200 ohms		
NUM 134	RFPE_Z2	11,73			1-400 ohms		
TEXT 135	Timer t2PE	ON			F/ON		
NUM 136	12PE	0,4		0-	60 sec		
TEXT 137	Operation_23	Forward		Zo	ne Operation	n	
TEXT 138	OperationPP_Z3	ON			FFION		
NUM 139	X1PP_Z3	6,32		0.	1-400 ohms		
NUM 140	R1PP_Z3	3,29		0	1-400 ohms		
NUM 141	RFPP_Z3	13,73			1-400 ohms		
TEXT 142	Timer t3PP	ON			FF/ON		
NUM 143	t3PP	0,8			60 sec	-	
TEXT 144	OperationPE_Z3	ON			FRON		
NUM 145	X1PE_Z3	6,32			1-400 ohms	_	
NUM 146	R1PE_Z3	3,29			1-400 ohms		
NUM 147	XOPE_Z3	34,39			1-1200 ohm		
NUM 148	ROPE_Z3	15		1.00	1-1200 ohm:		
NUM 149	RFPE_Z3	15,03			1-400 ohms	-	
TEXT 150 NUM 151	Timer t3PE t3PE	0N 0,8			FFION	-	
	LIPE				60 sec		

LZOP ID LINE 13 Name 66kV Chumdo feeder LZOP Rank 2 Group Displayed Relay Style PD532_1A Scheme MAIN	rmdoFdr(New) Advanced re Move
Name 66kV Chumdo feeder LZOP Rank 2 View LZOP Group Displayed Rename Copy Delete Compa Relay Style PD532_1A Select Style Manuffacturier: Relay Model: ALSTOM PD532 Category: Digital Package Scheme MAIN Select Style Manuffacturier: Relay Model: ALSTOM PD532 Category: Digital Package Find Find Fitter Taps by Function Tap Des Type Numbel Tap Name Find Fitter Taps by Function Text 737 DIST Characteristic Polygon Circle/Polygon TEXT 738 DIST Op mode zone 4 Normal Zone 4 Op, Mode NUM 740 DIST X1 (polygon) 0,13 0.1-200 ohms NUM 743 DIST R1PP (polygon) 4,63 0.1-200 ohms NUM 745 DIST R2PP (polygon) 4,63 0.1-200 ohms NUM 744 DIST R3PP (polygon) 4,63 0.1-200 ohms NUM 745 DIST R3PP (polygon) 4,63 <th>-</th>	-
Relay Style PD532_1A Select Style Matuffacturer: Relay Model: Category: ALSTOM PD532 Digital Package Relay Info Elements Common Taps Miscellaneous Memos Distribution Find Tap Name Find Fitter Taps by Function Find Fitter Taps by Function Type Numbel Tap Name Setting Remarks Range Tap Des TEXT 737 DIST Characteristic Polygon Circle/Polygon TEXT 738 DIST Op mode zone 4 Normal Zone 4 Op. Mode NUM 799 DIST <x1 (polygon)<="" td=""> 1,4 0.1-200 ohms Nums NUM 740 DIST X2 (polygon) 6,75 0.1-200 ohms Nums NUM 741 DIST X1 (polygon) 0,13 0.1-200 ohms NUM NUM 743 DIST R1,PP (polygon) 4,65 0.1-200 ohms NUM NUM 745 DIST R2,PP (polygon) 4,63 0.1-200 ohms NUM NUM 745 DIST R2,PP (polygon) 4,64<!--</th--><th>-</th></x1>	-
Relay Style PD532_1A Select Style Matuffacturer: Relay Model: Category: ALSTOM PD532 Digital Package Relay Info Elements Common Taps Miscellaneous Memos Distribution Find Tap Name Find Fitter Taps by Function Find Fitter Taps by Function Type Numbel Tap Name Setting Remarks Range Tap Des TEXT 737 DIST Characteristic Polygon Circle/Polygon TEXT 738 DIST Op mode zone 4 Normal Zone 4 Op. Mode NUM 799 DIST <x1 (polygon)<="" td=""> 1,4 0.1-200 ohms Nums NUM 740 DIST X2 (polygon) 6,75 0.1-200 ohms Nums NUM 741 DIST X1 (polygon) 0,13 0.1-200 ohms NUM NUM 743 DIST R1,PP (polygon) 4,65 0.1-200 ohms NUM NUM 745 DIST R2,PP (polygon) 4,63 0.1-200 ohms NUM NUM 745 DIST R2,PP (polygon) 4,64<!--</th--><th></th></x1>	
Twining Signe PDS32_1A Select Signe Relay Model: PDS32 Scheine MAIN Category: Digital Package Relay Info Elements Common Taps Miscellaneous Memos Find Tap Name Find Filter Taps by Function Type Numbel Tap Name Setting Remarks Range Tap Des TEXT 737 DIST Characteristic Polygon ClickelPolygon TEXT 738 DIST Op mode zone 4 Normal Zone 4 Op. Mode NUM 740 DIST X1 (polygon) 1,4 0.1-200 ohms NUM 742 DIST X3 (polygon) 6,75 0.1-200 ohms NUM 743 DIST R1.PG (polygon) 4,63 0.1-200 ohms NUM 745 DIST R2.PG (polygon) 4,65 0.1-200 ohms NUM 745 DIST R2.PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2.PG (polygon) 4,65	ie sove
Scheme MAIN Category: Digital Package Relay Info Elements Common Taps Miscellaneous Memos Find Tap Name Find Find Filter Taps by Function Type Numbe Tap Name Setting Remarks Range Tap Des TEXT 7.37 DIST Characteristic Polygon Circle/Polygon TEXT 7.38 DIST Op mode zone 4 Normal Zone 4 Op. Mode NUM 739 DIST X1 (polygon) 1,4 0.1-200 ohms NUM 740 DIST X2 (polygon) 8,61 0.1-200 ohms NUM 742 DIST X4 (polygon) 0,13 0.1-200 ohms NUM 743 DIST R1,PO (polygon) 4,63 0.1-200 ohms NUM 744 DIST R1,PP (polygon) 4,65 0.1-200 ohms NUM 745 DIST R2,PP (polygon) 4,63 0.1-200 ohms NUM 746 DIST	
Type Numbe Tap Name Setting Remarks Range Tap Des TEXT 7.37 DIST Characteristic Polygon Circle/Polygon TEXT 7.38 DIST Op. mode zone 4 Normal Zone 4 Op. Mode NUM 739 DIST X1 (polygon) 1,4 0.1-200 ohms NUM 740 DIST X2 (polygon) 3,61 0.1-200 ohms NUM 741 DIST X3 (polygon) 6,75 0.1-200 ohms NUM 742 DIST X4 (polygon) 0,13 0.1-200 ohms NUM 743 DIST R1.PG (polygon) 4,63 0.1-200 ohms NUM 744 DIST R1.PG (polygon) 4,65 0.1-200 ohms NUM 745 DIST R2.PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2.PF (polygon) 4,63 0.1-200 ohms NUM 748 DIST R3.PF (polygon) 4,64 0.1-200 ohms NUM 748 DIST R3.PF (polygon) 4,64 0.1-200 ohms NUM 749 <t< th=""><th></th></t<>	
Find Tap NameFindFilter Taps by FunctionTypeNumbe Tap NameSettingRemarksRangeTap DesTEXT737DIST CharacteristicPolygonCircle/PolygonTEXT738DIST Op. mode zone 4NormalZone 4 Op. ModeNUM739DIST X1 (polygon)1,40.1-200 ohmsNUM740DIST X2 (polygon)3,610.1-200 ohmsNUM741DIST X3 (polygon)6,750.1-200 ohmsNUM742DIST R1 PG (polygon)4,630.1-200 ohmsNUM743DIST R1 PG (polygon)4,650.1-200 ohmsNUM745DIST R2 PG (polygon)4,630.1-200 ohmsNUM746DIST R2 PG (polygon)4,630.1-200 ohmsNUM748DIST R3 PF (polygon)4,640.1-200 ohmsNUM749DIST R3 PF (polygon)4,650.1-200 ohmsNUM749DIST R3 PF (polygon)4,650.1-200 ohmsNUM749DIST R3 PF (polygon)4,640.1-200 ohmsNUM749DIST R3 PF (polygon)4,650.1-200 ohmsNUM749DIST R4 PF (polygon)4,650.1-200 ohmsNUM749DIST R4 PF (polygon)4,650.1-200 ohmsNUM750DIST R4 PF (polygon)4,650.1-200 ohmsNUM750DIST R4 PF (polygon)4,650.1-200 ohms	
TEXT 737 DIST Characteristic Polygon Circle/Polygon TEXT 738 DIST Op. mode zone 4 Normal Zone 4 Op. Mode NUM 739 DIST X1 (polygon) 1,4 0.1-200 ohms NUM 740 DIST X2 (polygon) 3,61 0.1-200 ohms NUM 741 DIST X3 (polygon) 6,75 0.1-200 ohms NUM 742 DIST X4 (polygon) 0,13 0.1-200 ohms NUM 743 DIST R1.PG (polygon) 4,63 0.1-200 ohms NUM 744 DIST R1.PP (polygon) 4,65 0.1-200 ohms NUM 745 DIST R2.PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2.PP (polygon) 4,63 0.1-200 ohms NUM 746 DIST R3.PP (polygon) 4,64 0.1-200 ohms NUM 748 DIST R3.PP (polygon) 4,64 0.1-200 ohms NUM	
TEXT 737 DIST Characteristic Polygon Circle/Polygon TEXT 738 DIST Op. mode zone 4 Normal Zone 4 Op. Mode NUM 739 DIST X1 (polygon) 1,4 0.1-200 ohms NUM 740 DIST X2 (polygon) 3,61 0.1-200 ohms NUM 741 DIST X3 (polygon) 6,75 0.1-200 ohms NUM 742 DIST X4 (polygon) 0,13 0.1-200 ohms NUM 743 DIST R1.PG (polygon) 4,63 0.1-200 ohms NUM 744 DIST R1.PP (polygon) 4,65 0.1-200 ohms NUM 745 DIST R2.PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2.PP (polygon) 4,63 0.1-200 ohms NUM 746 DIST R3.PP (polygon) 4,64 0.1-200 ohms NUM 748 DIST R3.PP (polygon) 4,64 0.1-200 ohms NUM	
NUM 739 DIST X1 (polygon) 1,4 0.1-200 ohms NUM 740 DIST X2 (polygon) 3,61 0.1-200 ohms NUM 741 DIST X3 (polygon) 6,75 0.1-200 ohms NUM 742 DIST X3 (polygon) 6,75 0.1-200 ohms NUM 743 DIST R1,PG (polygon) 0,13 0.1-200 ohms NUM 744 DIST R1,PG (polygon) 4,63 0.1-200 ohms NUM 745 DIST R2,PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2,PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2,PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2,PG (polygon) 4,63 0.1-200 ohms NUM 747 DIST R3,PG (polygon) 4,63 0.1-200 ohms NUM 748 DIST R3,PF (polygon) 4,64 0.1-200 ohms NUM 749 DIST R4,PG (polygon) 4,64 0.1-200 ohms NUM 750 DIST R4,PF (polygon)	
NUM 740 DIST X2 (polygon) 3,61 0.1-200 ohms NUM 741 DIST X3 (polygon) 6,75 0.1-200 ohms NUM 742 DIST X4 (polygon) 0,13 0.1-200 ohms NUM 742 DIST X4 (polygon) 0,13 0.1-200 ohms NUM 743 DIST R1.PG (polygon) 4,63 0.1-200 ohms NUM 745 DIST R1.PG (polygon) 4,65 0.1-200 ohms NUM 745 DIST R2.PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2.PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2.PG (polygon) 4,63 0.1-200 ohms NUM 748 DIST R3.PG (polygon) 4,63 0.1-200 ohms NUM 748 DIST R3.PP (polygon) 4,64 0.1-200 ohms NUM 749 DIST R4.PG (polygon) 4,64 0.1-200 ohms NUM 750 DIST R4.PP (polygon) 4,65 0.1-200 ohms	
NUM 740 DIST X2 (polygon) 3,61 0.1-200 ohms NUM 741 DIST X3 (polygon) 6,75 0.1-200 ohms NUM 742 DIST X4 (polygon) 0,13 0.1-200 ohms NUM 742 DIST R1,PG (polygon) 4,63 0.1-200 ohms NUM 743 DIST R1,PG (polygon) 4,63 0.1-200 ohms NUM 744 DIST R1,PP (polygon) 4,65 0.1-200 ohms NUM 745 DIST R2,PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2,PP (polygon) 4,63 0.1-200 ohms NUM 746 DIST R3,PP (polygon) 4,63 0.1-200 ohms NUM 747 DIST R3,PP (polygon) 4,64 0.1-200 ohms NUM 748 DIST R4,PG (polygon) 4,64 0.1-200 ohms NUM 750 DIST R4,PP (polygon) 4,65 0.1-200 ohms	
NUM 741 DIST X3 (polygon) 6,75 0.1-200 ohms NUM 742 DIST X4 (polygon) 0,13 0.1-200 ohms NUM 743 DIST R1,PG (polygon) 4,63 0.1-200 ohms NUM 743 DIST R1,PG (polygon) 4,63 0.1-200 ohms NUM 744 DIST R1,PP (polygon) 4,65 0.1-200 ohms NUM 746 DIST R2,PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2,PP (polygon) 4,63 0.1-200 ohms NUM 747 DIST R3,PP (polygon) 4,63 0.1-200 ohms NUM 748 DIST R3,PP (polygon) 4,64 0.1-200 ohms NUM 749 DIST R4,PG (polygon) 4,64 0.1-200 ohms NUM 750 DIST R4,PP (polygon) 4,65 0.1-200 ohms	
NUM 743 DIST R1,PG (polygon) 4,63 0.1-200 ohms NUM 744 DIST R1,PP (polygon) 4,65 0.1-200 ohms NUM 745 DIST R2,PG (polygon) 4,63 0.1-200 ohms NUM 745 DIST R2,PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2,PP (polygon) 4,63 0.1-200 ohms NUM 747 DIST R3,PG (polygon) 4,63 0.1-200 ohms NUM 748 DIST R3,PP (polygon) 4,64 0.1-200 ohms NUM 749 DIST R4,PG (polygon) 4,64 0.1-200 ohms NUM 750 DIST R4,PP (polygon) 4,65 0.1-200 ohms	
NUM 744 DIST R1,PP (polygon) 4,65 0.1-200 ohms NUM 745 DIST R2,PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2,PP (polygon) 4,62 0.1-200 ohms NUM 747 DIST R3,PG (polygon) 4,63 0.1-200 ohms NUM 748 DIST R3,PP (polygon) 4,64 0.1-200 ohms NUM 749 DIST R4,PG (polygon) 4,64 0.1-200 ohms NUM 750 DIST R4,PP (polygon) 4,65 0.1-200 ohms	
NUM 745 DIST R2/PG (polygon) 4,63 0.1-200 ohms NUM 746 DIST R2/PP (polygon) 4,62 0.1-200 ohms NUM 747 DIST R3/PG (polygon) 4,63 0.1-200 ohms NUM 747 DIST R3/PG (polygon) 4,63 0.1-200 ohms NUM 748 DIST R3/PP (polygon) 4,64 0.1-200 ohms NUM 749 DIST R4/PG (polygon) 4,64 0.1-200 ohms NUM 750 DIST R4/PP (polygon) 4,65 0.1-200 ohms	
NUM 746 DIST R2,PP (polygon) 4,62 0.1-200 ohms NUM 747 DIST R3,PG (polygon) 4,63 0.1-200 ohms NUM 748 DIST R3,PP (polygon) 4,64 0.1-200 ohms NUM 749 DIST R4,PG (polygon) 4,64 0.1-200 ohms NUM 750 DIST R4,PP (polygon) 4,65 0.1-200 ohms	
NUM 747 DIST R3,PG (polygon) 4,63 0.1-200 ohms NUM 748 DIST R3,PP (polygon) 4,64 0.1-200 ohms NUM 749 DIST R4,PG (polygon) 4,64 0.1-200 ohms NUM 750 DIST R4,PP (polygon) 4,65 0.1-200 ohms	
NUM 748 DIST R3,PP (polygon) 4,64 0.1-200 ohms NUM 749 DIST R4,PG (polygon) 4,64 0.1-200 ohms NUM 750 DIST R4,PP (polygon) 4,65 0.1-200 ohms	
NUM 749 DIST R4,PG (polygon) 4,64 0.1-200 phms NUM 750 DIST R4,PP (polygon) 4,65 0.1-200 phms	
NUM 750 DIST R4.PP (polygon) 4.65 0.1-200 phms	
TEVT 750 DIGT Direction MI Executed directional Toolo Direction	
TEXT 769 DIST Direction N2 Forward directional Zone Dir.	
TEXT 770 DIST Direction N3 Forward directional Zone Dir.	
TEXT 771 DIST Direction N4 Backward directional Zone Dir.	
TEXT 772 DIST Direction N5 Forward directional Zone Dir.	
NUM 773 DIST Oper val Vmemory 0,01 0.01-1	
BOTH 774 DIST 11 0 0-10.0 s	
BOTH 775 DIST 12 0,4 0-10.0 s	
BOTH 776 DIST 13 0,8 0-10.0 s BOTH 777 DIST 14 1 0-10.0 s	

Local Zo	ne of F	rotect	tion Data				11	Device Ta Relay Tag		82 80		la chived
Substat LZOP ID		Jemir	2000		Cha	inge LZOP		Device N	1110	PC	0532_OlakhaFd	r(New)
Name	-	66kV	Jemina-Olakha feede	r at Jen	N			Active Gr	oup	Г		_
LZOPR	ank	2			V	ew LZOP					~	Advanced
Group Désp	dayed	Г			٠	Rename	T	Сору	Delet	•	Compare	Move
Relay Sty	Ae	PD53	2_1A		Selec	ct Style	_	ufacturer	100000			
Schem		MAIN		•				ay Model: egory:	PD532 Digital		age	
telay info	Elem	ents	Common Taps Mise	cellane	ous N	temos						
Find T				B	1000	A CONTRACTOR OF A CONTRACTOR O	Taps t	y Function		_		
Type N	lumbe	Tap N	lame	Se	ting		Ren	narks Ra	nge		Tap Descriptio	
TEXT 7			Characteristic		lygon		1		le/Polygo	n		
TEXT 7	38	DIST	Op. mode zone 4		ormal				e 4 Op. M			
NUM 7			X1 (polygon)	2,	02				200 ohm			
NUM 7	40		X2 (polygon)	3,					200 ohms			
NUM 7	41		X3 (polygon)	4,	39			0.1	200 ohms	1		
NUM 7	42	DIST	X4 (polygon)	0,	2			0.1	200 ohm:	1		
NUM 7	43	DIST	R1,PG (polygon)	5,	73			0.1	200 ohms	£		
NUM 7	44		R1,PP (polygon)	6,	5			0.1	200 ohm:	1		
NUM 7	45	DIST	R2,PG (polygon)	7,	38			0.1	200 ohm:	1		
NUM 7	46	DIST	R2,PP (polygon)	6,	15			0.1	200 ohm:	£		
NUM 7	47	DIST	R3,PG (polygon)	7,	9			0.1	200 ohm	i		
NUM 7	48	DIST	R3,PP (polygon)	6,	8			0.1	200 ohm:	ŧ		
NUM 7		DIST	R4,PG (polygon)	5,	82			0.1	200 ohms	\$\$		
NUM 7	50	DIST	R4,PP (polygon)	6,1	88			0.1	200 ohm:	1		
TEXT 7	68	DIST	Direction N1	Fo	rward d	lirectional		Zon	e Dir.			
TEXT 7	69	DIST	Direction N2	Fo	rward d	lirectional		Zon	e Dir.			
TEXT 7		DIST	Direction N3	Fo	rward d	lirectional		Zon	e Dir.			
TEXT 7	71	DIST	Direction N4	83	ckward	directional		Zon	e Dir.			
TEXT 7		DIST	Direction N5	Fo	rward d	lirectional		Zon	e Dir.			
NUM 7	73	DIST	Oper val. Vmemory	0,0	1			0.01	-1			
BOTH 7	74	DIST	ti	0				0-10	0.0 s			
BOTH 7	75	DIST	12	0,4				0-10	0.0 s			
BOTH 7	76	DIST	13	0,8	1			0-10).0 s			
BOTH 7	77	DIST	14	1				0-10	0.0 s			

Local Z	one of	e Remarks Protection Data			10743	vice Tag lay Tag:	: 81 79		Archived
Subst. LZOP		Olakha LINE 14	1	Change LZOP	De	vice Nan	10 7	SA611_Ola-Je	m(New)
Name		66kV Olakha-Jemi	na feeder		Ac	tive Grou	p [
LZOP	Rank	2		View LZOP					<< Advanced
Group Dis	played			• Rename	- Co	py	Delete	Compare	Move
Relay S	tyle	7SA6xx_V4.3_1A	× 5	elect Style	Manufa	000000000000	SIEMENS 7SA6		
Scher	ne	MAIN	•		Relay N Catego		Digital Pac	kage	
telay info	Bem	ents Common Ta	Miscellaneous	Memos					
Find	Tap Na	me	Find	Filter	Taps by F	unction			
		Tap Name	Settin	g	Remark	s Rang	e	Tap Descrip	tion
BOTH	a strand a strand	1241 R load (ph-E)		2	-		00 ohms		
NUM	1.	1242 phi load (ph-				20-60			
BOTH		1243 R load (ph-pl		2	-		00 ohms	-	
NUM		1244 phi load (ph-				20-60			
TEXT	111111	1301 Op. mode Z1					ReviNorvina 600 ohms	cu	
NUM		1302 R(Z1) ph-ph 1303 X(Z1)	8,772 6,238		-		600 ohms		
NUM		1304 RE(Z1) ph-E	11,89	ġ.			600 ohms		
BOTH	and the second second	1305 T1-1phase	0	0		0-30			
BOTH	100000000000000000000000000000000000000	1306 T1-multi-pha				0-30			
NUM		1307 Zone Reduct				0-45		5	
TEXT		1311 Op. mode Z2		rd			ReviNon/Ina	setting	
NUM		1312 R(Z2) ph-ph	10,43	5		0.05-	600 ohms	3	
NUM	110000000000000000000000000000000000000	1313 X(Z2)	9,912			0.05-	600 ohms	00	
NUM	120	1314 RE(Z2) ph-E	14,96	3		0.05-	600 ohms	modified	
BOTH	121	1315 T2-1phase	0,4			0-30	sec	di i	
BOTH	122	1316 T2-multi-pha	se 0,4		1	0-30	sec	ō	
TEXT		1317 Trip 1 pole Z			_	NOM	ES	0	
TEXT		1321 Op. mode Z3		3.7	-		RevNon/Ina	cti	
NUM		1322 R(Z3) ph-ph	13,53				600 ohms		
NUM	1111111111	1323 X(Z3)	12,54		_		600 ohms		
NUM	1.000	1324 RE(Z3) ph-E	16.91	9	_		600 ohms	-	
BOTH		1325 T3 DELAY	0,8		_	0-30		-	
NUM		1324 RE(Z3) ph-E	16.91	g	_		600 ohms		
BOTH	0.000	1325 T3 DELAY	0,8	22	-	0-30			
TEXT	10000	1331 Op. mode Z4			-		Rev/Non/Ina	cti	
INCM	130	1332 R(Z4) ph-ph	7,582		-		600 ohms		
NUM	4.9.4	1333 X(Z4)	5,556				600 ohms		

	one of ition	e Remarks Protection Data Olakha LINE 15		Cha	ange LZOP	1	Device Relay 1 Device	lag:		80 78 78A6	∏ 11_0la-Se	Archived m(New)	
Name LZOP F	Ranik	66kV Olakha-S	emtokha feeder	Vi	ew LZOP		Active	Group				<< Advance	bd
iroup Dis	played			۲	Rename		Сору		Delete	(Compare	Move	
Relay St	yte	7SA6xx_V4.3_1	IA 💌	Sele	ct Style		lanufactur		SIEMEN	8			
Schen	ne	MAIN					elay Mode ategory:		7SA6 Digital P:	ackag	9		
telav info	Elerr	ents Common	Taps Miscella	neous M	lemos				-	-			
	Tap Na			Find		ſap	s by Funct	ion				¥	
Type	Numb	e Tap Name		Setting		IR	emarks [F	Range		T	ap Descrip		-
BOTH		1241 R load (p	h-E)	71,802) ohms				_
NUM	107	1242 phi load	(ph-E)	45		Т	2	20-60 (deg.				
BOTH	108	1243 R load (p	h-ph)	71,802			(0.1-600) ohms				
NUM	109	1244 phi load	(ph-ph)	45			2	20-60 (leg.				
TEXT	110	1301 Op. mod	e Z1	Forward			F	wd/Re	wNon/In	nacti			
NUM	111	1302 R(Z1) ph	-ph	25,949		Т	(0.05-60	0 ohms				
NUM	112	1303 X(Z1)		2,676			(0.05-60	0 ohms				
NUM	113	1304 RE(Z1) p	h-E	24,468			(0.05-60	0 ohms				
BOTH		1305 T1-1phas	se	0		T	(0-30 se	ec .				
BOTH	115	1306 T1-multi-	phase	0			(0-30 se	ec				
NUM	116	1307 Zone Rei	duction	0			(0-45 de	eg.				
TEXT	117	1311 Op. mod	e Z2	Forward		Т	F	wd/Re	wNon/In	nacti			
NUM	118	1312 R(Z2) ph	-ph	26,015			(0.05-60	0 ohms				
NUM	119	1313 X(Z2)		4,225		Т	(0.05-60	0 ohms				
NUM	120	1314 RE(Z2) p	h-E	24,468			(0.05-60	0 ohms				
BOTH	121	1315 T2-1phas		0,4		Т	(0-30 se	ec				
BOTH	122	1316 T2-multi-		0,4		T	(0-30 se	ec				
TEXT		1317 Trip 1 po		NO		Τ	1	NOME	8				
TEXT	124	1321 Op. mod	e Z3	Forward			F	wd/Re	wNon/In	nacti			
NUM		1322 R(Z3) ph		26,055			0	0.05-60	0 ohms				
NUM	126	1323 X(Z3)		6,549			(0.05-60	0 ohms				
NUM	127	1324 RE(Z3) p	h-E	24,461			(0.05-60	0 ohms				
BOTH		1325 T3 DELA		0,8			(0-30 se	ec.				
TEXT	129	1331 Op. mod	e Z4	Inactive		T	1	Fwd/R	ewNon/Ir	nacti			-
NUM		1332 R(Z4) ph		10,61		$^{+}$			00 ohms				
NUM		1333 X(Z4)		11,155		$^{+}$			00 ohms				
111111	132	1334 RE(Z4) p		20,61		-			00 ohms				-

		Remarks otection Data			Device Tag Relay Tag:		77 CArchived		
Substat LZOP ID		Semtokha LINE 15	Cha	nge LZOP	Device Nar	ne R	EL511_Sem-O	a(New)	
Name		66kV Semtokha-Olakh	a feeder		Active Gro	ap [
LZOP R	ank	2	Vie	W LZOP				< Advance	
roup Disp	layed			Rename	Copy	Delete	Compare	Move	
Relay Sty	4e [REL511_V2.3_1A	Select	t Style	Manufacturer:	ABB			
Schem	;	MAIN	•		Relay Model: Category:	REL511_V Digital Pac			
elay info	Eleme	nts Common Taps	Miscellaneous M	lemos					
Find T	ap Nam	ю	Find	Filter 1	aps by Function			•	
Type N	lumbe	Tap Name	Setting		Remarks Rand	20	Tap Descript	on	
TEXT 1	07	Operation_Z1	Forward			Operation			
TEXT 1		OperationPP_Z1	ON		OFF	ON			
NUM 1		X1PP_Z1	0,35			00 ohms			
NUM 1		R1PP_Z1	0,1			00 ohms			
NUM 1		RFPP_Z1	3,99			00 ohms			
TEXT 1		Timer t1PP	ON		OFF				
NUM 1		t1PP	0		0-60				
TEXT 1		OperationPE_Z1	ON		OFF/				
NUM 1		X1PE_Z1	0,35			00 ohms	-		
NUM 1		R1PE_Z1	0,1			00 ohms	-		
NUM 1		X0PE_Z1	1,89583			200 ohms	-		
NUM 1		ROPE_Z1	0,45			200 ohms			
NUM 1		RFPE_Z1	3,18			00 ohms			
TEXT 1		Timer t1PE	ON		OFF/				
TEXT 1		tiPE	0		0-60				
		Operation_Z2	Forward ON			Operation			
TEXT 1 NUM 1		OperationPP_Z2	1,21		OFF/	00 ohms			
NUM 1	_	X1PP_Z2 R1PP_Z2	0,54			00 ohms			
NUM 1		RFPP_Z2	3,77			00 ohms			
TEXT 1		Timer t2PP	ON		OFF				
NUM 1	_	12PP	0,4		0-60	-			
TEXT 1		OperationPE_Z2	ON		OFF/				
	_								
NUM 1		X1PE_Z2	1,21			00 ohms			
NUM 1		R1PE_Z2	0,54			00 ohms			
NUM 1		XOPE_Z2	6,6			200 ohms	-		
NUM 1		ROPE_Z2	2,48			200 ohms			
NUM 1 TEXT 1		RFPE_Z2	3,01			00 ohms	-		
NUM 1		Timer t2PE t2PE	ON		OFF		-		
TEXT 1			0,4 Forward		0-60		-		
TEXT 1		Operation_Z3 OperationPP_Z3	OFF		OFF	Operation	1		
NUM 1		X1PP_Z3	0,45			00 ohms	1		
NUM 1		R1PP_Z3	0,45			00 ohms			
NUM 1		RFPP_Z3	2,69			00 ohms	1		
TEXT 1		Timer t3PP	0N		OFF				
NUM 1		13PP	0,8		0-60				
TEXT 1		OperationPE_Z3	0,0 ON		OFFI				
NUM 1		X1PE_Z3	1,37			00 ohms			
NUM 1		R1PE_Z3	0,23			00 ohms			
NUM 1		X0PE_Z3	7,43			200 ohms			
NUM 1		R0PE_Z3	1,07			200 ohms			
NUM 1		RFPE_Z3	3,6			00 ohms			
TEXT 1		Timer t3PE	ON		OFF				
NUM 1		I3PE	0,8		0-60				
		Operation_Z4	off			Operation			

Location	Contraction of		arks tion Data				- 1	Device Ta Relay Tag		72		Archived
Substa	tion	Semt	okha		Cha	nge LZOP		Device N	See 1		0532_DLingFdr	(New)
Name LZOP I	-		DLing feeder		Vie	W LZOP		Active Gr	oup	Г		Advanced
Group Dis	planet.				•	Rename	-	Сору	Delet	10	Compare	Move
or oup tors	prayed				2	rename	-	Copy			Compare	Move
Relay St	yle	PD53	2_1A	•	Selec	t Style		anufacturer		1000		
Schen	90	MAIN		-				elay Model: ategory:	PD53 Digita	Sec. 2	100P	
		COLUMN 1	Commun Trans Los	-		eren er			Cigito			
	arrestore.		Common Taps Misc	ellaneou	IS M	Contraction of the local division of the			_			
Find	Tap Na	me		Find		Filter	Taps	s by Function				-
Type	Numb	Tap	lame	Sett	ng		R	emarks Ra	nge		Tap Descriptio	n:
TEXT	737	DIST	Characteristic	Poly	gon			Cir	le/Polygo	n		
TEXT	738	DIST	Op. mode zone 4	Norr	nal			Zor	e 4 Op. h	fode		
NUM	739	DIST	X1 (polygon)	1,97				0.1	200 ohm	s		
NUM	740	DIST	X2 (polygon)	2,96	ê.			0.1	-200 ohm	IS .		
NUM	741	DIST	X3 (polygon)	4,44				0.1	200 ohm	\$		
NUM	742	DIST	X4 (polygon)	0,1				0.1	200 ohm	s		
NUM	743	DIST	R1,PG (polygon)	3,84				0.1	-200 ohm	s		
NUM	744	DIST	R1,PP (polygon)	3,85				0.1	-200 ohm	s		
NUM	745	DIST	R2,PG (polygon)	3,84				0.1	200 ohm	IS .		
NUM	746	DIST	R2,PP (polygon)	3,84				0.1	200 ohm	15		
NUM	747	DIST	R3,PG (polygon)	3,84				0.1	200 ohm	\$		
NUM	748	DIST	R3,PP (polygon)	3,82				0.1	200 ohm	s		
NUM	749		R4,PG (polygon)	3,79				0.1	200 ohm	s		
NUM	750	DIST	R4,PP (polygon)	3,79	É			0.1	200 ohm	IS		
TEXT		DIST	Direction N1	Forw	ard d	irectional	1	Zon	e Dir.			
TEXT		DIST	Direction N2	Forw	ard d	irectional		Zon	e Dir.			
TEXT	770	DIST	Direction N3	Forw	ard d	irectional		Zon	e Dir.			
TEXT	771	DIST	Direction N4	Back	ward	directional		Zon	e Dir.			
TEXT		DIST	Direction N5	Forw	and d	irectional		Zon	e Dir.			
NUM		DIST	Oper.val.Vmemory	0,01				0.01	-1			
BOTH	774	DIST	t1	0				0-10).0 s			
BOTH	775	DIST	12	0,4				0-10	1.0 s			
BOTH	776	DIST	t3	0,8				0-10).0 s			
BOTH	777	DIST	14	1				0-10	.0 s			

		e Remarks Protection Data			Device Ta Relay Tag		76 CArchived 74		
Subst		Lobeysa	Cha	inge LZOP	Device Na	me R	EL511_Lob-Se	m(New)	
LZOP	D	LINE 17 66kV Lobeysa-Semi	tokha feeder		Active Gro				
LZOP	Dank	2		ew LZOP	ACTIVE OF	ч р			
		1.						< Advanced	
roup Dis	splayed		•	Rename	Сору	Delete	Compare	Move	
Relay S	tyle	REL511_V2.3_1A	• Sele	t Style	Manufacturer: Relay Model:	ABB REL511_V	2		
Scher	ne	MAIN	*		Category:	Digital Pac			
elav info	Elem	ents Common Tap	Miscellaneous N	lemos					
	Tap Na		Find		aps by Function				
Type	Numb	Tap Name	Setting		Remarks Ran	ge	Tap Descript	ion	
TEXT		Operation_Z1	Forward			e Operation			
TEXT	108	OperationPP_Z1	ON		OFF	/ON			
NUM		X1PP_Z1	4,34		0.1-	400 ohms			
NUM		R1PP_Z1	2,26			400 ohms			
NUM		RFPP_Z1	8,56			400 ohms			
TEXT		Timer t1PP	ON			/0N			
TEXT		t1PP OperationPE_Z1	0 ON) sec /ON	-		
NUM		X1PE_Z1	4,34			400 ohms			
NUM		R1PE_Z1	2,26			400 ohms			
NUM		X0PE_Z1	23,6			1200 ohms			
NUM		R0PE_Z1	10,3			1200 ohms			
NUM		RFPE_Z1	19,52			400 ohms			
TEXT	120	Timer t1PE	ON		OFF	/ON			
NUM	121	t1PE	0		0-60) sec			
TEXT	122	Operation_Z2	Forward		Zon	e Operation			
TEXT	123	OperationPP_Z2	ON		OFF	/0N			
NUM		X1PP_Z2	6,51			400 ohms			
NUM		R1PP_Z2	3,38			400 ohms			
NUM		RFPP_Z2	12,83			400 ohms			
TEXT		Timer t2PP	ON			/0N			
NUM		t2PP	0,4) sec			
TEXT		OperationPE_Z2	ON		OFF				
NUM		X1PE_Z2	6,51			100 ohms			
NUM		R1PE_Z2	3,38			400 ohms			
NUM NUM		X0PE_Z2 R0PE_Z2	35,4			1200 ohms 1200 ohms	-		
NUM		REPE_Z2	15,44			400 ohms			
TEXT		Timer t2PE	25 ON		OFF		-		
NUM		t2PE	0,4			sec			
TEXT		Operation_Z3	Forward			e Operation			
TEXT		OperationPP_Z3	ON		OFF				
NUM		X1PP_Z3	10,84			400 ohms			
NUM		R1PP_Z3	5,64			00 ohms			
NUM	141	RFPP_Z3	21,39			400 ohms			
TEXT		Timer t3PP	ON		OFF	ION			
NUM		t3PP	0,8			sec			
TEXT		OperationPE_Z3	ON		OFF				
NUM		X1PE_Z3	10,84			100 ohms			
NUM		R1PE_Z3	5,64			400 ohms			
NUM		X0PE_Z3	58,99			200 ohms			
NUM		ROPE_Z3	25,74			1200 ohms			
NUM		RFPE_Z3	25			400 ohms			
TEXT		Timer t3PE t3PE	ON 0,8		OFF				
NUM					0-60				

		Remarks otection Data			Devic Relay	e Tag: Tag:	75 73		Archived
Substati		Lobeysa	Cha	nge LZOP	Devic	e Name	R	EL511_Lob-Ru	ri(New)
LZOP ID Name		LINE 19 56kV Lobeysa-Basochu	feeder		Activ	e Group			
LZOP R		2		w LZOP		o or oup			
		-							< Advanced
iroup Displ	layed		۲	Rename	Сору		Delete	Compare	Move
Relay Styl	te [REL511_V2.3_1A	• Selec	t Style	Manufacte Relay Mod		88 EL511_V	2	
Scheme	• •	MAIN	-		Category:	C	igital Paci	cage	
elay info	Eleme	nts Common Taps	liscellaneous M	emos					
Find Ta	ap Nam	e	Find	🔲 Filter 1	aps by Fund	tion			*
		Tap Name	Setting		Remarks	Range		Tap Descript	ion
TEXT 1		Operation_Z1	Forward			Zone Op			
TEXT 1		OperationPP_Z1	ON			OFF/ON		_	
NUM 1		K1PP_Z1	3,52		_	0.1-400			
NUM 1		R1PP_Z1	1,83			0.1-400			
NUM 1		RFPP_Z1	11,5		-	0.1-400			
TEXT 1		Timer t1PP	ON			OFF/ON			
NUM 1		1PP	0			0-60 se			
TEXT 1		OperationPE_Z1	ON			OFF/ON			
NUM 1		KIPE_ZI	3,52			0.1-400			
NUM 1		R1PE_Z1	1,83			0.1-400			
NUM 1		KOPE_Z1	19,18			0.1-120			
NUM 1		ROPE_Z1	8,36			0.1-120			
TEXT 1		RFPE_Z1 Timer t1PE	12,5 ON			0.1-400 OFF/ON			
NUM 1		11PE	0		-	0-60 se			
TEXT 1		Operation_Z2	Forward			Zone Op			
TEXT 1		OperationPP_Z2	ON			OFF/ON			
NUM 1		KIPP_Z2	7,47			0.1-400			
NUM 1		R1PP_Z2	5,04			0.1-400			
NUM 1		RFPP_Z2	14			0.1-400			
TEXT 1		Timer t2PP	ON			OFF/ON			
NUM 1		2PP	0,4			0-60 se			
TEXT 1		OperationPE_Z2	ON			OFF/ON			
NUM 1	30 3	(1PE_Z2	7,46			0.1-400	ohms		
NUM 1		R1PE_Z2	5,04			0.1-400			
NUM 1		KOPE_Z2	40,6			0.1-120			
NUM 1		ROPE_Z2	23			0.1-120			
NUM 1		RFPE_Z2	15			0.1-400	ohms		
TEXT 1		Timer t2PE	ÓN			OFFION			
NUM 1		2PE	0,4			0-60 se	1		
TEXT 1	37 (Operation_Z3	Forward			Zone Og	eration		
TEXT 1		OperationPP_Z3	OFF			OFFION			
NUM 1	39 3	(1PP_Z3	10,96			0.1-400	ohms		
NUM 1		R1PP_Z3	6,15			0.1-400			
NUM 1		RFPP_Z3	5			0.1-400			
TEXT 1		Timer t3PP	ON			OFFION			
NUM 1		3PP	0,8			0-60 se			
TEXT 1		OperationPE_Z3	ON			OFFION			
NUM 1		KIPE_Z3	10,96			0.1-400			
NUM 1		R1PE_Z3	6,15			0.1-400			
NUM 1		KOPE_Z3	63,93			0.1-120			
NUM 1		ROPE_Z3	27,88			0.1-120			
NUM 1		RFPE_Z3	15			0.1-400			
TEXT 1		Timer t3PE 3PE	0N 0,8			OFF/ON 0-60 se			
		107 E	101.05						

		Remarks Protection Data			Devic Relay	e Tag: Tag:	74		Archived
Subst	ation	Rurichu	Chi	ange LZOP		e Name		EL511_Baso F	dr(New)
LZOP	D	LINE 22	–				i i	_	
Name		66kV Rurichu-Baso			Activ	e Group			
LZOP	Rank	2		ew LZOP				_	< Advanced
roup Dis	played		۲	Rename	Сору		Delete	Compare	Move
Relay S	tyle	REL511_V2.3_1A	▼ Sele	ct Style	Manufactu Relay Mod		98 EL511_V	2	
Scher	ne	MAIN			Category:		igital Pac		
elay info				Mernos					
_	Tap Nar		Find	Filter I	Taps by Fund			In	<u> </u>
		Tap Name	Setting		Remarks		oration	Tap Descript	ion
TEXT		Operation_Z1	Forward			Zone Op	eration		
TEXT		OperationPP_Z1	0N			OFF/ON	ohme		
NUM		X1PP_Z1	0,67		-	0.1-400			
NUM		R1PP_Z1	0,37			0.1-400			
TEXT		RFPP_Z1 Timer t1PP	6,01 ON			0.1-400 OFF/ON			
NUM		tIPP	0		-	0-60 se			
TEXT		OperationPE_Z1	ON			OFF/ON			
NUM		X1PE_Z1	0,67			0.1-400		-	
NUM		R1PE_Z1	0,37			0.1-400			
NUM		X0PE_Z1	3,87			0.1-120			
NUM		R0PE_Z1	1,69			0.1-120			
NUM		RFPE_Z1	5,33			0.1-400			
TEXT		Timer t1PE	ON			OFF/ON			
NUM		t1PE	0			0-60 se			
TEXT		Operation_Z2	Forward			Zone Op			
TEXT		OperationPP_Z2	ON			OFF/ON			
NUM		X1PP_Z2	2,22			0.1-400	ohms		
NUM		R1PP_Z2	0,47			0.1-400	ohms		
NUM	126	RFPP_Z2	7,06			0.1-400	ohms		
TEXT	127	Timer t2PP	ON			OFF/ON			
NUM	128	t2PP	0,4			0-60 se	2		
TEXT	129	OperationPE_Z2	ON			OFF/ON			
NUM	130	X1PE_Z2	2,22			0.1-400	ohms		
NUM	131	R1PE_Z2	0,47			0.1-400	ohms		
NUM	132	X0PE_Z2	1,332			0.1-120	0 ohms		
NUM	133	R0PE_Z2	2,11			0.1-120	0 ohms		
NUM	134	RFPE_Z2	7,82			0.1-400	ohms		
TEXT	135	Timer t2PE	ON			OFF/ON			
NUM	136	t2PE	0,4			0-60 se	c		
TEXT	137	Operation_Z3	Reverse			Zone O	peration		
TEXT	138	OperationPP_Z3	ON			OFF/ON			
NUM	139	X1PP_Z3	0,94			0.1-400	ohms		
NUM	140	R1PP_Z3	2,46			0.1-400	ohms		
NUM	1	RFPP_Z3	4,15			0.1-400			
TEXT	1	Timer t3PP	ON		_	OFF/ON			
NUM		t3PP	1,5		_	0-60 se			
TEXT	1	OperationPE_Z3	ON		_	OFF/ON			
NUM	1	X1PE_Z3	0,99			0.1-400			
NUM	1	R1PE_Z3	2,46		_	0.1-400			
NUM		X0PE_Z3	5,40498			0.1-120			
NUM		R0PE_Z3	11,15		_	0.1-120			
NUM		RFPE_Z3	5,65		_	0.1-400			
TEXT		Timer t3PE	ON			OFF/ON			
NUM		t3PE	1,5		_	0-60 se			
	152	Operation_Z4	Off			Zone Or	eration		

cation	Device	Remark	ks			Devic	e Tag:	73	· · · · ·	Archived	
Local Z	one of F	Protectio	n Data			Relay	Tag:	71			
Subst		Basoch	-	Ch	ange LZOP	Devic	e Name	R	EL511_Rurich	u Fdr.	
LZOP I Name	D	LINE 22	: Isochu-Rurichu f			Activ	Group	i i			
LZOP	Pank	2			iew LZOP	- ACON	oroup				
LEOF		J*								<< Advanced	
roup Dis	splayed				Rename	Сору		Delete	Compare	Move	
Relay S	tyle	REL511	_V2.3_1A	 Sele 	ct Style	Manufactu		188 188			
Scher	me	MAIN		*		Relay Mod Category:		REL511_V Digital Pac			
elay Info	Elem	ents Co	mmon Taps M	iscellaneous I	lemos						
Find	Tap Nai	ne		Find	🗖 Filter 1	aps by Func	tion			•	
Type	Numbe	Tap Na	me	Setting		Remarks	Range		Tap Descrip	tion	
TEXT	107	Operati	on_Z1	Forward			Zone O	peration			
TEXT	108	Operation	onPP_Z1	ON			OFF/ON	4			
NUM	109	X1PP_Z	1	0,67			0.1-400	ohms			
NUM	110	R1PP_2	Z1	0,37			0.1-400	ohms			
NUM	111	RFPP_2	Z1	12,81			0.1-400				
TEXT		Timer t1	PP	ON			OFF/ON				
NUM		t1PP		0			0-60 se				
TEXT	1	Operation	onPE_Z1	ON			OFF/ON	4			
NUM	1	X1PE_2	:1	0,67			0.1-400				
NUM		R1PE_2		0,37			0.1-400				
NUM	1	X0PE_Z		3,87				10 ohms			
NUM		R0PE_2		1,69				10 ohms			
NUM		RFPE_2		12,74			0.1-400				
TEXT		Timer t1	PE	ON			OFF/ON				
NUM		t1PE		0			0-60 se				
TEXT		Operati	-	Forward				peration			
TEXT			onPP_Z2	ON			OFF/ON				
NUM		X1PP_Z		4,65			0.1-400				
NUM		R1PP_2		0,47			0.1-400				
NUM		RFPP_2		14,13			0.1-400 OFF/ON				
TEXT NUM		Timer t2 t2PP	(FF	ON			0-60 se				
TEXT			orDE 73	0,4 ON			OFF/ON				
7 7	•		onPE_Z2			-		-	-		
NUM		X1PE_Z		4,65			0.1-400				
NUM		R1PE_Z		0,47			0.1-400				
NUM		X0PE_Z		2,79			0.1-120				
NUM		ROPE_Z		2,11			0.1-120				
NUM TEXT		RFPE_2 Timer t2		14,18 ON			0.1-400 OFF/ON				
NUM		t2PE	re -	0.4			0-60 se				
TEXT		Operatio	n 73	0,4 Reverse				c peration			
TEXT			m_23 mPP_Z3	ON			OFF/ON				
NUM		X1PP_Z		2,44			0.1-400				
NUM		R1PP_Z		2,44			0.1-400				
NUM		RFPP_Z		4,17			0.1-400				
TEXT		Timer t3		ON			OFF/ON				
NUM		t3PP		1,5			0-60 se				
TEXT			nPE_Z3	ON			OFF/ON				
NUM		X1PE_Z	-	2,39			0.1-400				
NUM		R1PE_Z		2,46			0.1-400				
NUM		X0PE_Z		13,0484				0 ohms			
NUM		ROPE_Z		11,15				0 ohms			
NUM		RFPE_Z		5,16			0.1-400				
TEXT		Timer t3		ON			OFF/ON				
NUM		t3PE		1,5			0-60 se				
_	152		n_Z4	Off				- oeration			

132kV feeders:

		e Remarks				-	ce Tag:		61 59		□ Ar	chived	
Subst		Protection Data Nangkhor					y Tag:						_
LZOP		LINE 78		Ch	ange LZOP	Devie	ce Name		JEPAG	>_Nang-	KHP(N	lew 2)	
Name		132kV Nangkhor-Kuri	chu feeder			Activ	e Group						-
LZOP	Rank	3		V	iew LZOP								-
LEOF	r van in v	1.									<<	Advance	d
iroup Di	splayed			•	Rename	Сору	,	Delete		Compar	e	Move	
Relay S	tyle	EP311111BCDHF	•	Sele	ect Style	Manufact		ALSTOM					
Sche	me	MAIN	•			Relay Mo Category:		EPAC 31 Digital P					
elay Info	Elem	ents Common Taps	Miscellane	eous	Memos								
Find	Tap Na	me	F	Find	Filter T	aps by Fun	ction					•	
Type	Numb	e Tap Name	s	Setting		Remarks	Range		Т	ap Desc	ription	1	
NUM	3	1003 LIN Line Length	3	36			0.3-999	9.99 km	L	ine leng	th in kr	m	
NUM	4	1004 LIN Line Length	0),18			0.18-62	21.49 mi	les L	ine leng	th in m	iles	
NUM	5	1005 LIN Ku	1	200			1-20,00	00	V	T ratio			
NUM	6	1006 LIN Ki	3	300			1-20,00		C	T ratio			
TEXT		1007 LIN Known Cha		Cartesia	n						artesia	n, polar d	or other
NUM		1008 LIN Zd		0,001				999 ohm				e Z (sec.	
NUM	9	1009 LIN Phid	0				0-90 de					ice line a	
NUM	10							-					ngle ns) zone 1
_		100A LIN Z01		0,001				999 ohm					
NUM	11	100B LIN Phi01	0					90 deg				_	e (zone 1)
NUM	12	100C LIN Z02		0,001				999 ohm					ms) other zone
NUM	13	100D LIN Phi02	0)			-90 to +	90 deg	Z	ero sequ	Jeuce	line angle	e (other zones)
NUM	14	100E LIN Rd	1	,526			0.001-9	999 ohm	is P	'os. seq.	line R	(sec. oh	ms)
NUM	15	100F LIN Xd	3	3,868			0.001-9	999 ohm	is P	'os. seq.	line X	(sec. ohr	ns)
NUM	16	1010 LIN R01	3	3,804			0.001-9	999 ohm	is Z	ero seq.	line R	(sec. oh	ms) zone 1
NUM	17	1011 LIN X01	1	2,15			-999 - 9	999 ohm	is Z	ero seq.	line X	(sec. ohr	ns) zone 1
NUM	18	1012 LIN R02	3	3,804			0.001-9	999 ohm	is Z	ero seq.	line R	(sec. oh	ms) other zone
NUM	19	1013 LIN X02		2,15			-999 - 9	999 ohm					ns) other zone
NUM	20	1014 LIN K01r	0				-7 - 7			te(K0) fo		-	
NUM	21	1015 LIN K01x	0				-7 - 7			n(K0) for			
NUM	22	1016 LIN K02r	0				-7 - 7			te(K0) fo			
NUM	23	1017 LIN K02x	0				-7 - 7			n(K0) for			
NUM	24	1101 ZON Z1		, 3,33			0.1-200	lahme		1 1		nce (sec.	ohme)
NUM							0.1-200						
-		1102 ZON Z1 Overrea),1									ince (sec. ohm
NUM		1103 ZON T1	0				0-10 sec			ne 1 time			
NUM		1104 ZON Z2	4,	74			0.1-200					e (sec. ol	hms)
NUM		1105 ZON T2	0,				0-10 sec	:	_	ne 2 time			
NUM	29	1106 ZON Z3	5,	31			0.1-200	ohms	Zo	ne 3 imp	edanc	e (sec. ol	hms)
NUM		1107 ZON T3	0,	8			0-10 sec		Zo	ne 3 time	e delay	1	
TEXT	31	1108 ZON Dir. Z3	Fo	orwards	:		Forward:	s/Backw	an				
NUM		1109 ZON Z4		16			0.1-200	ohms	Zo	ne 4 imp	edanc	e (sec. ol	hms)
NUM		110A ZON T4	0,				0-10 sec			ne 4 time			
NUM		110B ZON Z5		04			0.1-200					e (sec. ol	hms)
NUM		110C ZON T5	0,				0-10 sec			ne 5 time			
NUM		110D ZON T>>	0				0-10 sec			time de			
NUM		110E ZON T>	0				0-10 sec			ime dela			
NUM		110F ZON Ph/Gnd RZ		5,66			0-200 oh		_	-E loop F	/	ne 1	
NUM		1110 ZON Ph/Ph RZ1		-			0-200 oh		_	-Ph loop r			
				6,79 9.24					_				
NUM		1111 ZON RLim Z2 1112 ZON RLim Z3		9,21 0,07			0-200 oh 0-200 oh		_	op R for:			
NUM								17000	11.0	op R for :			

ocation	Devic	e Remar	ks				- 1	e Tag:	63		Archived
Local Z	one of	Protectio	n Data				Relay	/Tag:	61		
Subst	ation	Nangkf	or		C	hange LZOP	Devic	e Name	EP	AC Nang-N	NLam(New)
LZOP	ID .	LINE 74	1		_						
Name		Nangkł	ior-Nanglam fe	eder at N	angkno	r end	Activ	e Group			-
LZOP	Rank	2				View LZOP					
		· · · ·								_	<< Advanced
roup Dis	enlaved					Rename	Copy		Delete	Compare	Move
roup Dis	spiayeu	· .				Kename	Сору		Delete	compare	e move
Relay S	tyle	EP3111	11BCDHF	*	Se	lect Style	Manufactu	urer: A	LSTOM		
Scher	-	MAIN		-			Relay Mod Category:		PAC 3136/ igital Pack		
elay info		ents C	ommon Taps	Miscella	neous	Memos			-		
	Tap Na	_		miscena	Find		aps by Fund	ction			•
-											
		e Tap Na			Setting		Remarks			Tap Desci	
NUM	-		N Line Length		31			0.3-999		Line lengt	
NUM			N Line Length		0,18				1.49 miles		h in miles
NUM		1005 L			1200			1-20,00		VT ratio	
NUM	6	1006 L			300			1-20,00		CT ratio	
TEXT	1		N Known Char		Cartes	ian					rtesian, polar or other
NUM	8	1008 L			0,462				99 ohms	Positive se	eq. line Z (sec. ohms)
NUM	9	1009 L	N Phid		68			0-90 de	g	Positive se	equence line angle
NUM	10	100A L	N Z01		12,731			0.001-9	99 ohms	Zero seq.	line Z (sec. ohms) zone 1
NUM	11	100B L	IN Phi01		72			-90 to +9	90 deg	Zero sequ	ence line angle (zone 1)
NUM	12	100C L	IN Z02		0,001			0.001-9	99 ohms	Zero seq.	line Z (sec. ohms) other zone:
NUM	13	100D L	IN Phi02		0			-90 to +9	30 deg	Zero sequ	ence line angle (other zones)
NUM	14	100E L	IN Rd		1,313			0.001-9	99 ohms	Pos. seq.	line R (sec. ohms)
NUM	15	100F L	N Xd		3,309			0.001-9	99 ohms	Pos. seq.	line X (sec. ohms)
NUM	16	1010 L	N R01		3,276			0.001-9	99 ohms	Zero seq.	line R (sec. ohms) zone 1
NUM	17	1011 L	N X01		10,46			-999 - 9	99 ohms	Zero seq.	line X (sec. ohms) zone 1
NUM	18	1012 L	N R02		3,276			0.001-9	99 ohms	Zero seq.	line R (sec. ohms) other zone
NUM	19	1013 L	N X02		10,46			-999 - 9	99 ohms	Zero seq.	line X (sec. ohms) other zone:
NUM	20	1014 L	N K01r		0			-7 - 7		Re(K0) for	zone 1
NUM	21	1015 LI	N K01x		0			-7 - 7		Im(K0) for	zone 1
NUM	22	1016 L	N K02r		0			-7 - 7		Re(K0) for	otherzones
NUM	23	1017 L	N K02x		0			-7 - 7		Im(K0) for	other zones
NUM	24	1101 Z	ON Z1		2,86			0.1-200	ohms	Zone 1 im	pedance (sec. ohms)
NUM	25	1102 Z	ON Z1 Overread	:h	0,1			0.1-200	ohms	Extended.	Zone 1 impedance (sec. ohm
NUM	26	1103 Z	ON T1		0			0-10 sec		Zone 1 tim	e delav
NUM	27	1104 Z			4,47			0.1-200			pedance (sec. ohms)
NUM	28	1105 Z			0,4			0-10 sec		Zone 2 tim	
NUM	29	1106 Z			14,67			0.1-200			pedance (sec. ohms)
NUM	30	1107 Z			0,8			0-10 sec		Zone 3 tim	1
			DN Dir. Z3		Forwar	ds			s/Backwari		
NUM		1109 Z			5,38			0.1-200			pedance (sec. ohms)
NUM		110A Z			0,8			0-10 sec		Zone 4 tim	
NUM		110B Z			0,89			0.1-200			pedance (sec. ohms)
_		110C Z			0,8			0-10 sec		Zone 5 tim	
	36		ON T>>		0			0-10 sec		I>> time de	
NUM		110E Z			0			0-10 sec		I> time del	
NUM			ON Ph/Gnd RZ1		20			0-200 of			R for zone 1
NUM			ON Ph/Ph RZ1		30			0-200 of			p R for zone 1
NUM			ON RLim Z2		30			0-200 of		Loop R for	
NUM	1		ON RLim Z2		30			0-200 of		Loop R for	
THOM	42		ON RLim 23		34			0-200 of			zones 4 and 5 (starter)

.ocation	Devic	e Remarks				Devic	e Tag:	55		Archived
Local Z	one of l	Protection Data				Relay	/Tag:	53		
Substa	ation	Nganlam		C	hange LZOP	Desic	e Name	FP	AC Nangla	am-Nangkho
LZOP	ID	LINE 74		-			e mante		re_rearigit	
Name		132kV Nganlam-Nag	khor feede	er 👘		Activ	e Group			*
LZOP	Rank	2			View LZOP					
									_	<< Advanced
roup Dis	nlawed				Rename	Copy		Delete	Compare	Move
roup bio	payea	1		-	rendine	copy		belete	compare	
Relay S	tyle	EP311111BCDHF	•	Se	lect Style	Manufactu Relay Mod		LSTOM PAC 3136/	2626	
Scher	ne	Demo				Category:		igital Pack		
elay Info	Elem	ents Common Taps	Miscella	neous	Memos					
Find	Tap Na	me	_	Find	📄 🗖 Filter T	aps by Fund	ction			•
Type	Numbe	Tap Name		Setting	1	Remarks	Range		Tap Desci	ription
NUM		1003 LIN Line Lengt	1	31			0.3-999	99 km	Line lengt	
NUM	4	1004 LIN Line Lengt		0,18					Line lengt	
_	5	1005 LIN Ku	-	1200			1-20,00		VT ratio	
NUM	6	1006 LIN Ki		300			1-20,00		CT ratio	
TEXT	7	1007 LIN Known Cha	ar.	Cartes	ian					rtesian, polar or other
NUM	8	1008 LIN Zd		0,462	1011			99 ohms		eq. line Z (sec. ohms)
NUM	9	1009 LIN Phid		68			0-90 de			equence line angle
NUM	10	100A LIN Z01		12,731				9 99 ohms		line Z (sec. ohms) zone 1
NUM	11	1008 LIN Phi01		72			-90 to +9			ence line angle (zone 1)
NUM	12	100C LIN Z02		0,001				99 ohms		line Z (sec. ohms) other zone
NUM	13	100D LIN Phi02		0,001			-90 to +9			ence line angle (other zones)
NUM	14	100E LIN Rd		1,313				99 ohms		line R (sec. ohms)
NUM	15	100F LIN Xd						99 ohms		line X (sec. ohms)
NUM	16	1010 LIN R01		3,309				99 ohms 99 ohms		
NUM	17			3,276						line R (sec. ohms) zone 1
NUM	18	1011 LIN X01		10,46				99 ohms		line X (sec. ohms) zone 1
NUM		1012 LIN R02		3,309				99 ohms		line R (sec. ohms) other zone
_	19	1013 LIN X02		10,46				99 ohms		line X (sec. ohms) other zone
NUM	20	1014 LIN K01r		0			-7 - 7		Re(K0) for	
NUM	21	1015 LIN K01x		0			-7 - 7		Im(K0) for	
NUM	22	1016 LIN K02r		0			-7 - 7			otherzones
NUM	23	1017 LIN K02x		0			-7 - 7			otherzones
NUM		1101 ZON Z1		3,14			0.1-200			pedance (sec. ohms)
NUM		1102 ZON Z1 Overre	ach	0,1		1	0.1-200		1	Zone 1 impedance (sec. ohm
NUM		1103 ZON T1		0			0-10 se		Zone 1 tin	
NUM		1104 ZON Z2		5,18			0.1-200			pedance (sec. ohms)
NUM		1105 ZON T2		0,4			0-10 se		Zone 2 tin	
NUM		1106 ZON Z3		0,89			0.1-200	ohms	Zone 3 im	pedance (sec. ohms)
NUM		1107 ZON T3		1			0-10 se	¢	Zone 3 tin	ne delay
TEXT	1	1108 ZON Dir. Z3		Backw	ards		Forward	is/Backwar	1	
NUM		1109 ZON Z4		6,6			0.1-200			pedance (sec. ohms)
NUM		110A ZON T4		0,8			0-10 se	c	Zone 4 tin	
NUM		110B ZON Z5		0,1			0.1-200	ohms		pedance (sec. ohms)
NUM	35	110C ZON T5		0			0-10 se	c	Zone 5 tin	ne delay
NUM	36	110D ZON T>>		0			0-10 se	¢	l>> time d	elay
NUM	37	110E ZON T>		0			0-10 se	c	I> time de	lay
NUM	38	110F ZON Ph/Gnd R	Z1	7,48			0-200 o	hms	Ph-E loop	R for zone 1
NUM	39	1110 ZON Ph/Ph RZ	1	7,54			0-200 o	hms	Ph-Ph loo	p R for zone 1
NUM	40	1111 ZON RLim Z2		9,22			0-200 o	hms	Loop R for	
NUM		1112 ZON RLim Z3		5			0-200 o		Loop R fo	
	42	1113 ZON RLim Star	ter	9,65			0-200 o			r zones 4 and 5 (starter)

ocation.	-						1	e Tag:	57		C Archived
Local Z	one of	Protec	tion Data				Relay	/Tag:	55		
Substa	ation	Ngar	lam		C	hange LZOP	Devic	e Name	EP	AC_Nlam-	Tinti(New)
LZOPI	ID	LINE			_						
Name			V Nanglam-Ting	tibi feeder			Activ	e Group			*
LZOP	Rank	2				View LZOP					<< Advanced
roup Dis	played					Rename	Copy		Delete	Compare	e Move
Relay S	tyle	EP31	1111BCDHF	-	Se	lect Style	Manufacto Relay Mod		LSTOM PAC 3136/	2526	
Scher	ne	MAIN		*			Category:		igital Pack		
elay Info	Elen	nents	Common Taps	Miscella	neous	Memos					
Find	Tap Na	me			Find	🗖 Filter T	aps by Fun	ction			•
Type	Numb	e Tap I	Name		Setting	1	Remarks	Range		Tap Desc	ription
NUM	and the owner of the		LIN Line Length		84,5			0.3-999	99 km	Line lengt	
NUM	4		LIN Line Length		0,18			0.18-62	1.49 miles	Line lengt	th in miles
NUM	5	1005	LIN Ku		1200			1-20,00	0	VT ratio	
NUM	6	1006	LIN KI		300			1-20,00	0	CT ratio	
TEXT	7	1007	LIN Known Char		Cartes	ian		Cartesia	n/Polar/X0	Line Z - ca	artesian, polar or other
NUM	8	1008	LIN Zd		0,001			0.001-9	99 ohms	Positive s	eq. line Z (sec. ohms)
NUM	9	1009	LIN Phid		0			0-90 de	g	Positive s	equence line angle
NUM	10	100A	LIN Z01		0,001			0.001-9	99 ohms	Zero seq.	line Z (sec. ohms) zone 1
NUM	11	100B	LIN Phi01		0			-90 to +9	30 deg	Zero sequ	Jence line angle (zone 1)
NUM	12	1000	LIN Z02		0,001			0.001-9	99 ohms	Zero seq.	line Z (sec. ohms) other zone
NUM	13	100D	LIN Phi02		0			-90 to +9	30 deg	Zero sequ	uence line angle (other zones)
NUM	14	100E	LIN Rd		1,885			0.001-9	99 ohms	Pos. seq.	line R (sec. ohms)
NUM	15	100F	LIN Xd		4,781			0.001-9	99 ohms	Pos. seq.	line X (sec. ohms)
NUM	16	1010	LIN R01		4,702			0.001-9	99 ohms	Zero seq.	line R (sec. ohms) zone 1
NUM		1011	LIN X01		15,02			-999 - 9	99 ohms		line X (sec. ohms) zone 1
NUM	18	1012	LIN R02		4,702			0.001-9	99 ohms	Zero seq.	line R (sec. ohms) other zone
NUM	19	1013	LIN X02		15,02			-999 - 9	99 ohms	Zero seq.	line X (sec. ohms) other zone
NUM	20	1014	LIN K01r		0			-7 - 7		Re(K0) for	r zone 1
NUM		1015	LIN K01x		0			-7 - 7		Im(K0) for	rzone 1
NUM	22	1016	LIN K02r		0			-7 - 7		Re(K0) for	r other zones
NUM	23	1017	LIN K02x		0			-7 - 7		Im(K0) for	r other zones
NUM			ZON Z1		7,81			0.1-200			ipedance (sec. ohms)
NUM	25	1102	ZON Z1 Overrea	ch	0,1			0.1-200	ohms	Extended	Zone 1 impedance (sec. ohm
NUM		1103	ZON T1		0			0-10 se	c	Zone 1 tin	me delay
NUM	27	1104	ZON Z2		12,2			0.1-200	ohms	Zone 2 im	npedance (sec. ohms)
NUM	28	1105	ZON T2		0,4			0-10 se	¢	Zone 2 tin	me delay
NUM	29	1108	ZON Z3		2,44			0.1-200	ohms		npedance (sec. ohms)
NUM		1107	ZON T3		1			0-10 se	Ċ	Zone 3 tin	ne delay
TEXT		1108	ZON Dir. Z3		Backw	ards		Forward	is/Backwa	n	
NUM	1	1109	ZON Z4		17,09			0.1-200		Zone 4 im	npedance (sec. ohms)
NUM	1		ZON T4		0,8			0-10 se		Zone 4 tin	,
NUM			3 ZON Z5		0,1			0.1-200			npedance (sec. ohms)
NUM			ZON T5		0			0-10 se		Zone 5 tin	
NUM		1100	DZON T>>		0			0-10 se	¢	I>> time d	jelay
NUM			ZON T>		0			0-10 se		I> time de	
NUM		110F	ZON Ph/Gnd RZ	1	4,62			0-200 o			R for zone 1
NUM		_	ZON Ph/Ph RZ1		4,62			0-200 o			op R for zone 1
NUM	1	_	ZON RLim Z2		5			0-200 0	hms	Loop R fo	
NUM	1	1112	ZON RLim Z3		5			0-200 o	hms	Loop R fo	
NUM	42	1113	ZON RLim Start	er	5			0-200 o	hms	Loop R fo	r zones 4 and 5 (starter)

	_	e Remark					- 1	e Tag:	58		☐ Archived		
Local Zo	one of l	Protection	i Data				Relay	/Tag:	56				
Substa		Tintibi			C	hange LZOP	Devic	e Name	EP.	AC_Ting-N	Lam(New)		
LZOPI	D	LINE 73											
Name			ingtibi-Nangla	im feeder			Activ	e Group			<u> </u>		
LZOP F	Rank	2				View LZOP					<< Advanced		
							-				< Advanced		
òroup Dis	played					Rename	Сору		Delete	Compare	e Move		
Delevier	I	-					Manufact	urer: A	LSTOM				
Relay St		MAIN	1BCDHF	•	Se	lect Style	Relay Mod	iel: E	PAC 3136/				
Schen		·	- 1				Category:	D	igital Pack	age			
telay info		_	mmon Taps	Miscella		Memos							
	fap Na	, ,			Find	Filter T	aps by Fun				X		
		a Tap Nam			Setting		Remarks			Tap Desc			
NUM			I Line Length		84,5			0.3-999		Line lengt			
	4		Line Length		0,18				1.49 miles	-	h in miles		
_	5	1005 LIN			1200			1-20,00		VT ratio			
	6	1006 LIN			300		-	1-20,00		CT ratio			
TEXT			Known Char		Cartes	ian					artesian, polar or other		
	8	1008 LIN			0,001			0.001-9	99 ohms	Positive s	eq. line Z (sec. ohms)		
	9	1009 LIN	I Phid		0			0-90 de	g		equence line angle		
NUM		100A LIN	I Z01		0,001			0.001-9	99 ohms	Zero seq.	line Z (sec. ohms) zone 1		
	11	100B LIN	I Phi01		0			-90 to +9	90 deg	Zero sequ	ence line angle (zone 1)		
NUM	12	100C LIN	N Z02		0,001			0.001-9	99 ohms	Zero seq.	line Z (sec. ohms) other zone		
NUM	13	100D LIN	N Phi02		0			-90 to +9	30 deg	Zero sequ	ence line angle (other zones)		
NUM	14	100E LIN	l Rd		1,885			0.001-9	99 ohms	Pos. seq.	line R (sec. ohms)		
NUM	15	100F LIN	1 Xd		4,781			0.001-9	99 ohms	Pos. seq.	line X (sec. ohms)		
NUM	16	1010 LIN	I R01		4,702			0.001-9	99 ohms	Zero seq.	line R (sec. ohms) zone 1		
NUM	17	1011 LIN	1×01		15,02			-999 - 9	99 ohms	Zero seq.	line X (sec. ohms) zone 1		
NUM	18	1012 LIN	I R02		4,702			0.001-9	99 ohms	Zero seq.	line R (sec. ohms) other zone		
NUM	19	1013 LIN	1×02		15,02			-999 - 9	99 ohms	Zero seq.	line X (sec. ohms) other zone		
NUM	20	1014 LIN	l K01r		0			-7 - 7		Re(K0) for	rzone 1		
NUM	21	1015 LIN	I K01x		0			-7 - 7		Im(K0) for	zone 1		
NUM	22	1016 LIN	1 K02r		0			-7 - 7		Re(K0) for	r other zones		
NUM	23	1017 LIN	1 K02x		0			-7 - 7		Im(K0) for	other zones		
NUM	24	1101 ZO	N Z1		7,68			0.1-200	ohms	Zone 1 im	pedance (sec. ohms)		
NUM	25	1102 ZO	N Z1 Overrea	:h	0,1			0.1-200	ohms	Extended	Zone 1 impedance (sec. ohm		
NUM	26	1103 ZO	N T1		0			0-10 se	c	Zone 1 tim	ne delav		
NUM		1104 ZO			11,39			0.1-200			pedance (sec. ohms)		
	28	1105 ZO			0,4			0-10 se		Zone 2 tim			
_	29	1106 ZO			2,44			0.1-200			pedance (sec. ohms)		
	30	1107 ZO			1			0-10 se		Zone 3 tim			
TEXT		1108 ZO			Backw	ards			Is/Backwar				
NUM		1109 ZO			13,44			0.1-200			pedance (sec. ohms)		
NUM		110A ZO			0,8			0-10 se		Zone 4 tim			
NUM		110B ZO			0,1			0.1-200			pedance (sec. ohms)		
NUM		1100 ZO			0			0-10 se		Zone 5 tin			
NUM		110D Z0			0			0-10 se		I>> time d			
NUM		110E ZO			0			0-10 se		I> time de			
NUM			N Ph/Gnd RZ	1	7,53			0-200 0			R for zone 1		
NUM													
			N Ph/Ph RZ1		7,33			0-200 0			p R for zone 1		
NUM			N RLim Z2		8,36			0-200 0		Loop R for			
NUM	41	1112 ZO	N RLim Z3		5			0-200 o	nms	Loop R for	rzone 3		

		e Remarks				- 1	e Tag:	59		Archived
Local Z	one of	Protection Data	1			Relay	/Tag:	57		
Subst	ation	Tintibi		C	hange LZOP	Devic	e Name	EP	AC_Tinti-G	el(New)
LZOP	ID	LINE 72		_						
Name		132kV Tingtib	i-Gelephu feed	ler		Activ	e Group			*
LZOP	Rank	2	-		View LZOP					<< Advanced
		,	-						_	<< Advanced
iroup Dis	splayed				Rename	Copy		Delete	Compare	Move
	,,			-						
Relay S	tyle	EP311111BC	DHF	▼ Se	lect Style	Manufact Relay Mod		LSTOM PAC 3136/	3536	
Scher	ne	MAIN		*		Category:		igital Pack		
telay info	Elem	ents Commo	n Taps Misc	ellaneous	Memos					
Find	Tap Na	me		Find	🗖 Filter Ta	aps by Fun	ction			•
Type	Numb	e Tap Name		Setting		Remarks	Range	,	Tap Descr	
NUM		1003 LIN Line	Length	44,5		. comonica	0.3-999.	99 km	Line lengt	
NUM	-	1004 LIN Line	-	0,18					Line lengt	
NUM		1005 LIN Ku		1			1-20,000		VT ratio	
NUM		1006 LIN Ki		1			1-20,000		CT ratio	
TEXT		1007 LIN Kno	wn Char	Cartes	ian					rtesian, polar or other
NUM	8	1008 LIN Zd	in one	0,001				99 ohms		eq. line Z (sec. ohms)
NUM	9	1009 LIN Phic	4	0,001			0-90 de			equence line angle
_	10	100A LIN Z01	,	0,001				99 ohms		line Z (sec. ohms) zone 1
NUM	11	100B LIN Phil	01	0,001			-90 to +9			ence line angle (zone 1)
NUM		1000 LIN Z02		0,001				99 ohms		line Z (sec. ohms) other zone
NUM		100D LIN Phil		0,001			-90 to +9			ence line angle (other zones)
NUM		100E LIN Rd	02	1,886				99 ohms		line R (sec. ohms)
NUM	15	100F LIN Xd		4,782				99 ohms		line X (sec. ohms)
NUM	16	1010 LIN R01		4,702				99 ohms		line R (sec. ohms) zone 1
NUM	17	1011 LIN X01		15,02				99 ohms		line X (sec. ohms) zone 1
NUM	18	1012 LIN R02	,	5,759				99 ohms		line R (sec. ohms) other zone
NUM	19	1012 LIN K02		18,39				99 ohms		line X (sec. ohms) other zone
NUM	20	1013 LIN K01		0			-7 - 7	aa onins		
NUM				0			-7 - 7		Re(K0) for	
NUM		1015 LIN K01 1016 LIN K02		0			-7 - 7		Im(K0) for	
_				0						other zones
NUM		1017 LIN K02	X	-			-7 - 7			other zones
NUM		1101 ZON Z1	Ouerreach	4,11			0.1-200			pedance (sec. ohms) Zene 1 impedance (sec. ohm
NUM	25	1102 ZON Z1	overreach	0,1			0.1-200			Zone 1 impedance (sec. ohm
NUM	26	1103 ZON T1		0			0-10 sec		Zone 1 tim	
NUM	27	1104 ZON Z2		8,15			0.1-200			pedance (sec. ohms)
NUM	28	1105 ZON T2		0,4			0-10 sec		Zone 2 tim	
NUM	29	1106 ZON Z3		1,28			0.1-200			pedance (sec. ohms)
NUM	30	1107 ZON T3		1			0-10 sec		Zone 3 tim	ie delay
TEXT		1108 ZON Dir	. Z3	Backwa	ards		Forward	s/Backwar		
NUM		1109 ZON Z4		10,92			0.1-200	ohms	Zone 4 im	pedance (sec. ohms)
NUM		110A ZON T4		0,8			0-10 sec	:	Zone 4 tim	
NUM	34	110B ZON Z5		0,1			0.1-200	ohms	Zone 4 im	pedance (sec. ohms)
NUM	35	110C ZON T5		0			0-10 sec	:	Zone 5 tim	ie delay
NUM	36	110D ZON T>	>	0			0-10 sec	0	I>> time de	elay
NUM	37	110E ZON T>		0			0-10 sec	0	I> time del	ay
NUM	38	110F ZON Ph	Gnd RZ1	4			0-200 of	nms	Ph-E loop	R for zone 1
NUM	39	1110 ZON Ph	Ph RZ1	4			0-200 of	nms	Ph-Ph loop	p R for zone 1
NUM	40	1111 ZON RL	im Z2	5			0-200 of	nms	Loop R for	zone 2
NUM		1112 ZON RL		5			0-200 of		Loop R for	
	42	1113 ZON RL		5			0-200 of			zones 4 and 5 (starter)

		e Remarks			- 1	e Tag:	60	I	Archived
Local Z	one of F	Protection Data			Relay	/ Tag:	58		
Substa		Gelephu	C	hange LZOP	Devic	e Name	EP.	AC_Gphu-T	ing(New)
LZOPI	D	LINE 72							
Name		132kV Gelephu-Tingtibi fe			Activ	e Group			_
LZOP	Rank	2		View LZOP					<< Advanced
					-				- Advanced
roup Dis	played			Rename	Copy		Delete	Compare	Move
Relay St	tyle	EP311111BCDHF	▼ Se	lect Style	Manufact		LSTOM		
Scher		MAIN			Relay Mod Category:		PAC 3136/ igital Pack		
		, 	_		category.		igital r acto	age	
elay info	Elem	ents Common Taps Mis	cellaneous	Memos					
Find	Tap Nai	me	Find	📔 🗖 Filter T	aps by Fun	ction			-
Type	Numbe	Tap Name	Setting	-	Remarks	Range	,	Tap Descri	
NUM		1003 LIN Line Length	44,5		riteriterina	0.3-999.	99 km	Line length	
	4	1004 LIN Line Length	0,18					Line length	
NUM		1005 LIN Ku	1			1-20,000		VT ratio	
NUM		1006 LIN KI	1			1-20,000		CT ratio	
	7	1007 LIN Known Char	Cartes	ian					rtesian, polar or other
	8	1008 LIN Zd	0,001				99 ohms		q. line Z (sec. ohms)
	9	1009 LIN Phid	0			0-90 de			quence line angle
NUM	-	100A LIN Z01	0,001				99 ohms		ine Z (sec. ohms) zone 1
NUM	11	100B LIN Phi01	0			-90 to +9			ence line angle (zone 1)
NUM	12	100C LIN Z02	0,001				99 ohms		ine Z (sec. ohms) other zone
_	13	100D LIN Phi02	0,001			-90 to +9			ence line angle (other zones)
	14	100E LIN Rd	1,886				99 ohms		ine R (sec. ohms)
	15	100F LIN Xd	4,782				99 ohms		ine X (sec. ohms)
	16	1010 LIN R01	4,702				99 ohms		ine R (sec. ohms) zone 1
	17	1011 LIN X01	15,02				99 ohms		ine X (sec. ohms) zone 1
	18	1012 LIN R02	6,488				99 ohms		ine R (sec. ohms) other zone
_	19	1013 LIN X02	20,72				99 ohms		ine X (sec. ohms) other zone
NUM	20	1014 LIN K01r	0			-7 - 7	55 011115	Re(K0) for	
_	20	1015 LIN K01x	0			-7 - 7		Im(K0) for a	
_	22	1016 LIN K02r	0			-7 - 7			other zones
	23	1017 LIN K02x	0			-7 - 7			other zones
NUM		1101 ZON Z1	4,11			0.1-200	ohme		pedance (sec. ohms)
NUM		1102 ZON Z1 Overreach	0,1			0.1-200			Cone 1 impedance (sec. ohm
			0		-				
	26	1103 ZON T1	-			0-10 sec		Zone 1 time	
_	27	1104 ZON Z2	7,1			0.1-200			edance (sec. ohms)
	28	1105 ZON T2	0,4			0-10 sec		Zone 2 time	
	29 30	1106 ZON Z3	1,28			0.1-200			edance (sec. ohms)
	30 31	1107 ZON T3		ardo		0-10 sec	: s/Backwari	Zone 3 time	e uelay
		1108 ZON Dir. Z3	Backwa 12.95	ai u 5					adance leas shows
NUM		1109 ZON Z4	12,95			0.1-200			edance (sec. ohms)
NUM		110A ZON T4	0,8			0-10 sec 0.1-200		Zone 4 time Zone 4 imp	
		110B ZON Z5	0,1					Zone 5 time	edance (sec. ohms)
NUM		110C ZON T5	0			0-10 sec			
NUM		110D ZON T>>	0			0-10 sec		I>> time de	
NUM	37	110E ZON T>	0			0-10 sec		I> time dela	
		110F ZON Ph/Gnd RZ1	4			0-200 of		Ph-E loop F	
NUM		1110 ZON Ph/Ph RZ1	4			0-200 of			R for zone 1
NUM		1111 ZON RLim Z2	5			0-200 of		Loop R for:	
	41	1112 ZON RLim Z3	0			0-200 of	ims	Loop R for:	zone a

220kV Feeders:

		Remarks			Device		85		Archived	
		Protection Data			Relay	Tag:	83			
Subst LZOP		Semtokha LINE 18	Ch	ange LZOP	Device	e Name	R	EL511 Sem_R	uri(New)	
Name		220kV Semtokha	-Rurichu feeder		Active	Group				
LZOP	Rank	2	V	iew LZOP						
								<< Advanced		
roup Dis	played		*	Rename	Сору	0	elete	Compare	Move	
Relay S	ndo	DEL 614, VO.2, 44		ct Style	Manufactu	rer: A8	98			
Scher		REL511_V2.3_1# Demo	Sele	ici siyie	Relay Mod Category:		EL511_V			
			aps Miscellaneous	Homos	category.		gitai naci	nage		
	Tap Nar		Find		aps by Func	tion			•	
		Tap Name	Setting		Remarks	Range		Tap Descript		
TEXT		Operation_Z1	Forward		_	Zone Op	eration	Tap Descript	1011	
TEXT		OperationPP_Z1	ON			OFF/ON				
NUM		X1PP_Z1	1,5			0.1-400	ohms			
NUM	110	R1PP_Z1	0,56			0.1-400	ohms			
NUM	111	RFPP_Z1	3			0.1-400	ohms			
TEXT	112	Timer t1PP	ON			OFF/ON				
NUM	113	t1PP	0			0-60 sec				
TEXT	114	OperationPE_Z1	ON			OFF/ON				
NUM	115	X1PE_Z1	1,5			0.1-400	ohms			
NUM	116	R1PE_Z1	0,56			0.1-400	ohms			
NUM	117	X0PE_Z1	4,48			0.1-1200	ohms			
NUM	118	R0PE_Z1	1,2			0.1-1200	ohms			
NUM	119	RFPE_Z1	6,73			0.1-400	ohms			
TEXT	120	Timer t1PE	ON			OFF/ON				
NUM		t1PE	0			0-60 sec				
TEXT		Operation_Z2	Forward			Zone Op	eration			
TEXT		OperationPP_Z2	ON			OFF/ON				
NUM		X1PP_Z2	2,68			0.1-400	ohms			
NUM		R1PP_Z2	1			0.1-400				
NUM		RFPP_Z2	4,51			0.1-400	ohms			
TEXT		Timer t2PP	ON			OFF/ON				
NUM		t2PP	0,4			0-60 sec				
TEXT		OperationPE_Z2	ON			OFF/ON				
NUM		X1PE_Z2	2,68			0.1-400				
NUM		R1PE_Z2	1			0.1-400				
NUM		X0PE_Z2	8,04			0.1-1200				
NUM		R0PE_Z2	2,14			0.1-1200				
NUM		RFPE_Z2	7,5			0.1-400	onms			
TEXT		Timer t2PE	ON			OFF/ON				
NUM		t2PE	0,4			0-60 sec				
TEXT		Operation_Z3	Forward			Zone Op	eration			
TEXT		OperationPP_Z3				OFF/ON	her -			
NUM		X1PP_Z3	3,49			0.1-400				
NUM		R1PP_Z3	1,31			0.1-400				
NUM		RFPP_Z3	5,87			0.1-400	unns			
TEXT		Timer t3PP	ON			OFF/ON				
NUM		t3PP OperationPE_Z3	0,8			0-60 sec				
TEXT						OFF/ON	abrac			
NUM NUM		X1PE_Z3 R1PE_Z3	3,49			0.1-400				
NUM		X0PE_Z3				0.1-1200		-		
NUM		R0PE_Z3	10,47 2,79			0.1-1200				
NUM		REPE_Z3	7,5			0.1-400				
TEXT		Timer t3PE	0N			OFF/ON	11113			
NUM		t3PE	0,8			0-60 sec				
	191	OF L	0,0		-	0.00 260	eration			

cation Device Ren Local Zone of Prote-						11	Devic Relay			86 CArchived 84		
Substa		Ruric		C	hange LZOP		Devic	e Name	R	EL511 Ruri-	Semt_NEW	
LZOP I Name	D	220k3	18 / Rurichu-Semtokh	a feeder			Active	Group	i i			
LZOPI	Dank	2			View LZOP		ACTIVE	oroup				
LZOPI	n ann a	1.			in the second						<< Advanced	
roup Dis	played				Rename		Сору		Delete	Compar	e Move	
Relay S	tyle	REL5	11_V2.3_1A	▼ Sel	ect Style		wfactu ay Mod		88 EL511_V	2		
Scher	ne	Demo)	*			egory:		igital Pac			
elay Info	Elem	ents	Common Taps M	iscellaneous	Memos							
Find	Tap Na	me		Find	Filter 1	Taps I	ay Func	tion				
Type	Numb	e Tap N	lame	Setting		Rer	narks	Range		Tap Desc	ription	
TEXT	107	Opera	ation_Z1	Forwar	d			Zone Op	peration			
TEXT			ationPP_Z1	ON				OFF/ON				
NUM		X1PP	-	1,82		_		0.1-400		_		
NUM		R1PP	-	0,56		-		0.1-400		_		
NUM		RFPP	-	8,52				0.1-400		_		
TEXT			r t1PP	ON		-		OFF/ON				
NUM		t1PP	HeeDE 74	0		-		0-60 se				
TEXT			ationPE_Z1	0N		-		OFF/ON				
NUM		X1PE	-	1,82		-		0.1-400				
NUM		R1PE X0PE	_	0,56		-		0.1-400				
NUM		ROPE		5,44		-		0.1-120				
NUM		RFPE	-	7,47		-		0.1-120				
TEXT			rt1PE	ON		-		OFF/ON				
NUM		t1PE		0		-		0-60 se				
TEXT			ation_Z2	Forwar	đ			Zone O				
TEXT			ationPP_Z2	ON		-		OFF/ON				
NUM		X1PP	-	4,1				0.1-400				
NUM	125	R1PP	-	1				0.1-400	ohms			
NUM	126	RFPP	_Z2	8,97				0.1-400	ohms			
TEXT	127	Time	r t2PP	ON				OFF/ON				
NUM	128	t2PP		0,4				0-60 se	c			
TEXT	129	Operation	ationPE_Z2	ON				OFF/ON				
NUM	130	X1PE	Z2	4,12				0.1-400	ohms			
NUM		R1PE	-	1				0.1-400				
NUM	132	X0PE	_Z2	12,36				0.1-120	0 ohms			
NUM		ROPE	_Z2	2,14				0.1-120	0 ohms			
NUM		RFPE	_Z2	7,98				0.1-400	ohms			
TEXT			r t2PE	ON		_		OFF/ON				
NUM		t2PE		0,4				0-60 se				
TEXT			ation_Z3	Forwar	d	-			peration			
TEXT			ationPP_Z3	ON		-		OFF/ON				
NUM		X1PP	-	5,43		-		0.1-400				
NUM		R1PP	-	1,31		-		0.1-400				
NUM		RFPP	-	8,99		-		0.1-400				
TEXT			r t3PP	ON		-		OFF/ON				
TEXT		t3PP Oper	ationPE_Z3	0,8		-		0-60 se				
NUM		X1PE		ON 5,43		-		OFF/ON 0.1-400		-		
NUM		R1PE	-	5,43		-		0.1-400		1		
NUM		X0PE		1,31		-		0.1-400				
NUM		ROPE	-	2,79		-		0.1-120		-		
NUM		RFPE		7,99		-		0.1-120				
TEXT			r t3PE	7,99 ON		-		OFF/ON				
NUM		t3PE	IN E	0,8		-		0-60 se				
11000	152		ation_Z4	Off		-			eration	-		

67N Relay setting details:

Substation: Kurichhu - R1

LZOP: "132kV Kurichu Nangkhor feeder" (LINE)

67N CDD21 New Tag: 92 Model CDD 21 CDD21 1A 0.2-0.8 Style Branch Main CT: 45-42 Ckt 1 (132.0 kV) to 42 Bus 132Nang (Nangkhor) CT Ratio: 500.00 YY Pickup 0.20 R. Amps Time Dial 0.22377 Characteristic NORMAL INVERSE Test Time 0.96 seconds Test Current 500.00 percent

Substation: Nangkhor - R2

LZOP: "Nangkhor-Nanglam feeder at Nangknor end" (LINE)

67N CDD21 NalamFdr. Tag: 86 CDD 21 Model Style CDD21 1A 0.2-0.8 Branch Main CT: 42-41 Ckt 1 (132.0 kV) to 41 Bus_132Nglam (Nganlam) CT Ratio: 300.00 YY Pickup 0.20 R. Amps Time Dial 0.19905 Characteristic NORMAL INVERSE Test Time 0.85 seconds Test Current 500.00 percent

Substation: Nganlam - R3

LZOP: "132 kV Nanglam-Tingtibi feeder" (LINE)

67N CDD21 TitibiFdr. Tag: 89 Model CDD 21 CDD21 1A 0.2-0.8 Style Branch Main CT: 41-40 Ckt 1 (132.0 kV) to 40 Bus 132Ting (Tintibi) CT Ratio: 300.00 YY Pickup 0.20 R. Amps Time Dial 0.16398 Characteristic NORMAL INVERSE Test Time 0.70 seconds 500.00 percent Test Current

Substation: Tintibi- R4

LZOP: "132kV Tingtibi-Gelephu feeder" (LINE)

```
67n CDD21-Gele Fdr. Tag:
                           94
Model
               CDD 21
Style
               CDD21 1A 0.2-0.8
Branch Main CT: 40-39 Ckt 1 (132.0 kV) to 39 Bus 132Gelp (Gelephu)
CT Ratio:
               300.00 YY
Pickup
                0.20 R. Amps
Time Dial
               0.12900
Characteristic NORMAL INVERSE
Test Time 0.55 seconds
Test Current
                500.00 percent
Substation: Gelephu - R5
```

LZOP: "132kV Gelephu-Salakati feeder" (LINE)

67N CDD21 New 93 Tag: CDD 21 Model CDD21 1A 0.2-0.8 Style Branch Main CT: 39-37 Ckt 1 (132.0 kV) to 37 Bus 132Salak (Salakati) 300.00 YY CT Ratio: 0.20 R. Amps Pickup Time Dial 0.10001 Characteristic NORMAL INVERSE Test Time 0.43 seconds Test Current 500.00 percent

Substation: Nangkhor - R1

LZOP: "132kV Nangkhor-Kurichu feeder" (LINE)

CDD21 EF(New) Tag: 96 CDD 21 Model CDD21 1A 0.2-0.8 Style Branch Main CT: 42-45 Ckt 1 (132.0 kV) to 45 Bus_132KHP (Kurichhu) CT Ratio: 300.00 YY 0.20 R. Amps Pickup Time Dial 0.15600 Characteristic NORMAL INVERSE Test Time 0.67 seconds Test Current 500.00 percent

Substation: Nganlam - R2

LZOP: "132kV Nganlam-Nagkhor feeder" (LINE)

67N CDD21 Nang(New) Tag: 97 CDD 21 Model Style CDD21 1A 0.2-0.8 Branch Main CT: 41-42 Ckt 1 (132.0 kV) to 42 Bus 132Nang (Nangkhor) CT Ratio: 300.00 YY 0.20 R. Amps Pickup Time Dial 0.13592 NORMAL INVERSE Characteristic Test Time 0.58 seconds Test Current 500.00 percent

Substation: Tintibi - R3

LZOP: "132 kV Tingtibi-Nanglam feeder" (LINE)

67N CDD21 NLam(New) Tag: 98 Model CDD 21 CDD21 1A 0.2-0.8 Style Branch Main CT: 40-41 Ckt 1 (132.0 kV) to 41 Bus 132Nglam (Nganlam) CT Ratio: 300.00 YY Pickup 0.20 R. Amps Time Dial 0.11734 Characteristic NORMAL INVERSE 0.50 seconds Test Time Test Current 500.00 percent

Substation: Gelephu - R4

LZOP: "132kV Gelephu-Tingtibi feeder" (LINE)

```
67N CDD21-Tibi(New) Tag:
                              99
                 CDD 21
Model
                 CDD21 1A 0.2-0.8
Style
Branch Main CT: 39-40 Ckt 1 (132.0 kV) to 40 Bus 132Ting (Tintibi)
                 300.00 YY
CT Ratio:
                 0.20 R. Amps
Pickup
               0.10000
Time Dial
Characteristic NORMAL INVERSE
                0.43 seconds
Test Time 0.43 seconds
Test Current 500.00 percent
```

220kV feeders:

Substation: Semtokha

LZOP: "220kV Semtokha-Rurichu feeder" (LINE)

REL511 Sem Ruri(New) Tag: 83 Model REL511 V2 REL511 V2.3 1A Style 1. Time OC IN> TEF Branch Main CT: 4-13 Ckt 1 (220.0 kV) to 13 Bus 220Ruri (Rurichu) CT Ratio: 300.00 YY 0.20 R. Amps Pickup Time Dial 0.10000 Characteristic STD INVERSE (IEC-A) 0.43 seconds Test Time Test Current 500.00 percent

4. Inst. OC IN1>
Branch Main CT: 4-13 Ckt 1 (220.0 kV) to 13 Bus_220Ruri (Rurichu)
CT Ratio: 300.00 YY
Pickup 4.46 R. Amps
Total Oper. Time 0.02 seconds

Substation: Chukha

LZOP: "220kV Chukha-Semtokha feeder" (LINE)

REL 511 Chukha-Semto Tag: 11 Model REL511 V2 REL511 V2.3 1A Style 2. Time OC IN>_TEF Branch Main CT: 2-4 Ckt 1 (220.0 kV) to 4 Bus 220Semto (Semtokha) CT Ratio: 600.00 YY Pickup 0.15 R. Amps Time Dial 0.11000 Characteristic STD INVERSE (IEC-A) Test Time 0.47 seconds Test Current 500.00 percent 5. Inst. OC IN1> Branch Main CT: 2-4 Ckt 1 (220.0 kV) to 4 Bus 220Semto (Semtokha) CT Ratio: 600.00 YY Pickup 3.87 R. Amps Total Oper. Time 0.02 second

Substation: Malbesa

LZOP: "Malbesa to Chukha(220kV)" (LINE) MC R 1 Tag: 2 REL511_V2 Model Style REL511_V2.3_1A 3. Time OC IN>_TEF Branch Main CT: 20-2 Ckt 1 (220.0 kV) to 2 Bus 220CHP (Chukha) CT Ratio: 800.00 YY Pickup 0.15 R. Amps Time Dial 0.12000 Characteristic STD INVERSE (IEC-A) Test Time 17.21 seconds Test Current 105.00 percent 6. Inst. OC IN1> Branch Main CT: 20-2 Ckt 1 (220.0 kV) to 2 ${\tt Bus_220CHP}$ (Chukha) CT Ratio: 800.00 YY Pickup 3.95 R. Amps Total Oper. Time 0.02 seconds